# UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

# FORM 8-K

### **CURRENT REPORT**

Pursuant to Section 13 or 15(d) of the Securities Exchange Act of 1934

Date of Report (Date of earliest event reported): June 29, 2005

# ENTERPRISE PRODUCTS PARTNERS L.P.

(Exact name of registrant as specified in its charters)

**Delaware** 

(State or Other Jurisdiction of Incorporation or Organization)

1-14323

(Commission File Number)

76-0568219

(I.R.S. Employer Identification No.)

2727 North Loop West, Houston, Texas

(Address of Principal Executive Offices)

**77008-1044** (Zip Code)

Registrant's Telephone Number, including Area Code: (713) 880-6500

Check the appropriate box below if the Form 8-K filing is intended to simultaneously satisfy the filing obligation of the registrant under any of the following provisions (see General Instruction A.2. below):

- o Written communications pursuant to Rule 425 under the Securities Act (17 CFR 230.425)
- o Soliciting material pursuant to Rule 14a-12 under the Exchange Act (17 CFR 240.14a-12)
- o Pre-commencement communications pursuant to Rule 14d-2(b) under the Exchange Act (17 CFR 240.14d-2(b))
- o Pre-commencement communications pursuant to Rule 13e-4(c) under the Exchange Act (17 CFR 240.13e-4(c))

# TABLE OF CONTENTS

Item 2.03. Creation of a Direct Financial Obligation or an Obligation under an Off-Balance Sheet Arrangement of a Registrant

<u>Item 8.01. Other Events</u> <u>Item 9.01. Financial Statements and Exhibits</u>

**SIGNATURES** 

**EXHIBIT INDEX** 

Amended & Restated Agreement of Limited Partnership
Amended & Restated Credit Agreement

### Item 2.03. Creation of a Direct Financial Obligation or an Obligation under an Off-Balance Sheet Arrangement of a Registrant.

On June 29, 2005, Cameron Highway Oil Pipeline Company, a Delaware general partnership ("Cameron Highway"), executed an Amended and Restated Credit Agreement (the "Credit Agreement") with the lenders named therein (the "Lenders") and SunTrust Bank, as Administrative Agent and Collateral Agent, and borrowed the full amount thereunder of \$415,000,000 (the "Loan"). Cameron Highway is owned 50% by Cameron Highway Pipeline I, L.P., a Delaware limited partnership and indirect wholly-owned subsidiary ("Enterprise Sub") of Enterprise Products Partners, L.P. ("Enterprise"), and 50% by two subsidiaries ("Valero Subs") of Valero Energy Corporation ("Valero"). The Loan has a maturity date of June 28, 2006 and is secured by (1) mortgages on and pledges of substantially all of the assets of Cameron Highway, (2) mortgages on and pledges of certain assets related to certain rights of way and pipeline assets of an indirect wholly-owned subsidiary of Enterprise that serves as the operator of the Cameron Highway Pipeline, (3) pledges by Enterprise Sub and Valero Subs of their partnership interests in Cameron Highway, and (4) letters of credit in the amount of \$14,000,000 each (the "Letters of Credit") issued by Enterprise Products Operating L.P. (a wholly-owned subsidiary of Enterprise) and Valero. Except for the foregoing, the Lenders do not have any recourse against the assets of Enterprise or any of its subsidiaries under the Credit Agreement.

A portion of the proceeds of the Loan were used to refinance Cameron Highway's existing debt, having an aggregate outstanding principal amount of \$325,000,000, and to make cash distributions to the owners of Cameron Highway. In connection with this refinancing, Cameron Highway is expected to incur approximately \$22 million in one-time make whole premiums and related fees and costs, which include \$6.3 million of non-cash charges. Enterprise's equity earnings from Cameron Highway for the three and six months ended June 30, 2005 will be reduced by its 50% share of such costs.

A copy of the Credit Agreement is attached as Exhibit 4.1 hereto and is incorporated herein by reference. Capitalized terms used herein without definitions are used as defined in the Credit Agreement.

The Loan bears interest at a variable rate at either (at the election of Cameron Highway from time to time) (1) the greater of (a) the Prime Rate or (b) the Federal Funds Rate plus 0.5%, or (2) the Adjusted LIBO Rate plus, in each case, the Applicable Margin.

The Credit Agreement contains various covenants restricting Cameron Highway's ability to: (1) incur certain indebtedness; (2) grant certain liens; (3) enter into certain merger or consolidation transactions; (4) make certain investments; (5) make certain restricted payments; (6) enter into certain hedging agreements; (7) enter into certain transactions with affiliates; (8) form any subsidiaries; (9) make any material changes in the Cameron Highway System; (10) enter into any sale and leaseback transaction; or (11) enter into or amend certain other agreements.

The Credit Agreement also requires Cameron Highway to have EBITDA of at least \$8,000,000 for the quarter ended December 31, 2005 and \$10,000,000 for the quarter ended March 31, 2006. EBITDA is defined as net income less any non-cash income and gains included therein, plus interest expense, depreciation, depletion and impairment, amortization of leasehold

and intangibles and other similar non-cash items and taxes; provided that gains or losses on the disposition of assets not in the ordinary course of business shall not be included in EBITDA.

Pursuant to the terms of the Credit Agreement, the following actions by Cameron Highway shall constitute events of default under the Credit Agreement:

- a failure to pay principal or interest on any loan, or other amounts due, under the Credit Agreement;
- if a representation or warranty shall have been proven to be incorrect when made;
- the failure to observe or perform covenants or agreements under the Credit Agreement;
- the commencement of proceedings under federal, state or foreign bankruptcy, insolvency, receivership or similar laws or similar bankruptcy-type default provisions with respect to Cameron Highway or any Material Producer;
- the appointment or application for the appointment of a receiver, trustee, custodian or similar party over a substantial portion of the assets of Cameron Highway or any Material Producer;
- failure to make a payment on any Material Indebtedness (indebtedness exceeding \$5,000,000) or the occurrence of an event of default under any Material Indebtedness;
- the entry of one or more judgments for the payment of money in an aggregate uninsured amount equal to or greater than \$10,000,000;
- the occurrence of a Change of Control of Cameron Highway;
- the abandonment, constructive total loss or condemnation of the Cameron Highway System;
- termination of or certain breaches under the Interconnect Agreement or any Effective Date Purchase and Sale Agreement;
- the Cameron Highway System becomes subject to a material Environmental Liability;
- the invalidity of Liens under the Security Documents; or
- any material provision of any Material Project Document shall cease to be valid and binding on any party thereto.

If an event of default (as defined in the Credit Agreement) occurs, then the lenders may: (1) terminate their commitments under the Credit Agreement; (2) declare any outstanding loans under the Credit Agreement to be immediately due and payable; (3) exercise all remedies

contained in the Security Documents and other Loan Documents, or available at law or in equity; and (4) draw in full on the Letters of Credit.

### Item 8.01. Other Events.

Also filed as an exhibit hereto is a copy of the Amended and Restated Agreement of Limited Partnership of Enterprise Products Operating L.P. dated as of July 31, 1998, that has been restated for convenience to include all amendments through December 10, 2003. Enterprise Products Operating L.P. is a whollyowned subsidiary of Enterprise.

### Item 9.01. Financial Statements and Exhibits.

(a) Financial statements of businesses acquired.

Not applicable.

(b) Pro forma financial information.

Not applicable.

- (c) Exhibits.
- 3.1 Amended and Restated Agreement of Limited Partnership of Enterprise Products Operating L.P. dated as of July 31, 1998 (restated to include all amendments through December 10, 2003).
- 4.1 Amended and Restated Credit Agreement dated as of June 29, 2005, among Cameron Highway Oil Pipeline Company, the Lenders party thereto, and SunTrust Bank, as Administrative Agent and Collateral Agent.

Date: July 1, 2005

# **SIGNATURES**

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this Report to be signed on its behalf by the undersigned hereunto duly authorized.

# ENTERPRISE PRODUCTS PARTNERS L.P.

By: Enterprise Products GP, LLC,

its General Partner

By: /s/ Michael J. Knesek

Name: Michael J. Knesek

Title: Senior Vice President, Controller and Principal Accounting Officer of Enterprise Products GP, LLC

-5-

# EXHIBIT INDEX

Exhibit <u>Number</u>	Exhibit Description
3.1	Amended and Restated Agreement of Limited Partnership of Enterprise Products Operating L.P. dated as of July 31, 1998 (restated to include all amendments through December 10, 2003).
4.1	Amended and Restated Credit Agreement dated as of June 29, 2005, among Cameron Highway Oil Pipeline Company, the Lenders party thereto, and SunTrust Bank, as Administrative Agent and Collateral Agent.

## AMENDED AND RESTATED

## AGREEMENT OF LIMITED PARTNERSHIP

OF

## ENTERPRISE PRODUCTS OPERATING L.P.

As Amended by

Reorganization Agreement, dated December 10, 2003

## TABLE OF CONTENTS

	ARTICLE I DEFINITIONS	Page
1.1 1.2	Definitions Construction	1 1
	ARTICLE II ORGANIZATION	
2.1	Formation	1
2.2	Name	1
2.3	Registered Office; Registered Agent; Principal Office; Other Offices	2
2.4	Purpose and Business	2
2.5	Powers	3
2.6 2.7	Power of Attorney Term	3
2.7	Title to Partnership Assets	4 4
	ARTICLE III	
	RIGHTS OF THE LIMITED PARTNER	
3.1	Limitation of Liability	5
3.2	Management of Business	5
3.3	Rights of the Limited Partner Relating to the Partnership	5
3.4	Outside Activities of the Limited Partner	6
	ARTICLE IV TRANSFER OF PARTNERSHIP INTERESTS	
4.1	Transfer Generally	6
4.2	Transfer of General Partner Interest	6
4.3	Transfer of the Limited Partner's Partnership Interests	8
4.4	Restrictions on Transfers	8
	ARTICLE V	
	CAPITAL CONTRIBUTIONS AND RELATED MATERS	
5.1	Prior Contributions; Closing Date Contribution of MLP	8
5.2	Additional Capital Contributions	8
5.3	Interest and Withdrawal	8
5.4	Capital Accounts	9
5.5	Loans from Partners	11
5.6	Limited Preemptive Rights	11
	-i-	

5.7	Fully Paid and Non-Assessable Nature of Limited Partner Partnership Interests	
	ARTICLE VI ALLOCATIONS AND DISTRIBUTIONS	
6.1 6.2 6.3	Allocations for Capital Account Purposes Allocations for Tax Purposes Distributions	11 15 17
	ARTICLE VII MANAGEMENT AND OPERATION OF BUSINESS	
7.1	Management	17
7.2	Certificate of Limited Partnership	19
7.3	Restrictions on General Partner's Authority	20
7.4	Reimbursement of the General Partner	20
7.5	Outside Activities	21
7.6	Loans from the General Partner; Loans or Contributions from the Partnership; Contracts with Affiliates; Certain Restrictions on the General Partner	22
7.7	Indemnification	23
7.8	Liability of Indemnitees	25
7.9	Resolution of Conflicts of Interest	26
7.10	Other Matters Concerning the General Partner	27
7.11	Reliance by Third Parties	28
	ARTICLE VIII BOOKS, RECORDS, ACCOUNTING AND REPORTS	
8.1 8.2	Records and Accounting Fiscal Year	28 29
	ARTICLE IX	
	TAX MATTERS	
9.1	Preparation of Tax Returns	29
9.2	Tax Elections	29
9.3	Tax Controversies	29
9.4	Withholding	29
	ARTICLE X ADMISSION OF PARTNERS	
10.1	Admission of the General Partner and Limited Partner	30
10.1	Admission of the General Partner and Limited Partner  Admission of Substituted Limited Partners	30
10.2	Admission of Substituted Entitled Fartners  Admission of a Successor or Transferee General Partner	30
10.3	Admission of Additional Limited Partners	30
10.5	Amendment of Agreement and Certificate of Limited Partnership	31
	- ii -	

	A D. W. C. W. W.	Page
	ARTICLE XI	
	WITHDRAWAL OR REMOVAL OF PARTNERS	
11.1	Withdrawal of the General Partner	31
11.2	Removal of the General Partner	32
11.3	Interest of Departing Partner and Successor General Partner	32
11.4	Withdrawal of the Limited Partner	33
	ARTICLE XII DISSOLUTION AND LIQUIDATION	
10.1	Divide the	22
12.1	Dissolution	33
12.2	Continuation of the Business of the Partnership After Dissolution	34
12.3 12.4	Liquidator Liquidation	34 35
12.4	Cancellation of Certificate of Limited Partnership	36
12.6	Return of Capital Contributions	36
12.7	Waiver of Partition	36
12.8	Capital Account Restoration	36
	ARTICLE XIII	
	AMENDMENT OF PARTNERSHIP AGREEMENT	
13.1	Amendment to be Adopted Solely by the General Partner	36
13.2	Amendment Procedures	37
	ARTICLE XIV	
	MERGER	
14.1	Authority	37
14.2	Procedure for Merger or Consolidation	38
14.3	Approval by the Limited Partner of Merger or Consolidation	38
14.4	Certificate of Merger	39
14.5	Effect of Merger	39
	ARTICLE XV	
	GENERAL PROVISIONS	
15.1	Addresses and Notices	40
15.2	References	40
15.3	Further Action	40
15.4	Binding Effect	40
15.5	Integration	40
15.6	Creditors	40
15.7	Waiver	40
15.8	Counterparts	40
15.9	Applicable Law	40
15.10	Invalidity of Provisions	41
15.11	Amendments to Reflect GP Reorganization Agreement	41
	- iii -	

# AMENDED AND RESTATED AGREEMENT OF LIMITED PARTNERSHIP OF ENTERPRISE PRODUCTS OPERATING L.P.

THIS AMENDED AND RESTATED AGREEMENT OF LIMITED PARTNERSHIP OF ENTERPRISE PRODUCTS OPERATING L.P. dated as of July 31, 1998, restated as amended by the GP Reorganization Agreement dated as of December 10, 2003, is entered into by and among Enterprise Products OLPGP, Inc., a Delaware corporation ("OLPGP"), as the General Partner, and Enterprise Products Partners L.P., a Delaware limited partnership (the "MLP"), as the Limited Partner, together with any other Persons who become Partners in the Partnership or parties hereto as provided herein. In consideration of the covenants, conditions and agreements contained herein, the parties hereto hereby agree as follows:

# ARTICLE I DEFINITIONS

- 1.1 *Definitions*. The definitions listed on Attachment I shall be for all purposes, unless otherwise clearly indicated to the contrary, applied to the terms used in this Agreement.
- 1.2 *Construction.* Unless the context requires otherwise: (a) any pronoun used in this Agreement shall include the corresponding masculine, feminine or neuter forms, and the singular form of nouns, pronouns and verbs shall include the plural and vice versa; (b) references to Articles and Sections refer to Articles and Sections of this Agreement; and (c) "include" or "includes" means includes, without limitation, and "including" means including, without limitation.

### ARTICLE II ORGANIZATION

- 2.1 *Formation.* The General Partner and the MLP have previously formed the Partnership as a limited partnership pursuant to the provisions of the Delaware Act and hereby amend and restate the original Agreement of Limited Partnership of Enterprise Products Operating L.P., as amended by the First Amendment, in its entirety. This amendment and restatement shall become effective on the date of this Agreement. Except as expressly provided to the contrary in this Agreement, the rights, duties (including fiduciary duties), liabilities and obligations of the Partners and the administration, dissolution and termination of the Partnership shall be governed by the Delaware Act. All Partnership Interests shall constitute personal property of the owner thereof for all purposes and a Partner has no interest in specific Partnership property.
- 2.2 *Name.* The name of the Partnership shall be "Enterprise Products Operating L.P." The Partnership's business may be conducted under any other name or names deemed necessary or appropriate by the General Partner in its sole discretion. The words "Limited Partnership," "L.P.," "Ltd." or similar words or letters shall be included in the Partnership's name where necessary for the purpose of complying with the laws of any jurisdiction that so requires. The General Partner in its discretion may change the name of the Partnership at any

time and from time to time and shall notify the Limited Partner of such change in the next regular communication to the Limited Partner.

- 2.3 *Registered Office*; *Registered Agent*; *Principal Office*; *Other Offices*. Unless and until changed by the General Partner, the registered office of the Partnership in the State of Delaware shall be located at 1209 Orange Street, New Castle County, Wilmington, Delaware 19801, and the registered agent for service of process on the Partnership in the State of Delaware at such registered office shall be The Corporation Trust Company. The principal office of the Partnership shall be located at 2727 North Loop West, Houston, Texas 77008 or such other place as the General Partner may from time to time designate by notice to the Limited Partner. The Partnership may maintain offices at such other place or places within or outside the State of Delaware as the General Partner deems necessary or appropriate. The address of the General Partner shall be 2727 North Loop West, Houston, Texas 77008 or such other place as the General Partner may from time to time designate by notice to the Limited Partner.
- 2.4 *Purpose and Business*. The purpose and nature of the business to be conducted by the Partnership shall be to conduct the following businesses or activities, in each case provided that any such business or activity is conducted within North America:
- (a) to serve as a partner, member or other equity owner of any Subsidiaries of the Partnership pursuant to the applicable partnership agreements, limited liability company agreements or other applicable documents and, in connection therewith, to exercise all of the rights and powers conferred upon the Partnership under such agreements or documents;
- (b) to acquire, manage, lease, sell, operate and otherwise deal with any and all assets or properties contributed or transferred to the Partnership (by operation of law or otherwise) by the General Partner, its Affiliates or any other Persons prior to or in connection with the consummation of the transactions taking place on the Closing Date and any similar assets or properties and, in connection therewith, to exercise all of the rights and powers conferred upon the Partnership pursuant to any agreements relating to such assets;
- (c) to engage directly in, or to enter into or form any corporation, limited liability company, partnership, joint venture or other arrangement to engage indirectly in, any type of business or activity engaged in by EPC or its Affiliates immediately prior to the Closing Date and, in connection therewith, to exercise all of the rights and powers conferred upon the Partnership pursuant to the agreements relating to such business or activity;
- (d) to engage directly in, or enter into or form any corporation, partnership, joint venture, limited liability company or other arrangement to engage indirectly in, any business activity that is approved by the General Partner and which may lawfully be conducted by a limited partnership organized pursuant to the Delaware Act and, in connection therewith, to exercise all of the rights and powers conferred upon the Partnership pursuant to the agreements relating to such business activity; provided, however, that the General Partner determines in good faith, prior to the conduct of such activity, that the conduct by the Partnership of such activity is not likely to result in the Partnership being treated as an association taxable as a corporation for federal income tax purposes; and

(e) to do anything necessary or appropriate to the foregoing, including, without limitation, the making of capital contributions to a Group Member, the MLP or any Subsidiary of the MLP.

The General Partner has no obligation or duty to the Partnership, the Limited Partner or any Assignee to propose or approve, and in its sole discretion may decline to propose or approve, the conduct by the Partnership of any business.

2.5 *Powers.* The Partnership shall be empowered to do any and all acts and things necessary, appropriate, proper, advisable, incidental to or convenient for the furtherance and accomplishment of the purposes and business described in Section 2.4 and for the protection and benefit of the Partnership.

### 2.6 Power of Attorney.

- (a) The Limited Partner and each Assignee hereby constitutes and appoints the General Partner and, if a Liquidator (other than the General Partner) shall have been selected pursuant to Section 12.3, the Liquidator, severally (and any successor to either thereof by merger, transfer, assignment, election or otherwise) and each of their authorized officers and attorneys-in-fact, as the case may be, with full power of substitution, as his true and lawful agent and attorney-in-fact, with full power and authority in his name, place and stead, to:
- (i) execute, swear to, acknowledge, deliver, file and record in the appropriate public offices (A) all certificates, documents and other instruments (including this Agreement and the Certificate of Limited Partnership and all amendments or restatements hereof or thereof) that the General Partner or the Liquidator deems necessary or appropriate to form, qualify or continue the existence or qualification of the Partnership as a limited partnership (or a partnership in which the limited partners have limited liability) in the State of Delaware and in all other jurisdictions in which the Partnership may conduct business or own property; (B) all certificates, documents and other instruments that the General Partner or the Liquidator deems necessary or appropriate to reflect, in accordance with its terms, any amendment, change, modification or restatement of this Agreement; (C) all certificates, documents and other instruments (including conveyances and a certificate of cancellation) that the General Partner or the Liquidator deems necessary or appropriate to reflect the dissolution and liquidation of the Partnership pursuant to the terms of this Agreement; (D) all certificates, documents and other instruments relating to the admission, withdrawal, removal or substitution of any Partner pursuant to, or other events described in, Article IV, X, XI or XII; (E) all certificates, documents and other instruments relating to the determination of the rights, preferences and privileges of any class or series of Partnership Interests; and (F) all certificates, documents and other instruments (including agreements and a certificate of merger) relating to a merger or consolidation of the Partnership pursuant to Article XIV; and
- (ii) execute, swear to, acknowledge, deliver, file and record all ballots, consents, approvals, waivers, certificates, documents and other instruments necessary or appropriate, in the discretion of the General Partner or the Liquidator, to make, evidence, give, confirm or ratify any vote, consent, approval, agreement or other action that is made or given by the Partners hereunder or is consistent with the terms of this Agreement or is necessary or

appropriate, in the discretion of the General Partner or the Liquidator, to effectuate the terms or intent of this Agreement; provided, that when the approval of the Limited Partner is required by any provision of this Agreement, the General Partner or the Liquidator may exercise the power of attorney made in this Section 2.6(a)(ii) only after the necessary vote, consent or approval of the Limited Partner is obtained.

Nothing contained in this Section 2.6(a) shall be construed as authorizing the General Partner to amend this Agreement except in accordance with Article XIII or as may be otherwise expressly provided for in this Agreement.

- (b) The foregoing power of attorney is hereby declared to be irrevocable and a power coupled with an interest, and it shall survive and, to the maximum extent permitted by law, not be affected by the subsequent death, incompetency, disability, incapacity, dissolution, bankruptcy or termination of the Limited Partner or Assignee and the transfer of all or any portion of the Limited Partner's or Assignee's Partnership Interest and shall extend to the Limited Partner's or Assignee's heirs, successors, assigns and personal representatives. The Limited Partner or Assignee hereby agrees to be bound by any representation made by the General Partner or the Liquidator acting in good faith pursuant to such power of attorney; and the Limited Partner or Assignee hereby waives, to the maximum extent permitted by law, any and all defenses that may be available to contest, negate or disaffirm the action of the General Partner or the Liquidator taken in good faith under such power of attorney. The Limited Partner or Assignee shall execute and deliver to the General Partner or the Liquidator, within 15 days after receipt of the request therefor, such further designation, powers of attorney and other instruments as the General Partner or the Liquidator deems necessary to effectuate this Agreement and the purposes of the Partnership.
- 2.7 *Term.* The term of the Partnership commenced upon the filing of the Certificate of Limited Partnership in accordance with the Delaware Act and shall continue in existence until the close of Partnership business on December 31, 2088, or until the earlier termination of the Partnership in accordance with the provisions of Article XII. The existence of the Partnership as a separate legal entity shall continue until the cancellation of the Certificate of Limited Partnership as provided in the Delaware Act.
- 2.8 *Title to Partnership Assets.* Title to Partnership assets, whether real, personal or mixed and whether tangible or intangible, shall be deemed to be owned by the Partnership as an entity, and no Partner or Assignee individually or collectively, shall have any ownership interest in such Partnership assets or any portion thereof. Title to any or all of the Partnership assets may be held in the name of the Partnership, the General Partner, one or more of its Affiliates or one or more nominees, as the General Partner may determine. The General Partner hereby declares and warrants that any Partnership assets for which record title is held in the name of the General Partner, one or more of its Affiliates or one or more nominees shall be held by the General Partner or such Affiliate or nominee for the use and benefit of the Partnership in accordance with the provisions of this Agreement; provided, however, that the General Partner shall use reasonable efforts to cause record title to such assets (other than those assets in respect of which the General Partner determines that the expense and difficulty of conveyancing makes transfer of record title to the Partnership impracticable) to be vested in the Partnership as soon as reasonably practicable; provided, further, that, prior to the withdrawal or removal of the General Partner or

as soon thereafter as practicable, the General Partner shall use reasonable efforts to effect the transfer of record title to the Partnership and, prior to any such transfer, will provide for the use of such assets in a manner satisfactory to the General Partner. All Partnership assets shall be recorded as the property of the Partnership in its books and records, irrespective of the name in which record title to such Partnership assets is held.

# ARTICLE III RIGHTS OF THE LIMITED PARTNER

- 3.1 *Limitation of Liability.* The Limited Partner and the Assignees shall have no liability under this Agreement except as expressly provided in this Agreement or the Delaware Act.
- 3.2 *Management of Business*. Neither the Limited Partner nor any Assignee, in its capacity as such, shall participate in the operation, management or control (within the meaning of Section 17-303(a) of the Delaware Act) of the Partnership's business, transact any business in the Partnership's name or have the power to sign documents for or otherwise bind the Partnership. Any action taken by any Affiliate of the General Partner or any officer, director, employee, member, partner, agent or trustee of the General Partner or any of its Affiliates, or any officer, director, employee, member, partner, agent or trustee of a Group Member, the MLP or any Subsidiary of the MLP, in its capacity as such, shall not be deemed to be participation in the control of the business of the Partnership by a limited partner of the Partnership (within the meaning of Section 17-303(a) of the Delaware Act) and shall not affect, impair or eliminate the limitations on the liability of the Limited Partner or Assignee under this Agreement.

### 3.3 Rights of the Limited Partner Relating to the Partnership.

- (a) In addition to other rights provided by this Agreement or by applicable law, and except as limited by Section 3.3(b), the Limited Partner shall have the right, for a purpose reasonably related to the Limited Partner's interest as a limited partner in the Partnership, upon reasonable demand and at the Limited Partner's own expense:
  - (i) to obtain true and full information regarding the status of the business and financial condition of the Partnership;
  - (ii) promptly after becoming available, to obtain a copy of the Partnership's federal, state and local tax returns for each year;
  - (iii) to have furnished to it, a current list of the name and last known business, residence or mailing address of each Partner;
- (iv) to have furnished to it, a copy of this Agreement and the Certificate of Limited Partnership and all amendments thereto, together with a copy of the executed copies of all powers of attorney pursuant to which this Agreement, the Certificate of Limited Partnership and all amendments thereto have been executed;
- (v) to obtain information regarding the amount of cash and a description and statement of the Net Agreed Value of any other Capital Contribution by each

Partner and which each Partner has agreed to contribute in the future, and the date on which each became a Partner; and

- (vi) to obtain such other information regarding the affairs of the Partnership as is just and reasonable.
- (b) Notwithstanding any other provision of this Agreement, the General Partner may keep confidential from the Limited Partner and any Assignee, for such period of time as the General Partner deems reasonable, (i) any information that the General Partner reasonably believes to be in the nature of trade secrets or (ii) other information the disclosure of which the General Partner in good faith believes (A) is not in the best interests of the MLP or the Partnership Group, (B) could damage the MLP or the Partnership Group or (C) that the MLP or any Group Member is required by law or by agreement with any third party to keep confidential (other than agreements with Affiliates the primary purpose of which is to circumvent the obligations set forth in this Section 3.3).
- 3.4 *Outside Activities of the Limited Partner*. Subject to the provisions of Section 7.5, which shall continue to be applicable to the Persons referred to therein, regardless of whether such Person shall also be a Limited Partner, the Limited Partner shall be entitled to and may have business interests and engage in business activities in addition to those relating to the Partnership, including business interests and activities in direct competition with the Partnership Group.

### ARTICLE IV TRANSFER OF PARTNERSHIP INTERESTS

### 4.1 Transfer Generally.

- (a) The term "transfer," when used in this Agreement with respect to a Partnership Interest, shall be deemed to refer to a transaction by which the General Partner assigns its Partnership Interest as a general partner in the Partnership to another Person who becomes the General Partner, or by which the Limited Partner assigns its Partnership Interest as a limited partner in the Partnership to another Person who becomes the Limited Partner or Assignee, and includes a sale, assignment, gift, pledge, encumbrance, hypothecation, mortgage, exchange or any other disposition by law or otherwise.
- (b) No Partnership Interest shall be transferred, in whole or in part, except in accordance with the terms and conditions set forth in this Article IV. Any transfer or purported transfer of a Partnership Interest not made in accordance with this Article IV shall be null and void.
- (c) Nothing contained in this Agreement shall be construed to prevent a disposition by any member of the General Partner of any or all of the issued and outstanding member interests of the General Partner.
- 4.2 *Transfer of General Partner Interest.* No provision of this Agreement shall be construed to prevent (and the Limited Partners do hereby expressly consent to) (i) the transfer by the General Partner of all or a portion of its Partnership Interest as the General Partner of the

Partnership to one or more Affiliates, which transferred Partnership Interest, to the extent not transferred to a successor General Partner, shall constitute a Limited Partner Partnership Interest or (ii) the transfer by the General Partner, in whole and not in part, of its General Partner Partnership Interest upon its merger, consolidation or other combination into any other Person or the transfer by it of all or substantially all of its assets to another Person if, in the case of a transfer described in either clause (i) or (ii) of this sentence, the rights and duties of the General Partner with respect to the General Partner Partnership Interest so transferred, or the rights and duties of a Limited Partner with respect to the Limited Partner Partnership Interest so transferred, are assumed by the transferee and the transferee agrees to be bound by the provisions of this Agreement; provided, however, that in either such case, the transferee is primarily controlled, directly or indirectly, by the MLP General Partner or any Person primarily controlling, directly or indirectly, the MLP General Partner; provided, further that in either such case, such transferee furnishes to the Partnership an Opinion of Counsel that such merger, consolidation, combination, transfer or assumption will not result in a loss of limited liability of the Limited Partners or cause the Partnership to be treated as an association taxable as a corporation or otherwise to be taxed as an entity for federal income tax purposes. In the case of a transfer pursuant to this Section 4.2 to a Person proposed as a successor general partner of the Partnership, the transferee or successor (as the case may be) shall be admitted to the Partnership as the General Partner immediately prior to the transfer of the Partnership Interest, and the business of the Partnership shall continue without dissolution.

4.3 *Transfer of the Limited Partner's Partnership Interests.* The Limited Partner may transfer all, but not less than all, of its Partnership Interest as a limited partner of the Partnership in connection with the merger, consolidation or other combination of the Limited Partner with or into any other Person or the transfer by the Limited Partner of all or substantially all of its assets to another Person, and following any such transfer such Person may become a Substituted Limited Partner pursuant to Article X. Except as set forth in the immediately preceding sentence, or in connection with any pledge of (or any related foreclosure on) the Limited Partner's Partnership Interest as a limited partner of the Partnership solely for the purpose of securing, directly or indirectly, indebtedness of the Partnership or the MLP, the Limited Partner may not transfer all or any part of its Partnership Interest or withdraw from the Partnership.

### 4.4 Restrictions on Transfers.

- (a) Notwithstanding the other provisions of this Article IV, no transfer of any Partnership Interest shall be made if such transfer would (i) violate the then applicable federal or state securities laws or rules and regulations of the Commission, any state securities commission or any other governmental authorities with jurisdiction over such transfer, (ii) terminate the existence or qualification of the Partnership or the MLP under the laws of the jurisdiction of its formation or (iii) cause the Partnership or the MLP to be treated as an association taxable as a corporation or otherwise to be taxed as an entity for federal income tax purposes (to the extent not already so treated or taxed).
- (b) The General Partner may impose restrictions on the transfer of Partnership Interests if a subsequent Opinion of Counsel determines that such restrictions are necessary to avoid a significant risk of the Partnership's or the MLP's becoming taxable as a corporation or

otherwise to be taxed as an entity for federal income tax purposes. The restrictions may be imposed by making such amendments to this Agreement as the General Partner may determine to be necessary or appropriate to impose such restrictions.

# ARTICLE V CAPITAL CONTRIBUTIONS AND RELATED MATERS

### 5.1 Prior Contributions; Closing Date Contribution of MLP.

- (a) to the date hereof, (i) the General Partner made certain Capital Contributions to the Partnership in exchange for an interest in the Partnership and has been admitted as the general partner of the Partnership, and (ii) the MLP made certain Capital Contributions to the Partnership in exchange for an interest in the Partnership and has been admitted as a Limited Partner of the Partnership.
- (b) On the Closing Date, the Partnership Interest of the General Partner in the Partnership shall be continued, subject to all of the rights, privileges and duties of the General Partner under this Agreement.
- (c) On the Closing Date, the MLP shall make a cash Capital Contribution to the Partnership equal to the net proceeds to the MLP from the Initial Offering and in exchange therefor the Partnership Interest of the MLP as a Limited Partner in the Partnership shall be continued, subject to the rights, privileges and duties of the Limited Partner hereunder.
- 5.2 *Additional Capital Contributions*. With the consent of the General Partner, the Limited Partner may, but shall not be obligated to, make additional Capital Contributions to the Partnership. Contemporaneously with the making of any such additional Capital Contributions by the Limited Partner, the General Partner shall be obligated to make an additional Capital Contribution to the Partnership in an amount equal to 0.001/99.999 of the cash, cash equivalents or Net Agreed Value of the additional Capital Contribution then made by the Limited Partner (including with respect to additional Capital Contributions by the Limited Partner of the net proceeds received by the MLP upon the issuance of Common Units pursuant to the Over-Allotment Option). Except as set forth in the immediately preceding sentence and Article XII, the General Partner shall not be obligated to make any additional Capital Contributions to the Partnership.
- 5.3 *Interest and Withdrawal*. No interest shall be paid by the Partnership on Capital Contributions. No Partner or Assignee shall be entitled to the withdrawal or return of its Capital Contribution, except to the extent, if any, that distributions made pursuant to this Agreement or upon termination of the Partnership may be considered as such by law and then only to the extent provided for in this Agreement. Except to the extent expressly provided in this Agreement, no Partner or Assignee shall have priority over any other Partner or Assignee either as to the return of Capital Contributions or as to profits, losses or distributions. Any such return shall be a compromise to which all Partners and Assignees agree within the meaning of 17-502(b) of the Delaware Act.

### 5.4 Capital Accounts.

- (a) The Partnership shall maintain for each Partner owning a Partnership Interest a separate Capital Account with respect to such Partnership Interest in accordance with the rules of Treasury Regulation Section 1.704-l(b)(2)(iv). Such Capital Account shall be increased by (i) the amount of all Capital Contributions made to the Partnership with respect to such Partnership Interest pursuant to this Agreement and (ii) all items of Partnership income and gain (including, without limitation, income and gain exempt from tax) computed in accordance with Section 5.4(b) and allocated with respect to such Partnership Interest pursuant to Section 6.1, and decreased by (x) the amount of cash or the Net Agreed Value of all actual and deemed distributions of cash or property made with respect to such Partnership Interest pursuant to this Agreement and (y) all items of Partnership deduction and loss computed in accordance with Section 5.4(b) and allocated with respect to such Partnership Interest pursuant to Section 6.1.
- (b) For purposes of computing the amount of any item of income, gain, loss or deduction which is to be allocated pursuant to Article VI and is to be reflected in the Partners' Capital Accounts, the determination, recognition and classification of any such item shall be the same as its determination, recognition and classification for federal income tax purposes (including, without limitation, any method of depreciation, cost recovery or amortization used for that purpose), provided, that:
- (i) Solely for purposes of this Section 5.4, the Partnership shall be treated as owning directly its proportionate share (as determined by the General Partner) of all property owned by any OLP Subsidiary that is classified as a partnership for federal income tax purposes.
- (ii) All fees and other expenses incurred by the Partnership to promote the sale of (or to sell) a Partnership Interest that can neither be deducted nor amortized under Section 709 of the Code, if any, shall, for purposes of Capital Account maintenance, be treated as an item of deduction at the time such fees and other expenses are incurred and shall be allocated among the Partners pursuant to Section 6.1.
- (iii) Except as otherwise provided in Treasury Regulation Section 1.704-l(b)(2)(iv)(m), the computation of all items of income, gain, loss and deduction shall be made without regard to any election under Section 754 of the Code which may be made by the Partnership and, as to those items described in Section 705(a)(1)(B) or 705(a)(2)(B) of the Code, without regard to the fact that such items are not includable in gross income or are neither currently deductible nor capitalized for federal income tax purposes. To the extent an adjustment to the adjusted tax basis of any Partnership asset pursuant to Section 734(b) or 743(b) of the Code is required, pursuant to Treasury Regulation Section 1.704-2(b)(2)(iv)(m) to be taken into account in determining Capital Accounts, the amount of such adjustment in the Capital Accounts shall be treated as an item of gain or loss.
- (iv) Any income, gain or loss attributable to the taxable disposition of any Partnership property shall be determined as if the adjusted basis of such property as of such date of disposition were equal in amount to the Partnership's Carrying Value with respect to such property as of such date.

- (v) In accordance with the requirements of Section 704(b) of the Code, any deductions for depreciation, cost recovery or amortization attributable to any Contributed Property shall be determined as if the adjusted basis of such property on the date it was acquired by the Partnership were equal to the Agreed Value of such property. Upon an adjustment pursuant to Section 5.4(d) to the Carrying Value of any Partnership property subject to depreciation, cost recovery or amortization, any further deductions for such depreciation, cost recovery or amortization attributable to such property shall be determined (A) as if the adjusted basis of such property were equal to the Carrying Value of such property immediately following such adjustment and (B) using a rate of depreciation, cost recovery or amortization derived from the same method and useful life (or, if applicable, the remaining useful life) as is applied for federal income tax purposes; provided, however, that, if the asset has a zero adjusted basis for federal income tax purposes, depreciation, cost recovery or amortization deductions shall be determined using any reasonable method that the General Partner may adopt.
- (vi) If the Partnership's adjusted basis in a depreciable or cost recovery property is reduced for federal income tax purposes pursuant to Section 48(q)(1) or 48(q)(3) of the Code, the amount of such reduction shall, solely for purposes hereof, be deemed to be an additional depreciation or cost recovery deduction in the year such property is placed in service and shall be allocated among the Partners pursuant to Section 6.1. Any restoration of such basis pursuant to Section 48(q)(2) of the Code shall, to the extent possible, be allocated in the same manner to the Partners to whom such deemed deduction was allocated.
- (c) A transferee of a Partnership Interest shall succeed to a pro rata portion of the Capital Account of the transferor relating to the Partnership Interest so transferred.
- (d) (i) In accordance with Treasury Regulation Section 1.704-l(b)(2)(iv)(f), on an issuance of additional Partnership Interests for cash or Contributed Property, the Capital Account of all Partners and the Carrying Value of each Partnership property immediately prior to such issuance shall be adjusted upward or downward to reflect any Unrealized Gain or Unrealized Loss attributable to such Partnership property, as if such Unrealized Gain or Unrealized Loss had been recognized on an actual sale of each such property immediately prior to such issuance and had been allocated to the Partners at such time pursuant to Section 6.1 in the same manner as any item of gain or loss actually recognized during such period would have been allocated. In determining such Unrealized Gain or Unrealized Loss, the aggregate cash amount and fair market value of all Partnership assets (including, without limitation, cash or cash equivalents) immediately prior to the issuance of additional Partnership Interests shall be determined by the General Partner using such reasonable method of valuation as it may adopt; provided, however, that the General Partner, in arriving at such valuation, must take fully into account the fair market value of the Partnership Interests of all Partners at such time. The General Partner shall allocate such aggregate value among the assets of the Partnership (in such manner as it determines in its discretion to be reasonable) to arrive at a fair market value for individual properties.
- (i) In accordance with Treasury Regulation Section 1.704-l(b)(2)(iv)(f), immediately prior to any actual or deemed distribution to a Partner of any Partnership property (other than a distribution of cash that is not in redemption or retirement of a Partnership Interest), the Capital Accounts of all Partners and the Carrying Value of all

Partnership property shall be adjusted upward or downward to reflect any Unrealized Gain or Unrealized Loss attributable to such Partnership property, as if such Unrealized Gain or Unrealized Loss had been recognized in a sale of such property immediately prior to such distribution for an amount equal to its fair market value, and had been allocated to the Partners, at such time, pursuant to Section 6.1 in the same manner as any item of gain or loss actually recognized during such period would have been allocated. In determining such Unrealized Gain or Unrealized Loss the aggregate cash amount and fair market value of all Partnership assets (including, without limitation, cash or cash equivalents) immediately prior to a distribution shall (A) in the case of an actual distribution which is not made pursuant to Section 12.4 or in the case of a deemed contribution and/or distribution occurring as a result of a termination of the Partnership pursuant to Section 708 of the Code, be determined and allocated in the same manner as that provided in Section 5.4(d)(i) or (B) in the case of a liquidating distribution pursuant to Section 12.4, be determined and allocated by the Liquidator using such reasonable method of valuation as it may adopt.

- 5.5 *Loans from Partners*. Loans by a Partner to the Partnership shall not constitute Capital Contributions. If any Partner shall advance funds to the Partnership in excess of the amounts required hereunder to be contributed by it to the capital of the Partnership, the making of such excess advances shall not result in any increase in the amount of the Capital Account of such Partner. The amount of any such excess advances shall be a debt obligation of the Partnership to such Partner and shall be payable or collectible only out of the Partnership assets in accordance with the terms and conditions upon which such advances are made.
- 5.6 *Limited Preemptive Rights*. Except as provided in Section 5.2, no Person shall have preemptive, preferential or other similar rights with respect to (a) additional Capital Contributions; (b) issuance or sale of any class or series of Partnership Interests, whether unissued, held in the treasury or hereafter created; (c) issuance of any obligations, evidences of indebtedness or other securities of the Partnership convertible into or exchangeable for, or carrying or accompanied by any rights to receive, purchase or subscribe to, any such Partnership Interests; (d) issuance of any right of subscription to or right to receive, or any warrant or option for the purchase of, any such Partnership Interests; or (e) issuance or sale of any other securities that may be issued or sold by the Partnership.
- 5.7 *Fully Paid and Non-Assessable Nature of Limited Partner Partnership Interests*. All Limited Partner Partnership Interests issued pursuant to, and in accordance with the requirements of, this Article V shall be fully paid and non-assessable Partnership Interests in the Partnership, except as such non-assessability may be affected by Section 17-607 of the Delaware Act.

### ARTICLE VI ALLOCATIONS AND DISTRIBUTIONS

6.1 *Allocations for Capital Account Purposes*. For purposes of maintaining the Capital Accounts and in determining the rights of the Partners among themselves, the Partnership's items of income, gain, loss and deduction (computed in accordance with Section 5.4(b)) shall be allocated among the Partners in each taxable year (or portion thereof) as provided hereinbelow.

- (a) *Net Income*. After giving effect to the special allocations set forth in Section 6.1(d), Net Income for each taxable year and all items of income, gain, loss and deduction taken into account in computing Net Income for such taxable year shall be allocated as follows:
- (i) First, 100% to the General Partner until the Net Income allocated to the General Partner pursuant to this Section 6.1(a)(i) for the current taxable year and all previous taxable years is equal to the aggregate Net Losses allocated to the General Partner pursuant to Section 6.1(b)(ii) for all previous taxable years;
  - (ii) Second, 100% to the General Partner and the Limited Partner, in accordance with their respective Percentage Interests.
- (b) Net Losses. After giving effect to the special allocations set forth in Section 6.1(d), Net Losses for each taxable period and all items of income, gain, loss and deduction taken into account in computing Net Losses for such taxable period shall be allocated as follows:
- (i) First, 100% to the General Partner and the Limited Partner, in accordance with their respective Percentage Interests; provided, that Net Losses shall not be allocated pursuant to this Section 6.1(b)(i) to the extent that such allocation would cause the Limited Partner to have a deficit balance in its Adjusted Capital Account at the end of such taxable year (or increase any existing deficit balance in its Adjusted Capital Account);
  - (ii) Second, the balance, if any, 100% to the General Partner.
- (c) Net Termination Gains and Losses. After giving effect to the special allocations set forth in Section 6.1(d), all items of income, gain, loss and deduction taken into account in computing Net Termination Gain or Net Termination Loss for such taxable period shall be allocated in the same manner as such Net Termination Gain or Net Termination Loss is allocated hereunder. All allocations under this Section 6.1(c) shall be made after Capital Account balances have been adjusted by all other allocations provided under this Section 6.1 and after all distributions of Available Cash provided under Section 6.4 have been made with respect to the taxable period ending on or before the Liquidation Date; provided, however, that solely for purposes of this Section 6.1(c), Capital Accounts shall not be adjusted for distributions made pursuant to Section 12.4.
- (i) If a Net Termination Gain is recognized (or deemed recognized pursuant to Section 5.4(d)), such Net Termination Gain shall be allocated between the General Partner and the Limited Partner in the following manner (and the Capital Accounts of the Partners shall be increased by the amount so allocated in each of the following subclauses, in the order listed, before an allocation is made pursuant to the next succeeding subclause):
- (A) First, to each Partner having a deficit balance in its Capital Account, in the proportion that such deficit balance bears to the total deficit balances in the Capital Accounts of all Partners, until each such Partner has been allocated Net Termination Gain equal to any such deficit balance in its Capital Account; and

- (B) Second, 100% to the General Partner and the Limited Partner in accordance with their respective Percentage Interests.
- (ii) If a Net Termination Loss is recognized (or deemed recognized pursuant to Section 5.4(d)), such Net Termination Loss shall be allocated to the Partners in the following manner:
- (A) First, 100% to the General Partner and the Limited Partner in proportion to, and to the extent of, the positive balances in their respective Capital Accounts; and
  - (B) Second, the balance, if any, 100% to the General Partner.
  - (d) Special Allocations. Notwithstanding any other provision of this Section 6.1, the following special allocations shall be made for such taxable period:
- (i) Partnership Minimum Gain Chargeback. Notwithstanding any other provision of this Section 6.1, if there is a net decrease in Partnership Minimum Gain during any Partnership taxable period, each Partner shall be allocated items of Partnership income and gain for such period (and, if necessary, subsequent periods) in the manner and amounts provided in Treasury Regulation Sections 1.704-2(f)(6), 1.704-2(g)(2) and 1.704-2(j)(2)(i), or any successor provision. For purposes of this Section 6.1(d), each Partner's Adjusted Capital Account balance shall be determined, and the allocation of income or gain required hereunder shall be effected, prior to the application of any other allocations pursuant to this Section 6.1(d) with respect to such taxable period (other than an allocation pursuant to Sections 6.1(d)(v) and 6.1(d)(vi)). This Section 6.1(d)(i) is intended to comply with the Partnership Minimum Gain chargeback requirement in Treasury Regulation Section 1.704-2(f) and shall be interpreted consistently therewith.
- (ii) Chargeback of Partner Nonrecourse Debt Minimum Gain. Notwithstanding the other provisions of this Section 6.1 (other than Section 6.1(d)(i)), except as provided in Treasury Regulation Section 1.704-2(i)(4), if there is a net decrease in Partner Nonrecourse Debt Minimum Gain during any Partnership taxable period, any Partner with a share of Partner Nonrecourse Debt Minimum Gain at the beginning of such taxable period shall be allocated items of Partnership income and gain for such period (and, if necessary, subsequent periods) in the manner and amounts provided in Treasury Regulation Sections 1.704-2(i)(4) and 1.704-2(j)(2)(ii), or any successor provisions. For purposes of this Section 6.1(d), each Partner's Adjusted Capital Account balance shall be determined, and the allocation of income or gain required hereunder shall be effected, prior to the application of any other allocations pursuant to this Section 6.1(d), other than Section 6.1(d)(i) and other than an allocation pursuant to Sections 6.1(d)(v) and 6.1(d)(vi), with respect to such taxable period. This Section 6.1(d)(ii) is intended to comply with the chargeback of items of income and gain requirement in Treasury Regulation Section 1.704-2(i)(4) and shall be interpreted consistently therewith.
- (iii) Qualified Income Offset. In the event any Partner unexpectedly receives any adjustments, allocations or distributions described in Treasury Regulation Sections 1.704-l(b)(2)(ii)(d)(4), 1.704-l(b)(2)(ii)(d)(5), or 1.704-l(b)(2)(ii)(d)(6), items of Partnership

income and gain shall be specially allocated to such Partner in an amount and manner sufficient to eliminate, to the extent required by the Treasury Regulations promulgated under Section 704(b) of the Code, the deficit balance, if any, in its Adjusted Capital Account created by such adjustments, allocations or distributions as quickly as possible unless such deficit balance is otherwise eliminated pursuant to Section 6.1(d)(i) or (ii).

- (iv) Gross Income Allocations. In the event any Partner has a deficit balance in its Capital Account at the end of any Partnership taxable period in excess of the sum of (A) the amount such Partner is required to restore pursuant to the provisions of this Agreement and (B) the amount such Partner is deemed obligated to restore pursuant to Treasury Regulation Sections 1.704-2(g) and 1.704-2(i)(5), such Partner shall be specially allocated items of Partnership gross income and gain in the amount of such excess as quickly as possible; provided, that an allocation pursuant to this Section 6.1(d)(iv) shall be made only if and to the extent that such Partner would have a deficit balance in its Capital Account as adjusted after all other allocations provided in this Section 6.1 have been tentatively made as if this Section 6.1(d)(iv) were not in this Agreement.
- (v) Nonrecourse Deductions. Nonrecourse Deductions for any taxable period shall be allocated to the Partners in accordance with their respective Percentage Interests. If the General Partner determines in its good faith discretion that the Partnership's Nonrecourse Deductions must be allocated in a different ratio to satisfy the safe harbor requirements of the Treasury Regulations promulgated under Section 704(b) of the Code, the General Partner is authorized, upon notice to the Limited Partner, to revise the prescribed ratio to the numerically closest ratio that does satisfy such requirements.
- (vi) Partner Nonrecourse Deductions. Partner Nonrecourse Deductions for any taxable period shall be allocated 100% to the Partner that bears the Economic Risk of Loss with respect to the Partner Nonrecourse Debt to which such Partner Nonrecourse Deductions are attributable in accordance with Treasury Regulation Section 1.704-2(i). If more than one Partner bears the Economic Risk of Loss with respect to a Partner Nonrecourse Debt, such Partner Nonrecourse Deductions attributable thereto shall be allocated between or among such Partners in accordance with the ratios in which they share such Economic Risk of Loss.
- (vii) Nonrecourse Liabilities. For purposes of Treasury Regulation Section 1.752-3(a)(3), the Partners agree that Nonrecourse Liabilities of the Partnership in excess of the sum of (A) the amount of Partnership Minimum Gain and (B) the total amount of Nonrecourse Built-in Gain shall be allocated among the Partners in accordance with their respective Percentage Interests.
- (viii) Code Section 754 Adjustments. To the extent an adjustment to the adjusted tax basis of any Partnership asset pursuant to Section 734(b) or 743(c) of the Code is required, pursuant to Treasury Regulation Section 1.704-l(b)(2)(iv)(m), to be taken into account in determining Capital Accounts, the amount of such adjustment to the Capital Accounts shall be treated as an item of gain (if the adjustment increases the basis of the asset) or loss (if the adjustment decreases such basis), and such item of gain or loss shall be specially allocated to the Partners in a manner consistent with the manner in which their Capital Accounts are required to be adjusted pursuant to such Section of the Treasury Regulations.

### (ix) Curative Allocation.

- (A) Notwithstanding any other provision of this Section 6.1, other than the Required Allocations, the Required Allocations shall be taken into account in making the Agreed Allocations so that, to the extent possible, the net amount of items of income, gain, loss and deduction allocated to each Partner pursuant to the Required Allocations and the Agreed Allocations, together, shall be equal to the net amount of such items that would have been allocated to each such Partner under the Agreed Allocations had the Required Allocations and the related Curative Allocation not otherwise been provided in this Section 6.1. Notwithstanding the preceding sentence, Required Allocations relating to (1) Nonrecourse Deductions shall not be taken into account except to the extent that there has been a decrease in Partner Nonrecourse Debt Minimum Gain. Allocations pursuant to this Section 6.1(d)(ix)(A) shall only be made with respect to Required Allocations to the extent the General Partner reasonably determines that such allocations will otherwise be inconsistent with the economic Agreement among the Partners. Further, allocations pursuant to this Section 6.1(d)(ix)(A) shall be deferred with respect to allocations pursuant to clauses (1) and (2) hereof to the extent the General Partner reasonably determines that such allocations are likely to be offset by subsequent Required Allocations.
- (B) The General Partner shall have reasonable discretion, with respect to each taxable period, to (1) apply the provisions of Section 6.1(d)(ix)(A) in whatever order is most likely to minimize the economic distortions that might otherwise result from the Required Allocations, and (2) divide all allocations pursuant to Section 6.1(d)(ix)(A) among the Partners in a manner that is likely to minimize such economic distortions.

### 6.2 Allocations for Tax Purposes.

- (a) Except as otherwise provided herein, for federal income tax purposes, each item of income, gain, loss and deduction shall be allocated among the Partners in the same manner as its correlative item of "book" income, gain, loss or deduction is allocated pursuant to Section 6.1.
- (b) In an attempt to eliminate Book-Tax Disparities attributable to a Contributed Property or Adjusted Property, items of income, gain, loss, depreciation, amortization and cost recovery deductions shall be allocated for federal income tax purposes among the Partners as follows:
- (i) (A) In the case of a Contributed Property, such items attributable thereto shall be allocated among the Partners in the manner provided under Section 704(c) of the Code that takes into account the variation between the Agreed Value of such property and its adjusted basis at the time of contribution; and (B) any item of Residual Gain or Residual Loss attributable to a Contributed Property shall be allocated among the Partners in the same manner as its correlative item of "book" gain or loss is allocated pursuant to Section 6.1.
- (ii) (A) In the case of an Adjusted Property, such items shall (1) first, be allocated among the Partners in a manner consistent with the principles of Section 704(c) of

the Code to take into account the Unrealized Gain or Unrealized Loss attributable to such property and the allocations thereof pursuant to Section 5.4(d)(i) or (ii), and (2) second, in the event such property was originally a Contributed Property, be allocated among the Partners in a manner consistent with Section 6.2(b)(i)(A); and (B) any item of Residual Gain or Residual Loss attributable to an Adjusted Property shall be allocated among the Partners in the same manner as its correlative item of "book" gain or loss is allocated pursuant to Section 6.1.

- (iii) The General Partner shall apply the principles of Treasury Regulation Section 1.704-3(d) to eliminate Book-Tax Disparities.
- (c) For the proper administration of the Partnership and for the preservation of uniformity of Units of the MLP (or any class or classes thereof), the General Partner shall have sole discretion to (i) adopt such conventions as it deems appropriate in determining the amount of depreciation, amortization and cost recovery deductions; (ii) make special allocations for federal income tax purposes of income (including, without limitation, gross income) or deductions; and (iii) amend the provisions of this Agreement as appropriate (x) to reflect the proposal or promulgation of Treasury Regulations under Section 704(b) or Section 704(c) of the Code or (y) otherwise to preserve or achieve uniformity of Units of the MLP (or any class or classes thereof). The General Partner may adopt such conventions, make such allocations and make such amendments to this Agreement as provided in this Section 6.2(c) only if such conventions, allocations or amendments would not have a material adverse effect on the Partners, the holders of any class or classes of Units of the MLP issued and outstanding or the Partnership, and if such allocations are consistent with the principles of Section 704 of the Code.
- (d) The General Partner in its discretion may determine to depreciate or amortize the portion of an adjustment under Section 743(b) of the Code attributable to unrealized appreciation in any Adjusted Property (to the extent of the unamortized Book-Tax Disparity) using a predetermined rate derived from the depreciation or amortization method and useful life applied to the Partnership's common basis of such property, despite any inconsistency of such approach with Proposed Treasury Regulation Section 1.168-2(n), Treasury Regulation Section 1.167(c)-l(a)(6) or Proposed Treasury Regulation Section 1.197-2(g)(3). If the General Partner determines that such reporting position cannot reasonably be taken, the General Partner may adopt depreciation and amortization conventions under which all purchasers acquiring Units of the MLP in the same month would receive depreciation and amortization deductions, based upon the same applicable rate as if they had purchased a direct interest in the Partnership's property. If the General Partner chooses not to utilize such aggregate method, the General Partner may use any other reasonable depreciation and amortization conventions to preserve the uniformity of the intrinsic tax characteristics of any class or classes of Units of the MLP that would not have a material adverse effect on the Limited Partner or the holders of any class or classes of Units of the MLP.
- (e) Any gain allocated to the Partners upon the sale or other taxable disposition of any Partnership asset shall, to the extent possible, after taking into account other required allocations of gain pursuant to this Section 6.2, be characterized as Recapture Income in the same proportions and to the same extent as such Partners (or their predecessors in interest) have been allocated any deductions directly or indirectly giving rise to the treatment of such gains as Recapture Income.

- (f) All items of income, gain, loss, deduction and credit recognized by the Partnership for federal income tax purposes and allocated to the Partners in accordance with the provisions hereof shall be determined without regard to any election under Section 754 of the Code which may be made by the Partnership; provided, however, that such allocations, once made, shall be adjusted as necessary or appropriate to take into account those adjustments permitted or required by Sections 734 and 743 of the Code.
- (g) The General Partner may adopt such methods of allocation of income, gain, loss or deduction between a transferor and a transferee of a Partnership Interest as it determines necessary, to the extent permitted or required by Section 706 of the Code and the regulations or rulings promulgated thereunder.

#### 6.3 Distributions.

- (a) Within 45 days following the end of each Quarter commencing with the Quarter ending on September 30, 1998, an amount equal to 100% of Available Cash with respect to such Quarter shall, subject to Section 17-607 of the Delaware Act, be distributed in accordance with this Article VI by the Partnership to the Partners in accordance with their respective Percentage Interests. The immediately preceding sentence shall not require any distribution of cash if and to the extent such distribution would be prohibited by applicable law or by any loan agreement, security agreement, mortgage, debt instrument or other agreement or obligation to which the Partnership is a party or by which it is bound or its assets are subject. All distributions required to be made under this Agreement shall be made subject to Section 17-607 of the Delaware Act.
- (b) In the event of the dissolution and liquidation of the Partnership, all receipts received during or after the Quarter in which the Liquidation Date occurs (other than from borrowings described in (a) (ii) of the definition of Available Cash) shall be applied and distributed solely in accordance with, and subject to the terms and conditions of, Section 12.4.
- (c) The General Partner shall have the discretion to treat taxes paid by the Partnership on behalf of, or amounts withheld with respect to, all or less than all of the Partners, as a distribution of Available Cash to such Partners.

# ARTICLE VII MANAGEMENT AND OPERATION OF BUSINESS

### 7.1 Management.

(a) The General Partner shall conduct, direct and manage all activities of the Partnership. Except as otherwise expressly provided in this Agreement, all management powers over the business and affairs of the Partnership shall be exclusively vested in the General Partner, and neither the Limited Partner nor any Assignee shall have any management power over the business and affairs of the Partnership. In addition to the powers now or hereafter granted a general partner of a limited partnership under applicable law or which are granted to the General Partner under any other provision of this Agreement, the General Partner, subject to Section 7.3, shall have full power and authority to do all things and on such terms as it, in its sole discretion, may deem necessary or appropriate to conduct the business of the Partnership, to exercise all

powers set forth in Section 2.5 and to effectuate the purposes set forth in Section 2.4, including the following:

- (i) the making of any expenditures, the lending or borrowing of money, the assumption or guarantee of, or other contracting for, indebtedness and other liabilities, the issuance of evidences of indebtedness, including indebtedness that is convertible into Partnership Securities, and the incurring of any other obligations;
- (ii) the making of tax, regulatory and other filings, or rendering of periodic or other reports to governmental or other agencies having jurisdiction over the business or assets of the Partnership;
- (iii) the acquisition, disposition, mortgage, pledge, encumbrance, hypothecation or exchange of any or all of the assets of the Partnership or the merger or other combination of the Partnership with or into another Person (the matters in this clause (iii) being subject, however, to any prior approval that may be required by Section 7.3);
- (iv) the use of the assets of the Partnership (including cash on hand) for any purpose consistent with the terms of this Agreement, including the financing of the conduct of the operations of the Partnership Group; subject to Section 7.6(a), the lending of funds to other Persons (including the MLP); the repayment of obligations of the MLP or any member of the Partnership Group; and the making of capital contributions to any member of the Partnership Group;
- (v) the negotiation, execution and performance of any contracts, conveyances or other instruments (including instruments that limit the liability of the Partnership under contractual arrangements to all or particular assets of the Partnership, with the other party to the contract to have no recourse against the General Partner or its assets other than its interest in the Partnership, even if same results in the terms of the transaction being less favorable to the Partnership than would otherwise be the case);
  - (vi) the distribution of Partnership cash;
- (vii) the selection and dismissal of employees (including employees having titles such as "president," "vice president," "secretary" and "treasurer") and agents, outside attorneys, accountants, consultants and contractors and the determination of their compensation and other terms of employment or hiring;
- (viii) the maintenance of such insurance for the benefit of the Partnership Group and the Partners (including the assets of the Partnership) as it deems necessary or appropriate;
- (ix) the formation of, or acquisition of an interest in, and the contribution of property and the making of loans to, any further limited or general partnerships, joint ventures, corporations or other relationships subject to the restrictions set forth in Section 2.4;

- (x) the control of any matters affecting the rights and obligations of the Partnership, including the bringing and defending of actions at law or in equity and otherwise engaging in the conduct of litigation and the incurring of legal expense and the settlement of claims and litigation;
  - (xi) the indemnification of any Person against liabilities and contingencies to the extent permitted by law; and
  - (xii) the undertaking of any action in connection with the Partnership's participation as a partner or equity owner of any Group Member.
- (b) Notwithstanding any other provision of this Agreement, the MLP Agreement, the Delaware Act or any applicable law, rule or regulation, each of the Partners and Assignees (i) approves, ratifies and confirms the execution, delivery and performance by the parties thereto of the MLP Agreement, the Underwriting Agreement, the EPCO Agreement and the other agreements described in or filed as a part of the Registration Statement that are related to the transactions contemplated by the Registration Statement; (ii) agrees that the General Partner (on its own or through any officer of the Partnership) is authorized to execute, deliver and perform the agreements referred to in clause (i) of this sentence and the other agreements, acts, transactions and matters described in or contemplated by the Registration Statement on behalf of the Partnership without any further act, approval or vote of the Partners or the Assignees or the other Persons who may acquire an interest in Partnership Securities; and (iii) agrees that the execution, delivery or performance by the General Partner, the MLP, any Group Member or any Affiliate of any of them, of this Agreement or any agreement authorized or permitted under this Agreement, shall not constitute a breach by the General Partner of any duty that the General Partner may owe the Partnership or the Limited Partner or the Assignees or any other Persons under this Agreement (or any other agreements) or of any duty stated or implied by law or equity.
- 7.2 *Certificate of Limited Partnership.* The General Partner has caused the Certificate of Limited Partnership to be filed with the Secretary of State of the State of Delaware as required by the Delaware Act and shall use all reasonable efforts to cause to be filed such other certificates or documents as may be determined by the General Partner in its sole discretion to be reasonable and necessary or appropriate for the formation, continuation, qualification and operation of a limited partnership (or a partnership in which the limited partners have limited liability) in the State of Delaware or any other state in which the Partnership may elect to do business or own property. To the extent that such action is determined by the General Partner in its sole discretion to be reasonable and necessary or appropriate, the General Partner shall file amendments to and restatements of the Certificate of Limited Partnership and do all things to maintain the Partnership as a limited partnership (or a partnership or other entity in which the limited partners have limited liability) under the laws of the State of Delaware or of any other state in which the Partnership may elect to do business or own property. Subject to the terms of Section 3.3(a), the General Partner shall not be required, before or after filing, to deliver or mail a copy of the Certificate of Limited Partnership, any qualification document or any amendment thereto to the Limited Partner or any Assignee.

### 7.3 Restrictions on General Partner's Authority.

- (a) The General Partner may not, without written approval of the specific act by the Limited Partner or by other written instrument executed and delivered by the Limited Partner subsequent to the date of this Agreement, take any action in contravention of this Agreement, including, except as otherwise provided in this Agreement, (i) committing any act that would make it impossible to carry on the ordinary business of the Partnership; (ii) possessing Partnership property, or assigning any rights in specific Partnership property, for other than a Partnership purpose; (iii) admitting a Person as a Partner; or (iv) amending this Agreement in any manner.
- (b) Except as provided in Articles XII and XIV, the General Partner may not sell, exchange or otherwise dispose of or approve on behalf of the Partnership the sale, exchange or other disposition of all or substantially all of the Partnership's assets in a single transaction or a series of related transactions without the approval of the Limited Partner; provided, however, that this provision shall not preclude or limit the General Partner's ability to mortgage, pledge, hypothecate or grant a security interest in all or substantially all of the assets of the Partnership and shall not apply to any forced sale of any or all of the assets of the Partnership pursuant to the foreclosure of, or other realization upon, any such encumbrance.

### 7.4 Reimbursement of the General Partner.

- (a) Except as provided in this Section 7.4 and elsewhere in this Agreement or in the MLP Agreement, the General Partner shall not be compensated for its services as general partner of the MLP or any Group Member.
- (b) Subject to any of the applicable limitations contained in the EPCO Agreement, the General Partner shall be reimbursed on a monthly basis, or such other reasonable basis as the General Partner may determine in its sole discretion, for (i) all direct and indirect expenses that it incurs or payments it makes on behalf of the Partnership (including amounts paid by the General Partner to EPC under the EPCO Agreement and including salary, bonus, incentive compensation and other amounts paid to any Person, including Affiliates of the General Partner, to perform services for the Partnership or for the General Partner in the discharge of its duties to the Partnership), and (ii) all other necessary or appropriate expenses allocable to the Partnership or otherwise reasonably incurred by the General Partner in connection with operating the Partnership's business (including expenses allocated to the General Partner by its Affiliates). The General Partner shall determine the expenses that are allocable to the Partnership in any reasonable manner determined by the General Partner in its sole discretion. Reimbursements pursuant to this Section 7.4 shall be in addition to any reimbursement to the General Partner as a result of indemnification pursuant to Section 7.7.
- (c) The General Partner, in its sole discretion and without the approval of the Limited Partner (who shall have no right to vote in respect thereof), may propose and adopt on behalf of the Partnership employee benefit plans, employee programs and employee practices or cause the Partnership to issue Partnership Securities, in connection with, or pursuant to, any employee benefit plan, employee program or employee practice maintained or sponsored by the General Partner or any of its Affiliates, in each case for the benefit of employees of the General

Partner, any Group Member or any Affiliate, or any of them, in respect of services performed, directly or indirectly, for the benefit of the Partnership Group. Expenses incurred by the General Partner in connection with any such plans, programs and practices shall be reimbursed in accordance with Section 7.4(b). Any and all obligations of the General Partner under any employee benefit plans, employee programs or employee practices adopted by the General Partner as permitted by this Section 7.4(c) shall constitute obligations of the General Partner hereunder and shall be assumed by any successor General Partner approved pursuant to Section 11.1 or 11.2 or the transferee of or successor to all of the General Partner's Partnership Interest as the general partner in the Partnership pursuant to Section 4.2.

### 7.5 Outside Activities.

- (a) After the Closing Date, the General Partner, for so long as it is the general partner of the Partnership (i) agrees that its sole business will be to act as the general partner or managing member of the Partnership, the MLP, and any other partnership or limited liability company of which the Partnership or the MLP is, directly or indirectly, a partner or managing member and to undertake activities that are ancillary or related thereto (including being a limited partner in the partnership), (ii) shall not engage in any business or activity or incur any debts or liabilities except in connection with or incidental to (A) its performance as general partner or managing member of one or more Group Members or as described in or contemplated by the Registration Statement or (B) the acquiring, owning or disposing of debt or equity securities in any Group Member and (iii) except to the extent permitted by the EPCO Agreement, shall not, and shall cause its Affiliates not to, engage in any Restricted Activity.
- (b) EPC has entered into the EPCO Agreement with the Partnership and the MLP, which agreement sets forth certain restrictions on the liability of EPC and its Affiliates to engage in Restricted Activities.
- (c) Except as specifically restricted by Section 7.5(a) and the EPCO Agreement, each Indemnitee (other than the General Partner) shall have the right to engage in businesses of every type and description and other activities for profit and to engage in and possess an interest in other business ventures of any and every type or description, whether in businesses engaged in or anticipated to be engaged in by any Group Member, independently or with others, including business interests and activities in direct competition with the business and activities of any Group Member, and none of the same shall constitute a breach of this Agreement or any duty express or implied by law to any Group Member or any Partner or Assignee. Neither any Group Member, the Limited Partner nor any other Person shall have any rights by virtue of this Agreement, the MLP Agreement or the partnership relationship established hereby or thereby in any business ventures of any Indemnitee.
- (d) Subject to the terms of the EPCO Agreement and Sections 7.5(a), 7.5(b), and 7.5(c), but otherwise notwithstanding anything to the contrary in this Agreement, (i) the engaging in competitive activities by any Indemnitees (other than the General Partner) in accordance with the provisions of this Section 7.5 is hereby approved by the Partnership and all Partners, (ii) it shall be deemed not to be a breach of the General Partner's fiduciary duty or any other obligation of any type whatsoever of the General Partner for the Indemnitees (other than the General Partner) to engage in such business interests and activities in preference to or the

exclusion of the Partnership and (iii) the General Partner and the Indemnitees shall have no obligation to present business opportunities to the Partnership.

- (e) The General Partner and any of its Affiliates may acquire Partnership Securities in addition to those acquired on the Closing Date and, except as otherwise provided in this Agreement, shall be entitled to exercise all rights relating to such Partnership Securities.
- (f) The term "Affiliates" when used in Sections 7.5(a) and 7.5(b) with respect to the General Partner shall not include any Group Member or any Subsidiary of a Group Member.

# 7.6 Loans from the General Partner; Loans or Contributions from the Partnership; Contracts with Affiliates; Certain Restrictions on the General Partner.

- (a) The General Partner or any of its Affiliates may lend to the MLP or any Group Member, and the MLP or any Group Member may borrow from the General Partner or any of its Affiliates, funds needed or desired by the MLP or the Group Member for such periods of time and in such amounts as the General Partner may determine; provided, however, that in any such case the lending party may not charge the borrowing party interest at a rate greater than the rate that would be charged the borrowing party or impose terms less favorable to the borrowing party than would be charged or imposed on the borrowing party by unrelated lenders on comparable loans made on an arm's-length basis (without reference to the lending party's financial abilities or guarantees). The borrowing party shall reimburse the lending party for any costs (other than any additional interest costs) incurred by the lending party in connection with the borrowing of such funds. For purposes of this Section 7.6(a) and Section 7.6(b), the term "Group Member" shall include any Affiliate of a Group Member that is controlled by the Group Member. No Group Member may lend funds to the General Partner or any of its Affiliates (other than the MLP, a Subsidiary of the MLP or another Group Member).
- (b) The Partnership may lend or contribute to any Group Member and any Group Member may borrow from the Partnership, funds on terms and conditions established in the sole discretion of the General Partner; provided, however, that the Partnership may not charge the Group Member interest at a rate less than the rate that would be charged to the Group Member (without reference to the General Partner's financial abilities or guarantees) by unrelated lenders on comparable loans. The foregoing authority shall be exercised by the General Partner in its sole discretion and shall not create any right or benefit in favor of any Group Member or any other Person.
- (c) The General Partner may itself, or may enter into an agreement, in addition to the EPCO Agreement, with any of its Affiliates to, render services to a Group Member or to the General Partner in the discharge of its duties as the general partner of the Partnership. Any services rendered to a Group Member by the General Partner or any of its Affiliates shall be on terms that are fair and reasonable to the Partnership; provided, however, that the requirements of this Section 7.6(c) shall be deemed satisfied as to (i) any transaction approved by Special Approval, (ii) any transaction, the terms of which are no less favorable to the Partnership Group than those generally being provided to or available from unrelated third parties or (iii) any transaction that, taking into account the totality of the relationships between

the parties involved (including other transactions that may be particularly favorable or advantageous to the Partnership Group), is equitable to the Partnership Group. The provisions of Section 7.4 shall apply to the rendering of services described in this Section 7.6(c).

- (d) Any Group Member may transfer assets to joint ventures, other partnerships, corporations, limited liability companies or other business entities in which it is or thereby becomes a participant upon such terms and subject to such conditions as are consistent with this Agreement and applicable law.
- (e) Neither the General Partner nor any of its Affiliates shall sell, transfer or convey any property to, or purchase any property from, the Partnership, directly or indirectly, except pursuant to transactions that are fair and reasonable to the Partnership; provided, however, that the requirements of this Section 7.6(e) shall be deemed to be satisfied as to (i) the transactions effected pursuant to Section 5.1 and any other transactions described in or contemplated by the Registration Statement, (ii) any transaction approved by Special Approval, (iii) any transaction, the terms of which are no less favorable to the Partnership than those generally being provided to or available from unrelated third parties, or (iv) any transaction that, taking into account the totality of the relationships between the parties involved (including other transactions that may be particularly favorable or advantageous to the Partnership), is equitable to the Partnership.
- (f) The General Partner and its Affiliates will have no obligation to permit the MLP or any Group Member to use any facilities or assets of the General Partner and its Affiliates, except as may be provided in contracts entered into from time to time specifically dealing with such use, nor shall there be any obligation on the part of the General Partner or its Affiliates to enter into such contracts.
- (g) Without limitation of Sections 7.6(a) through 7.6(f), and notwithstanding anything to the contrary in this Agreement, the existence of the conflicts of interest described in the Registration Statement are hereby approved by all Partners.

### 7.7 Indemnification.

(a) To the fullest extent permitted by law but subject to the limitations expressly provided in this Agreement, all Indemnitees shall be indemnified and held harmless by the Partnership from and against any and all losses, claims, damages, liabilities, joint or several, expenses (including legal fees and expenses), judgments, fines, penalties, interest, settlements or other amounts arising from any and all claims, demands, actions, suits or proceedings, whether civil, criminal, administrative or investigative, in which any Indemnitee may be involved, or is threatened to be involved, as a party or otherwise, by reason of its status as a Person of the type described in clauses (a)-(d) of the definition of the term "Indemnitee"; provided, that in each case the Indemnitee acted in good faith and in a manner that such Indemnitee reasonably believed to be in, or (in the case of a Person other than the General Partner) not opposed to, the best interests of the Partnership and, with respect to any criminal proceeding, had no reasonable cause to believe its conduct was unlawful; provided, further, no indemnification pursuant to this Section 7.7 shall be available to the General Partner with respect to its obligations incurred pursuant to the Underwriting Agreement (other than obligations incurred by the General Partner on behalf of

the Partnership or the MLP). The termination of any action, suit or proceeding by judgment, order, settlement, conviction or upon a plea of nolo contendere, or its equivalent, shall not create a presumption that the Indemnitee acted in a manner contrary to that specified above. Any indemnification pursuant to this Section 7.7 shall be made only out of the assets of the Partnership, it being agreed that the General Partner shall not be personally liable for such indemnification and shall have no obligation to contribute or loan any monies or property to the Partnership to enable it to effectuate such indemnification.

- (b) To the fullest extent permitted by law, expenses (including legal fees and expenses) incurred by an Indemnitee who is indemnified pursuant to Section 7.7(a) in defending any claim, demand, action, suit or proceeding shall, from time to time, be advanced by the Partnership prior to the final disposition of such claim, demand, action, suit or proceeding upon receipt by the Partnership of any undertaking by or on behalf of the Indemnitee to repay such amount if it shall be determined that the Indemnitee is not entitled to be indemnified as authorized in this Section 7.7.
- (c) The indemnification provided by this Section 7.7 shall be in addition to any other rights to which an Indemnitee may be entitled under any agreement, pursuant to any vote of the Partners, as a matter of law or otherwise, both as to actions in the Indemnitee's capacity as a Person of the type described in clauses (a)-(d) of the definition of the term "Indemnitee," and as to actions in any other capacity (including any capacity under the Underwriting Agreement), and shall continue as to an Indemnitee who has ceased to serve in such capacity and shall inure to the benefit of the heirs, successors, assigns and administrators of the Indemnitee.
- (d) The Partnership may purchase and maintain (or reimburse the General Partner or its Affiliates for the cost of) insurance, on behalf of the General Partner, its Affiliates and such other Persons as the General Partner shall determine, against any liability that may be asserted against or expense that may be incurred by such Person in connection with the Partnership's activities or such Person's activities on behalf of the Partnership, regardless of whether the Partnership would have the power to indemnify such Person against such liability under the provisions of this Agreement.
- (e) For purposes of this Section 7.7, the Partnership shall be deemed to have requested an Indemnitee to serve as fiduciary of an employee benefit plan whenever the performance by it of its duties to the Partnership also imposes duties on, or otherwise involves services by, it to the plan or participants or beneficiaries of the plan; excise taxes assessed on an Indemnitee with respect to an employee benefit plan pursuant to applicable law shall constitute "fines" within the meaning of Section 7.7(a); and action taken or omitted by it with respect to any employee benefit plan in the performance of its duties for a purpose reasonably believed by it to be in the interest of the participants and beneficiaries of the plan shall be deemed to be for a purpose which is in, or not opposed to, the best interests of the Partnership.
- (f) In no event may an Indemnitee subject the Limited Partner to personal liability by reason of the indemnification provisions set forth in this Agreement.

- (g) An Indemnitee shall not be denied indemnification in whole or in part under this Section 7.7 because the Indemnitee had an interest in the transaction with respect to which the indemnification applies if the transaction was otherwise permitted by the terms of this Agreement.
- (h) The provisions of this Section 7.7 are for the benefit of the Indemnitees, their heirs, successors, assigns and administrators and shall not be deemed to create any rights for the benefit of any other Persons.
- (i) No amendment, modification or repeal of this Section 7.7 or any provision hereof shall in any manner terminate, reduce or impair the right of any past, present or future Indemnitee to be indemnified by the Partnership, nor the obligations of the Partnership to indemnify any such Indemnitee under and in accordance with the provisions of this Section 7.7 as in effect immediately prior to such amendment, modification or repeal with respect to claims arising from or relating to matters occurring, in whole or in part, prior to such amendment, modification or repeal, regardless of when such claims may arise or be asserted.

#### 7.8 Liability of Indemnitees.

- (a) Notwithstanding anything to the contrary set forth in this Agreement, no Indemnitee shall be liable for monetary damages to the Partnership, the Limited Partner, the Assignees or any other Persons who have acquired interests in the Units, for losses sustained or liabilities incurred as a result of any act or omission if such Indemnitee acted in good faith.
- (b) Subject to its obligations and duties as General Partner set forth in Section 7.1 (a), the General Partner may exercise any of the powers granted to it by this Agreement and perform any of the duties imposed upon it hereunder either directly or by or through its agents, and the General Partner shall not be responsible for any misconduct or negligence on the part of any such agent appointed by the General Partner in good faith.
- (c) To the extent that, at law or in equity, an Indemnitee has duties (including fiduciary duties) and liabilities relating thereto to the Partnership or to the Partners, the General Partner and any other Indemnitee acting in connection with the Partnership's business or affairs shall not be liable to the Partnership or to any Partner for its good faith reliance on the provisions of this Agreement. The provisions of this Agreement, to the extent that they restrict or otherwise modify the duties and liabilities of an Indemnitee otherwise existing at law or in equity, are agreed by the Partners to replace such other duties and liabilities of such Indemnitee.
- (d) Any amendment, modification or repeal of this Section 7.8 or any provision hereof shall be prospective only and shall not in any way affect the limitations on the liability to the Partnership, the Limited Partner, the General Partner, and the Partnership's and General Partner's directors, officers and employees under this Section 7.8 as in effect immediately prior to such amendment, modification or repeal with respect to claims arising from or relating to matters occurring, in whole or in part, prior to such amendment, modification or repeal, regardless of when such claims may arise or be asserted.

#### 7.9 Resolution of Conflicts of Interest.

- (a) Unless otherwise expressly provided in this Agreement or the MLP Agreement, whenever a potential conflict of interest exists or arises between the General Partner or any of its Affiliates, on the one hand, and the Partnership, the MLP, any Partner or any Assignee, on the other, any resolution or course of action by the General Partner or its Affiliates in respect of such conflict of interest shall be permitted and deemed approved by all Partners, and shall not constitute a breach of this Agreement, of the MLP Agreement, of any agreement contemplated herein or therein, or of any duty stated or implied by law or equity, if the resolution or course of action is, or by operation of this Agreement is deemed to be, fair and reasonable to the Partnership. The General Partner shall be authorized but not required in connection with its resolution of such conflict of interest to seek Special Approval of such resolution. Any conflict of interest and any resolution of such conflict of interest shall be conclusively deemed fair and reasonable to the Partnership if such conflict of interest or resolution is (i) approved by Special Approval (as long as the material facts within the actual knowledge of the officers and directors of the General Partner and EPC regarding the proposed transaction were disclosed to the Audit and Conflicts Committee at the time it gave its approval), (ii) on terms no less favorable to the Partnership than those generally being provided to or available from unrelated third parties or (iii) fair to the Partnership, taking into account the totality of the relationships between the parties involved (including other transactions that may be particularly favorable or advantageous to the Partnership). The General Partner may also adopt a resolution or course of action that has not received Special Approval. The General Partner (including the Audit and Conflicts Committee in connection with Special Approval) shall be authorized in connection with its determination of what is "fair and reasonable" to the Partnership and in connection with its resolution of any conflict of interest to consider (A) the relative interests of any party to such conflict, agreement, transaction or situation and the benefits and burdens relating to such interest; (B) any customary or accepted industry practices and any customary or historical dealings with a particular Person; (C) any applicable generally accepted accounting practices or principles; and (D) such additional factors as the General Partner (including the Audit and Conflicts Committee) determines in its sole discretion to be relevant, reasonable or appropriate under the circumstances. Nothing contained in this Agreement, however, is intended to nor shall it be construed to require the General Partner (including the Audit and Conflicts Committee) to consider the interests of any Person other than the Partnership. In the absence of bad faith by the General Partner, the resolution, action or terms so made, taken or provided by the General Partner with respect to such matter shall not constitute a breach of this Agreement or any other agreement contemplated herein or a breach of any standard of care or duty imposed herein or therein or, to the extent permitted by law, under the Delaware Act or any other law, rule or regulation.
- (b) Whenever this Agreement or any other agreement contemplated hereby provides that the General Partner or any of its Affiliates is permitted or required to make a decision (i) in its "sole discretion" or "discretion," that it deems "necessary or appropriate" or "necessary or advisable" or under a grant of similar authority or latitude, except as otherwise provided herein, the General Partner or such Affiliate shall be entitled to consider only such interests and factors as it desires and shall have no duty or obligation to give any consideration to any interest of, or factors affecting, the Partnership, the MLP, the Limited Partner or, any limited partner of the Limited Partner or any Assignee, (ii) it may make such decision in its sole

discretion (regardless of whether there is a reference to "sole discretion" or "discretion") unless another express standard is provided for, or (iii) in "good faith" or under another express standard, the General Partner or such Affiliate shall act under such express standard and shall not be subject to any other or different standards imposed by this Agreement, the MLP Agreement, any other agreement contemplated hereby or under the Delaware Act or any other law, rule or regulation. In addition, any actions taken by the General Partner or such Affiliate consistent with the standards of "reasonable discretion" set forth in the definitions of Available Cash shall not constitute a breach of any duty of the General Partner to the Partnership, the Limited Partner or any limited partner of the Limited Partner. The General Partner shall have no duty, express or implied, to sell or otherwise dispose of any asset of the Partnership Group other than in the ordinary course of business. No borrowing by any Group Member or the approval thereof by the General Partner shall be deemed to constitute a breach of any duty of the General Partner to the Partnership or the Limited Partner by reason of the fact that the purpose or effect of such borrowing is directly or indirectly to (A) enable distributions to the General Partner or its Affiliates to exceed 0.001% of the total amount distributed to all partners or (B) hasten the expiration of the Subordination Period or the conversion of any Subordinated Units into Common Units.

- (c) Whenever a particular transaction, arrangement or resolution of a conflict of interest is required under this Agreement to be "fair and reasonable" to any Person, the fair and reasonable nature of such transaction, arrangement or resolution shall be considered in the context of all similar or related transactions.
- (d) The Limited Partner hereby authorizes the General Partner, on behalf of the Partnership as a partner or member of a Group Member, to approve of actions by the general partner or managing member of such Group Member similar to those actions permitted to be taken by the General Partner pursuant to this Section 7.9.

#### 7.10 Other Matters Concerning the General Partner.

- (a) The General Partner may rely and shall be protected in acting or refraining from acting upon any resolution, certificate, statement, instrument, opinion, report, notice, request, consent, order, bond, debenture or other paper or document believed by it to be genuine and to have been signed or presented by the proper party or parties.
- (b) The General Partner may consult with legal counsel, accountants, appraisers, management consultants, investment bankers and other consultants and advisers selected by it, and any act taken or omitted to be taken in reliance upon the opinion (including an Opinion of Counsel) of such Persons as to matters that the General Partner reasonably believes to be within such Person's professional or expert competence shall be conclusively presumed to have been done or omitted in good faith and in accordance with such opinion.
- (c) The General Partner shall have the right, in respect of any of its powers or obligations hereunder, to act through any of its duly authorized officers, a duly appointed attorney or attorneys-in-fact or the duly authorized officers of the Partnership. Each such attorney shall, to the extent provided by the General Partner in the power of attorney, have full

power and authority to do and perform each and every act and duty that is permitted or required to be done by the General Partner hereunder.

- (d) Any standard of care and duty imposed by this Agreement or under the Delaware Act or any applicable law, rule or regulation shall be modified, waived or limited, to the extent permitted by law, as required to permit the General Partner to act under this Agreement or any other agreement contemplated by this Agreement and to make any decision pursuant to the authority prescribed in this Agreement, so long as such action is reasonably believed by the General Partner to be in, or not inconsistent with, the best interests of the Partnership.
- 7.11 Reliance by Third Parties. Notwithstanding anything to the contrary in this Agreement, any Person dealing with the Partnership shall be entitled to assume that the General Partner and any officer of the General Partner authorized by the General Partner to act on behalf of and in the name of Partnership has full power and authority to encumber, sell or otherwise use in any manner any and all assets of the Partnership and to enter into any authorized contracts on behalf of the Partnership, and such Person shall be entitled to deal with the General Partner or any such officer as if it were the Partnership's sole party in interest, both legally and beneficially. The Limited Partner hereby waives any and all defenses or other remedies that may be available against such Person to contest, negate or disaffirm any action of the General Partner or any such officer in connection with any such dealing. In no event shall any Person dealing with the General Partner or any such officer or its representatives been complied with or to inquire into the necessity or expedience of any act or action of the General Partner or any such officer or its representatives. Each and every certificate, document or other instrument executed on behalf of the Partnership by the General Partner or its representatives shall be conclusive evidence in favor of any and every Person relying thereon or claiming thereunder that (i) at the time of the execution and delivery of such certificate, document or instrument was duly authorized and empowered to do so for and on behalf of the Partnership and (iii) such certificate, document or instrument was duly executed and delivered in accordance with the terms and provisions of this Agreement and is binding upon the Partnership.

# ARTICLE VIII BOOKS, RECORDS, ACCOUNTING AND REPORTS

8.1 *Records and Accounting.* The General Partner shall keep or cause to be kept at the principal office of the Partnership appropriate books and records with respect to the Partnership's business, including all books and records necessary to provide to the Limited Partner any information required to be provided pursuant to Section 3.3(a). Any books and records maintained by or on behalf of the Partnership in the regular course of its business, including books of account and records of Partnership proceedings, may be kept on, or be in the form of, computer disks, hard drives, punch cards, magnetic tape, photographs, micrographics or any other information storage device, provided, that the books and records so maintained are convertible into clearly legible written form within a reasonable period of time. The books of the Partnership shall be maintained, for financial reporting purposes, on an accrual basis in accordance with U.S. GAAP.

8.2 *Fiscal Year*. The fiscal year of the Partnership shall be a fiscal year ending December 31.

# ARTICLE IX TAX MATTERS

9.1 **Preparation of Tax Returns.** The Partnership shall timely file all returns of the Partnership that are required for federal, state and local income tax purposes on the basis of the accrual method and a taxable year ending on December 31. The tax information reasonably required by the Partners for federal and state income tax reporting purposes with respect to a taxable year shall be furnished to them within 90 days of the close of the calendar year in which the Partnership's taxable year ends. The classification, realization and recognition of income and deductions and other items shall be on the accrual method of accounting for federal income tax purposes.

#### 9.2 Tax Elections.

- (a) The Partnership shall make the election under Section 754 of the Code in accordance with applicable regulations thereunder, subject to the reservation of the right to seek to revoke any such election upon the General Partner's determination that such revocation is in the best interests of the Limited Partner.
- (b) The Partnership shall elect to deduct expenses incurred in organizing the Partnership ratably over a sixty-month period as provided in Section 709 of the Code.
- (c) Except as otherwise provided herein, the General Partner shall determine whether the Partnership should make any other elections permitted by the Code.
- 9.3 *Tax Controversies.* Subject to the provisions hereof, the General Partner is designated as the Tax Matters Partner (as defined in Section 6231 of the Code) and is authorized and required to represent the Partnership (at the Partnership's expense) in connection with all examinations of the Partnership's affairs by tax authorities, including resulting administrative and judicial proceedings, and to expend Partnership funds for professional services and costs associated therewith. Each Partner agrees to cooperate with the General Partner and to do or refrain from doing any or all things reasonably required by the General Partner to conduct such proceedings.
- 9.4 *Withholding*. Notwithstanding any other provision of this Agreement, the General Partner is authorized to take any action that it determines in its discretion to be necessary or appropriate to cause the Partnership to comply with any withholding requirements established under the Code or any other federal, state or local law including, without limitation, pursuant to Sections 1441, 1442, 1445 and 1446 of the Code. To the extent that the Partnership is required or elects to withhold and pay over to any taxing authority any amount resulting from the allocation or distribution of income to any Partner (including, without limitation, by reason of Section 1446 of the Code), the amount withheld may be treated as a distribution of cash pursuant to Section 6.3 in the amount of such withholding from such Partner.

# ARTICLE X ADMISSION OF PARTNERS

- 10.1 *Admission of the General Partner and Limited Partner*. As provided herein, the General Partner was previously admitted and shall continue as the sole general partner of the Partnership, and the MLP was previously admitted and shall continue as the sole limited partner of the Partnership.
- 10.2 Admission of Substituted Limited Partners. Any person that is the successor in interest to the Limited Partner as described in Section 4.3 shall be admitted to the Partnership as the Limited Partner upon (a) furnishing to the General Partner (i) acceptance in form satisfactory to the General Partner of all of the terms and conditions of this Agreement and (ii) such other documents or instruments as may be required to effect its admission as the Limited Partner in the Partnership and (b) obtaining the consent of the General Partner, which consent may be given or withheld in the General Partner's sole discretion. Such Person shall be admitted to the Partnership as the Limited Partner immediately prior to the transfer of the Partnership Interest, and the business of the Partnership shall continue without dissolution.
- 10.3 **Admission of a Successor or Transferee General Partner.** A successor General Partner approved pursuant to Section 11.1 or 11.2 or the transferee of or successor to all of the General Partner's Partnership Interest as the general partner in the Partnership pursuant to Section 4.2 who is proposed to be admitted as a successor General Partner shall, subject to compliance with the terms of Section 11.3, if applicable, be admitted to the Partnership as the General Partner, effective immediately prior to the withdrawal or removal of the General Partner pursuant to Section 11.1 or 11.2 or the transfer of the General Partner's Partnership Interest as a general partner in the Partnership pursuant to Section 4.2; provided, however, that no such successor shall be admitted to the Partnership until compliance with the terms of Section 4.2 has occurred and such successor has executed and delivered such other documents or instruments as may be required to effect such admission. Any such successor shall, subject to the terms hereof, carry on the business of the Partnership without dissolution.

### 10.4 Admission of Additional Limited Partners.

- (a) A Person (other than the General Partner, the MLP or a Substituted Limited Partner) who makes a Capital Contribution to the Partnership in accordance with this Agreement shall be admitted to the Partnership as an Additional Limited Partner only upon furnishing to the General Partner (i) evidence of acceptance in form satisfactory to the General Partner of all of the terms and conditions of this Agreement, including the granting of the power of attorney granted in Section 2.6 and (ii) such other documents or instruments as may be required in the discretion of the General Partner to effect such Person's admission as an Additional Limited Partner.
- (b) Notwithstanding anything to the contrary in this Section 10.4, no Person shall be admitted as an Additional Limited Partner without the consent of the General Partner, which consent may be given or withheld in the General Partner's discretion. The admission of any Person as an Additional Limited Partner shall become effective on the date upon which the

name of such Person is recorded as such in the books and records of the Partnership, following the consent of the General Partner to such admission.

10.5 **Amendment of Agreement and Certificate of Limited Partnership.** To effect the admission to the Partnership of any Partner, the General Partner shall take all steps necessary and appropriate under the Delaware Act to amend the records of the Partnership to reflect such admission and, if necessary, to prepare as soon as practicable an amendment to this Agreement and, if required by law, the General Partner shall prepare and file an amendment to the Certificate of Limited Partnership, and the General Partner may for this purpose, among others, exercise the power of attorney granted pursuant to Section 2.6.

# ARTICLE XI WITHDRAWAL OR REMOVAL OF PARTNERS

#### 11.1 Withdrawal of the General Partner.

- (a) The General Partner shall be deemed to have withdrawn from the Partnership upon the occurrence of any one of the following events (each such event herein referred to as an "Event of Withdrawal");
  - (i) the General Partner voluntarily withdraws from the Partnership by giving written notice to the other Partners;
  - (ii) the General Partner transfers all of its rights as General Partner pursuant to Section 4.2;
  - (iii) the General Partner is removed pursuant to Section 11.2;
  - (iv) [Reserved]
- (v) the General Partner (A) makes a general assignment for the benefit of creditors; (B) files a voluntary bankruptcy petition for relief under Chapter 7 of the United States Bankruptcy Code; (C) files a petition or answer seeking for itself a liquidation, dissolution or similar relief (but not a reorganization) under any law; (D) files an answer or other pleading admitting or failing to contest the material allegations of a petition filed against the General Partner in a proceeding of the type described in clauses (A)-(C) of this Section 11.1(a)(v); or (E) seeks, consents to or acquiesces in the appointment of a trustee (but not a debtor in possession), receiver or liquidator of the General Partner or of all or any substantial part of its properties;
- (vi) a final and non-appealable order of relief under Chapter 7 of the United States Bankruptcy Code is entered by a court with appropriate jurisdiction pursuant to a voluntary or involuntary petition by or against the General Partner; or
- (vii) (A) in the event the General Partner is a corporation, a certificate of dissolution or its equivalent is filed for the General Partner, or 90 days expire after the date of notice to the General Partner of revocation of its charter without a reinstatement of its charter, under the laws of its state of incorporation; (B) in the event the General Partner is a partnership

or limited liability company, the dissolution and commencement of winding up of the General Partner; (C) in the event the General Partner is acting in such capacity by virtue of being a trustee of the trust, the termination of the trust; (D) in the event the General Partner is a natural person, his death or adjudication of incompetency; (E) and otherwise in the event of the termination of the General Partner.

If an Event of Withdrawal specified in Section 11.1(a)(iv) (with respect to withdrawal), (v), (vi) or (vii)(A), (B), (C) or (E) occurs, the withdrawing General Partner shall give notice to the Limited Partner within 30 days after such occurrence. The Partners hereby agree that only the Events of Withdrawal described in this Section 11.1 shall result in the withdrawal of the General Partner from the Partnership.

- (b) Withdrawal of the General Partner from the Partnership upon the occurrence of an Event of Withdrawal shall not constitute a breach of this Agreement under the following circumstances: (i) at any time during the period beginning on the Closing Date and ending at 12:00 midnight, Eastern Standard Time, on December 31, 2008, the General Partner voluntarily withdraws by giving at least 90 days' advance notice of its intention to withdraw to the Limited Partner; provided, that prior to the effective date of such withdrawal, the Limited Partner approves such withdrawal and the General Partner delivers to the Partnership an Opinion of Counsel ("Withdrawal Opinion of Counsel") that such withdrawal (following the selection of the successor General Partner) would not result in the loss of the limited liability of the Limited Partner, any limited partner of the MLP, or any limited partner of any Group Member or cause the MLP or the Partnership to be treated as an association taxable as a corporation or otherwise to be taxed as an entity for federal income tax purposes; (ii) at any time after 12:00 midnight, Eastern Standard Time, on December 31, 2008, the General Partner voluntarily withdraws by giving at least 90 days' advance notice to the Limited Partner, such withdrawal to take effect on the date specified in such notice; or (iii) at any time that the General Partner ceases to be the General Partner pursuant to Section 11.1(a)(ii), (iii) or (iv). If the General Partner gives a notice of withdrawal pursuant to Section 11.1(a)(i) hereof, the Limited Partners may, prior to the effective date of such withdrawal, elect a successor General Partner. If, prior to the effective date of the General Partner's withdrawal, a successor is not selected by the Limited Partner as provided herein or the Partnership does not receive a Withdrawal Opinion of Counsel, the Partnership shall be dissolved in accordance with Section 12.1. Any successor General Partner elected in accordance with the terms of this Section 11.1 shall be subjec
- 11.2 *Removal of the General Partner.* The General Partner may be removed by the MLP, as Limited Partner. If the General Partner is removed pursuant to this Section 11.2, the MLP, as Limited Partner, may, prior to the effective date of such removal, elect a successor General Partner. The admission of any such successor General Partner to the Partnership shall be subject to the provisions of Section 10.3.

#### 11.3 Interest of Departing Partner and Successor General Partner.

(a) The Partnership Interest of a Departing Partner departing as a result of withdrawal or removal pursuant to Section 11.1 or 11.2 shall be purchased by the successor to the Departing Partner for cash in the manner specified in the MLP Agreement. Such purchase

shall be a condition to the admission to the Partnership of the successor as the General Partner. Notwithstanding the foregoing, an assignment of all or any portion of a General Partner's (or Departing General Partner's) Partnership Interest to the MLP, as Limited Partner, or to any other Person (other than an individual) the ownership interest of which is then transferred to the MLP, can be made in exchange for an increased interest in the MLP and in lieu of a cash purchase. Any successor General Partner shall indemnify the Departing General Partner as to all debts and liabilities of the Partnership arising on or after the effective date of the withdrawal or removal of the Departing Partner.

- (b) The Departing Partner shall be entitled to receive all reimbursements due such Departing Partner pursuant to Section 7.4, including any employeerelated liabilities (including severance liabilities), incurred in connection with the termination of any employees employed by such Departing Partner for the benefit of the Partnership.
- 11.4 *Withdrawal of the Limited Partner*. Without the prior written consent of the General Partner, which may be granted or withheld in its sole discretion, and except as provided in Section 10.1, the Limited Partner shall not have the right to withdraw from the Partnership.

# ARTICLE XII DISSOLUTION AND LIQUIDATION

- 12.1 *Dissolution.* The Partnership shall not be dissolved by the admission of Substituted Limited Partners or Additional Limited Partners or by the admission of a successor General Partner in accordance with the terms of this Agreement. Upon the removal or withdrawal of the General Partner, if a successor General Partner is elected pursuant to Section 11.1 or 11.2, the Partnership shall not be dissolved and such successor General Partner shall continue the business of the Partnership. The Partnership shall dissolve, and (subject to Section 12.2) its affairs shall be wound up, upon:
  - (a) the expiration of its term as provided in Section 2.7;
- (b) an Event of Withdrawal of the General Partner as provided in Section 11.1(a) (other than Section 11.1(a)(ii)), unless a successor is elected and an Opinion of Counsel is received as provided in Section 11.1(b) or 11.2 and such successor is admitted to the Partnership pursuant to Section 10.3;
  - (c) an election to dissolve the Partnership by the General Partner that is approved by the Limited Partner;
  - (d) entry of a decree of judicial dissolution of the Partnership pursuant to the provisions of the Delaware Act;
  - (e) the sale of all or substantially all of the assets and properties of the Partnership Group; or
  - (f) the dissolution of the MLP.

- 12.2 Continuation of the Business of the Partnership After Dissolution. Upon (a) dissolution of the Partnership following an Event of Withdrawal caused by the withdrawal or removal of the General Partner as provided in Section 11.1(a)(i) or (iii) and the failure of the Limited Partner to select a successor to such Departing Partner pursuant to Section 11.1 or 11.2, then within 90 days thereafter or (b) dissolution of the Partnership upon an event constituting an Event of Withdrawal pursuant to Section 11.1(a)(iv), (v) or (vi) of the MLP Agreement, then, to the maximum extent permitted by law, within 180 days thereafter, the Limited Partner may elect to reconstitute the Partnership and continue its business on the same terms and conditions set forth in this Agreement by forming a new limited partnership on terms identical to those set forth in this Agreement and having as a general partner a Person approved by the Limited Partner. In addition, upon dissolution of the Partnership pursuant to Section 12.1(f), if the MLP is reconstituted pursuant to Section 12.2 of the MLP Agreement, the reconstituted MLP may (whether or not it is the sole limited partner), within 180 days after such event of dissolution, as Limited Partner, elect to reconstitute the Partnership in accordance with the immediately preceding sentence. Upon any such election by the Limited Partner, all Partners shall be bound thereby and shall be deemed to have approved same. Unless such an election is made within the applicable time period as set forth above, the Partnership shall conduct only activities necessary to wind up its affairs. If such an election is so made, then:
- (i) the reconstituted Partnership shall continue until the end of the term set forth in Section 2.7 unless earlier dissolved in accordance with this Article XII;
- (ii) if the successor General Partner is not the former General Partner, then the interest of the former General Partner shall be purchased by the successor General Partner or converted into Common Units or purchased for cash by the MLP as provided in the MLP Agreement; and
- (iii) all necessary steps shall be taken to cancel this Agreement and the Certificate of Limited Partnership and to enter into and, as necessary, to file a new partnership agreement and certificate of limited partnership, and the successor General Partner may for this purpose exercise the powers of attorney granted the General Partner pursuant to Section 2.6; provided, that the right to approve a successor General Partner and to reconstitute and to continue the business of the Partnership shall not exist and may not be exercised unless the Partnership has received an Opinion of Counsel that (x) the exercise of the right would not result in the loss of limited liability of the Limited Partner or any limited partner of the MLP and (y) neither the Partnership, the reconstituted limited partnership nor any Group Member would be treated as an association taxable as a corporation or otherwise be taxable as an entity for federal income tax purposes upon the exercise of such right to continue.
- 12.3 *Liquidator*. Upon dissolution of the Partnership, unless the Partnership is continued under an election to reconstitute and continue the Partnership pursuant to Section 12.2, the General Partner shall select one or more Persons to act as Liquidator. The Liquidator (if other than the General Partner) shall be entitled to receive such compensation for its services as may be approved by the Limited Partner. The Liquidator (if other than the General Partner) shall agree not to resign at any time without 15 days' prior notice and may be removed at any time, with or without cause, by notice of removal approved by the Limited Partner. Upon dissolution, removal or resignation of the Liquidator, a successor and substitute Liquidator (who shall have

and succeed to all rights, powers and duties of the original Liquidator) shall within 30 days thereafter be approved by the Limited Partner. The right to approve a successor or substitute Liquidator in the manner provided herein shall be deemed to refer also to any such successor or substitute Liquidator approved in the manner herein provided. Except as expressly provided in this Article XII, the Liquidator approved in the manner provided herein shall have and may exercise, without further authorization or consent of any of the parties hereto, all of the powers conferred upon the General Partner under the terms of this Agreement (but subject to all of the applicable limitations, contractual and otherwise, upon the exercise of such powers, other than the limitation on sale set forth in Section 7.3(b)) to the extent necessary or desirable in the good faith judgment of the Liquidator to carry out the duties and functions of the Liquidator hereunder for and during such period of time as shall be reasonably required in the good faith judgment of the Liquidator to complete the winding up and liquidation of the Partnership as provided for herein.

- 12.4 *Liquidation*. The Liquidator shall proceed to dispose of the assets of the Partnership, discharge its liabilities, and otherwise wind up its affairs in such manner and over such period as the Liquidator determines to be in the best interest of the Partners, subject to Section 17-804 of the Delaware Act and the following:
- (a) Disposition of Assets. The assets may be disposed of by public or private sale or by distribution in kind to one or more Partners on such terms as the Liquidator and such Partner or Partners may agree. If any property is distributed in kind, the Partner receiving the property shall be deemed for purposes of Section 12.4(c) to have received cash equal to its fair market value; and contemporaneously therewith, appropriate cash distributions must be made to the other Partners. The Liquidator may, in its absolute discretion, defer liquidation or distribution of the Partnership's assets for a reasonable time if it determines that an immediate sale or distribution of all or part of the Partnership's assets would be impractical or would cause undue loss to the Partners. The Liquidator may, in its absolute discretion, distribute the Partnership's assets, in whole or in part, in kind if it determines that a sale would be impractical or would cause undue loss to the Partners.
- (b) Discharge of Liabilities. Liabilities of the Partnership include amounts owed to Partners otherwise than in respect of their distribution rights under Article VI. With respect to any liability that is contingent, conditional or unmatured or is otherwise not yet due and payable, the Liquidator shall either settle such claim for such amount as it thinks appropriate or establish a reserve of cash or other assets to provide for its payment. When paid, any unused portion of the reserve shall be distributed as additional liquidation proceeds.
- (c) Liquidation Distributions. All property and all cash in excess of that required to discharge liabilities as provided in Section 12.4(b) shall be distributed to the Partners in accordance with, and to the extent of, the positive balances in their respective Capital Accounts, as determined after taking into account all Capital Account adjustments (other than those made by reason of distributions pursuant to this Section 12.4(c)) for the taxable year of the Partnership during which the liquidation of the Partnership occurs (with such date of occurrence being determined pursuant to Treasury Regulation, Section 1.704-l(b)(2)(ii)(g)), and such distribution shall be made by the end of such taxable year (or, if later, within 90 days after said date of such occurrence).

- 12.5 *Cancellation of Certificate of Limited Partnership.* Upon the completion of the distribution of Partnership cash and property as provided in Section 12.4 in connection with the liquidation of the Partnership, the Partnership shall be terminated and the Certificate of Limited Partnership and all qualifications of the Partnership as a foreign limited partnership in jurisdictions other than the State of Delaware shall be canceled and such other actions as may be necessary to terminate the Partnership shall be taken.
- 12.6 *Return of Capital Contributions*. The General Partner shall not be personally liable for, and shall have no obligation to contribute or loan any monies or property to the Partnership to enable it to effectuate, the return of the Capital Contributions of the Limited Partner, or any portion thereof, it being expressly understood that any such return shall be made solely from Partnership assets.
  - 12.7 Waiver of Partition. To the maximum extent permitted by law, each Partner hereby waives any right to partition of the Partnership property.
- 12.8 *Capital Account Restoration*. No Partner shall have any obligation to restore any negative balance in its Capital Account upon liquidation of the Partnership.

#### ARTICLE XIII AMENDMENT OF PARTNERSHIP AGREEMENT

- 13.1 *Amendment to be Adopted Solely by the General Partner*. Each Partner agrees that the General Partner, without the approval of any Partner or Assignee, may amend any provision of this Agreement, and execute, swear to, acknowledge, deliver, file and record whatever documents may be required in connection therewith, to reflect:
- (a) a change in the name of the Partnership, the location of the principal place of business of the Partnership, the registered agent of the Partnership or the registered office of the Partnership;
  - (b) admission, substitution, withdrawal or removal of Partners in accordance with this Agreement;
- (c) a change that, in the sole discretion of the General Partner, is necessary or advisable to qualify or continue the qualification of the Partnership as a limited partnership or a partnership in which the limited partners have limited liability under the laws of any state or to ensure that no Group Member nor the MLP will be treated as an association taxable as a corporation or otherwise be taxable as an entity for federal income tax purposes;
- (d) a change that, in the discretion of the General Partner, (i) does not adversely affect the Limited Partner in any material respect, (ii) is necessary or advisable to satisfy any requirements, conditions or guidelines contained in any opinion, directive, order, ruling or regulation of any federal or state agency or judicial authority or contained in any federal or state statute (including the Delaware Act), compliance with any of which the General Partner determines in its discretion to be in the best interests of the Partnership and the Limited Partner, (iii) is required to effect the intent expressed in the Registration Statement or the intent of the provisions of this Agreement or is otherwise contemplated by this Agreement or (iv) is required

to conform the provisions of this Agreement with the provisions of the MLP Agreement as the provisions of the MLP Agreement may be amended, supplemented or restated from time to time;

- (e) a change in the fiscal year or taxable year of the Partnership and any changes that, in the discretion of the General Partner, are necessary or advisable as a result of a change in the fiscal year or taxable year of the Partnership including, if the General Partner shall so determine, a change in the definition of "Quarter" and the dates on which distributions are to be made by the Partnership;
- (f) an amendment that is necessary, in the Opinion of Counsel, to prevent the Partnership or the General Partner or its directors, officers, trustees or agents from in any manner being subjected to the provisions of the Investment Company Act of 1940, as amended, the Investment Advisers Act of 1940, as amended, or "plan asset" regulations adopted under the Employee Retirement Income Security Act of 1974, as amended, regardless of whether such are substantially similar to plan asset regulations currently applied or proposed by the United States Department of Labor;
  - (g) any amendment expressly permitted in this Agreement to be made by the General Partner acting alone;
  - (h) an amendment effected, necessitated or contemplated by a Merger Agreement approved in accordance with Section 14.3;
- (i) an amendment that, in the discretion of the General Partner, is necessary or advisable to reflect, account for and deal with appropriately the formation by the Partnership of, or investment by the Partnership in, any corporation, partnership, joint venture, limited liability company or other entity in connection with the conduct by the Partnership of activities permitted by the terms of Section 2.4;
  - (j) a merger or conveyance pursuant to Section 14.3(d); or
  - (k) any other amendments substantially similar to the foregoing.
- 13.2 *Amendment Procedures*. Except as provided in Section 13.1, all amendments to this Agreement shall be made in accordance with the following requirements: Amendments to this Agreement may be proposed only by or with the consent of the General Partner which consent may be given or withheld in its sole discretion. A proposed amendment shall be effective upon its approval by the Limited Partner.

#### ARTICLE XIV MERGER

14.1 *Authority*. The Partnership may merge or consolidate with one or more corporations, limited liability companies, business trusts or associations, real estate investment trusts, common law trusts or unincorporated businesses, including a general partnership or limited partnership, formed under the laws of the State of Delaware or any other state of the United States of America, pursuant to a written agreement of merger or consolidation ("Merger Agreement") in accordance with this Article XIV.

- 14.2 *Procedure for Merger or Consolidation.* Merger or consolidation of the Partnership pursuant to this Article XIV requires the prior approval of the General Partner. If the General Partner shall determine, in the exercise of its discretion, to consent to the merger or consolidation, the General Partner shall approve the Merger Agreement, which shall set forth:
  - (a) The names and jurisdictions of formation or organization of each of the business entities proposing to merge or consolidate;
- (b) The name and jurisdiction of formation or organization of the business entity that is to survive the proposed merger or consolidation (the "Surviving Business Entity");
  - (c) The terms and conditions of the proposed merger or consolidation;
- (d) The manner and basis of exchanging or converting the equity securities of each constituent business entity for, or into, cash, property or general or limited partner interests, rights, securities or obligations of the Surviving Business Entity; and (i) if any general or limited partner interests, securities or rights of any constituent business entity are not to be exchanged or converted solely for, or into, cash, property or general or limited partner interests, rights, securities or obligations of the Surviving Business Entity, the cash, property or general or limited partner interests, rights, securities or obligations of any limited partnership, corporation, trust or other entity (other than the Surviving Business Entity) which the holders of such general or limited partner interests, securities or rights are to receive in exchange for, or upon conversion of their general or limited partner interests, securities or rights, and (ii) in the case of securities represented by certificates, upon the surrender of such certificates, which cash, property or general or limited partner interests, rights, securities or obligations of the Surviving Business Entity or any general or limited partner interests or other entity (other than the Surviving Business Entity), or evidences thereof, are to be delivered;
- (e) A statement of any changes in the constituent documents or the adoption of new constituent documents (the articles or certificate of incorporation, articles of trust, declaration of trust, certificate or agreement of limited partnership or other similar charter or governing document) of the Surviving Business Entity to be effected by such merger or consolidation;
- (f) The effective time of the merger, which may be the date of the filing of the certificate of merger pursuant to Section 14.4 or a later date specified in or determinable in accordance with the Merger Agreement (provided, that if the effective time of the merger is to be later than the date of the filing of the certificate of merger, the effective time shall be fixed no later than the time of the filing of the certificate of merger and stated therein); and
  - (g) Such other provisions with respect to the proposed merger or consolidation as are deemed necessary or appropriate by the General Partner.

#### 14.3 Approval by the Limited Partner of Merger or Consolidation.

(a) Except as provided in Section 14.3(d), the General Partner, upon its approval of the Merger Agreement, shall direct that a copy or a summary of the Merger Agreement be submitted to the Limited Partner for its approval.

- (b) Except as provided in Section 14.3(d) the Merger Agreement shall be approved upon receiving the approval of the Limited Partner.
- (c) Except as provided in Section 14.3(d), after such approval by the Limited Partner, and at any time prior to the filing of the certificate of merger pursuant to Section 14.4, the merger or consolidation may be abandoned pursuant to provisions therefor, if any, set forth in the Merger Agreement.
- (d) Notwithstanding anything else contained in this Article XIV or in this Agreement, the General Partner is permitted, in its discretion and without Limited Partner approval, to (i) convert the Partnership or any Group Member to another type of limited liability entity as provided by Section 17-219 of the Delaware Act or (ii) merge the Partnership or any Group Member into, or convey all of the Partnership's assets to, another limited liability entity which shall be newly formed and shall have no assets, liabilities or operations at the time of such merger or conveyance other than those it receives from the Partnership or other Group Member, provided that in any such case (A) General Partner has received an Opinion of Counsel that the conversion, merger or conveyance, as the case may be, would not result in the loss of the limited liability of the Limited Partner or of any limited partner of the MLP or cause the Partnership or the MLP to be treated as an association taxable as a corporation or otherwise to be taxed as an entity for federal income tax purposes (to the extent not previously treated as such), (B) the sole purpose of such conversion, merger or conveyance is to effect a mere change in the legal form of the Partnership into another limited liability entity and (C) the governing instruments of the new entity provide the Limited Partner with rights and obligations that are, in all material respects, the same rights and obligations of the Limited Partner hereunder.
- 14.4 *Certificate of Merger.* Upon the required approval by the General Partner and the Limited Partner of a Merger Agreement, a certificate of merger shall be executed and filed with the Secretary of State of the State of Delaware in conformity with the requirements of the Delaware Act.

#### 14.5 Effect of Merger.

- (a) At the effective time of the certificate of merger:
- (i) all of the rights, privileges and powers of each of the business entities that has merged or consolidated, and all property, real, personal and mixed, and all debts due to any of those business entities and all other things and causes of action belonging to each of those business entities shall be vested in the Surviving Business Entity and after the merger or consolidation shall be the property of the Surviving Business Entity to the extent they were of each constituent business entity;
- (ii) the title to any real property vested by deed or otherwise in any of those constituent business entities shall not revert and is not in any way impaired because of the merger or consolidation;
- (iii) all rights of creditors and all liens on or security interests in property of any of those constituent business entities shall be preserved unimpaired; and

- (iv) all debts, liabilities and duties of those constituent business entities shall attach to the Surviving Business Entity, and may be enforced against it to the same extent as if the debts, liabilities and duties had been incurred or contracted by it.
- (b) A merger or consolidation effected pursuant to this Article XIV shall not be deemed to result in a transfer or assignment of assets or liabilities from one entity to another.

#### ARTICLE XV GENERAL PROVISIONS

- 15.1 *Addresses and Notices.* Any notice, demand, request, report or proxy materials required or permitted to be given or made to a Partner under this Agreement shall be in writing and shall be deemed given or made when received by it at the principal office of the Partnership referred to in Section 2.3.
  - 15.2 References. Except as specifically provided as otherwise, references to "Articles" and "Sections" are to Articles and Sections of this Agreement.
- 15.3 *Further Action.* The parties shall execute and deliver all documents, provide all information and take or refrain from taking action as may be necessary or appropriate to achieve the purposes of this Agreement.
- 15.4 *Binding Effect.* This Agreement shall be binding upon and inure to the benefit of the parties hereto and their heirs, executors, administrators, successors, legal representatives and permitted assigns.
- 15.5 *Integration*. This Agreement constitutes the entire agreement among the parties hereto pertaining to the subject matter hereof and supersedes all prior agreements and understandings pertaining thereto.
  - 15.6 *Creditors*. None of the provisions of this Agreement shall be for the benefit of, or shall be enforceable by, any creditor of the Partnership.
- 15.7 *Waiver.* No failure by any party to insist upon the strict performance of any covenant, duty, agreement or condition of this Agreement or to exercise any right or remedy consequent upon a breach thereof shall constitute waiver of any such breach of any other covenant, duty, agreement or condition.
- 15.8 *Counterparts*. This Agreement may be executed in counterparts, all of which together shall constitute an agreement binding on all the parties hereto, notwithstanding that all such parties are not signatories to the original or the same counterpart. Each party shall become bound by this Agreement immediately upon affixing its signature hereto, independently of the signature of any other party.
- 15.9 *Applicable Law.* This Agreement shall be construed in accordance with and governed by the laws of the State of Delaware, without regard to the principles of conflicts of law.

15.10 *Invalidity of Provisions*. If any provision of this Agreement is or becomes invalid, illegal or unenforceable in any respect, the validity, legality and enforceability of the remaining provisions contained herein shall not be affected thereby.

15.11 *Amendments to Reflect GP Reorganization Agreement.* In addition to the amendments to this Agreement contained in the GP Reorganization Agreement and notwithstanding any other provision of this Agreement to the contrary, this Agreement shall be deemed to be further amended and modified to the extent necessary, but only to the extent necessary, to carry out the purposes and intent of the GP Reorganization Agreement.

IN WITNESS WHEREOF, the parties hereto have executed this Agreement as of the date first written above.

### **GENERAL PARTNER:**

ENTERPRISE PRODUCTS OLPGP, INC.

By: /s/ Michael A. Creel
Name: Michael A. Creel
Title: Executive Vice President

# LIMITED PARTNER:

### ENTERPRISE PRODUCTS PARTNERS L.P.

By: Enterprise Products GP, LLC, as

General Partner

By: /s/ Michael A. Creel
Name: Michael A. Creel
Title: Executive Vice President

- 42 -

#### Attachment I

#### **DEFINITIONS**

"Additional Limited Partner" means a Person admitted to the Partnership as a Limited Partner pursuant to Section 10.4 and who is shown as such on the books and records of the Partnership.

"Adjusted Capital Account" means the Capital Account maintained for each Partner as of the end of each fiscal year of the Partnership, (a) increased by any amounts that such Partner is obligated to restore under the standards set by Treasury Regulation Section 1.704-l(b)(2)(ii)(c) (or is deemed obligated to restore under Treasury Regulation Sections 1.704-2(g) and 1.704-2(j)(5)) and (b)decreased by (i) the amount of all losses and deductions that, as of the end of such fiscal year, are reasonably expected to be allocated to such Partner in subsequent years under Sections 704(e)(2) and 706(d) of the Code and Treasury Regulation Section 1.751-l(b)(2)(ii), and (ii) the amount of all distributions that, as of the end of such fiscal year, are reasonably expected to be made to such Partner in subsequent years in accordance with the terms of this Agreement or otherwise to the extent they exceed offsetting increases to such Partner's Capital Account that are reasonably expected to occur during (or prior to) the year in which such distributions are reasonably expected to be made (other than increases as a result of a minimum gain chargeback pursuant to Section 6.1(d)(i) or 6.1(d)(ii)). The foregoing definition of Adjusted Capital Account is intended to comply with the provisions of Treasury Regulation Section 1.704-l(b)(2)(ii)(d) and shall be interpreted consistently therewith. The "Adjusted Capital Account" of a Partner in respect of a general partner interest or any other specified interest in the Partnership shall be the amount which such adjusted capital account would be if such general partner interest or other interest in the Partnership were the only interest in the Partnership held by a Partner from and after the date on which such general partner interest or other interest was first issued.

"Adjusted Property" means any property the Carrying Value of which has been adjusted pursuant to Section 5.4(d)(i) or 5.4(d)(ii). Once an Adjusted Property is deemed contributed to a new partnership in exchange for an interest in the new partnership, followed by the deemed liquidation of the Partnership for federal income tax purposes upon a termination of the Partnership pursuant to Treasury Regulation Section 1.708-(b)(l)(iv), such property shall thereafter constitute a Contributed Property until the carrying value of such property is subsequently adjusted pursuant to Section 5.4(d)(i) or 5.4(d)(ii).

"Affiliate" means, with respect to any Person, any other Person that directly or indirectly through one or more intermediaries controls, is controlled by or is under common control with (either directly or indirectly), the Person in question. As used herein, the term "control" means the possession, direct or indirect, of the power to direct or cause the direction of the management and policies of a Person, whether through ownership of voting securities, by contract or otherwise.

"Agreed Allocation" means any allocation, other than a Required Allocation, of an item of income, gain, loss or deduction pursuant to the provisions of Section 6.1, including, without limitation, a Curative Allocation (if appropriate to the context in which the term "Agreed Allocation" is used).

"Agreed Value" of any Contributed Property means the fair market value of such property or other consideration at the time of contribution as determined by the General Partner using such reasonable method of valuation as it may adopt. The General Partner shall, in its discretion, use such method as it deems reasonable and appropriate to allocate the aggregate Agreed Value of Contributed Properties contributed to the Partnership in a single or integrated transaction among each separate property on a basis proportional to the fair market value of each Contributed Property.

"Agreement" means this Amended and Restated Agreement of Limited Partnership of Enterprise Products Operating L.P., as it may be amended, supplemented or restated from time to time.

"Assignee" means a Person to whom one or more Partnership Interests of a Limited Partner have been transferred in the manner permitted under this Agreement but who has not been admitted as a Substituted Limited Partner.

"Audit and Conflicts Committee" means a committee of the board of directors of the MLP General Partner composed entirely of two or more directors who are neither security holders, officers nor employees of the MLP General Partner nor officers, directors or employees of any Affiliate of such entity.

"Available Cash" means, with respect to any Quarter ending prior to the Liquidation Date,

- (a) the sum of (i) all cash and cash equivalents of the Partnership Group on hand at the end of such Quarter, and (ii) all additional cash and cash equivalents of the Partnership Group on hand on the date of determination of Available Cash with respect to such Quarter resulting from (A) borrowings for working capital purposes made subsequent to the end of such Quarter or (B) Interim Capital Transactions after the end of such Quarter designated by the General Partner as Operating Surplus in accordance with clause (a)(iii)(A) of the definition of Operating Surplus, less
- (b) the amount of any cash reserves that is necessary or appropriate in the reasonable discretion of the General Partner to (i) provide for the proper conduct of the business of the Partnership Group (including reserves for future capital expenditures and anticipated for the future credit needs of the Partnership Group) subsequent to such Quarter, (ii) comply with applicable law or any loan agreement, security agreement, mortgage, debt instrument or other agreement or obligation to which any member of the Partnership Group is a party or by which it is bound or its assets are subject or (iii) provide funds for distributions under Section 6.4 or 6.5 of the MLP Agreement in respect of any one or more of the next four Quarters; provided, however, that the General Partner may not establish cash reserves pursuant to (iii) above if the effect of such reserves would be that the MLP is unable to distribute the Minimum Quarterly Distribution on all Common Units with respect to such Quarter; and, provided further, that disbursements made by a Group Member or cash reserves established, increased or reduced after the end of such Quarter but on or before the date of determination of Available Cash with respect to such Quarter shall be deemed to have been made, established, increased or reduced, for

purposes of determining Available Cash, within such Quarter if the General Partner so determines.

Notwithstanding the foregoing, "Available Cash" with respect to the Quarter in which the Liquidation Date occurs and any subsequent Quarter shall equal zero.

"Book-Tax Disparity" means with respect to any item of Contributed Property or Adjusted Property, as of the date of any determination, the difference between the Carrying Value of such Contributed Property or Adjusted Property and the adjusted basis thereof for federal income tax purposes as of such date. A Partner's share of the Partnership's Book-Tax Disparities in all of its Contributed Property and Adjusted Property will be reflected by the difference between such Partner's Capital Account balance as maintained pursuant to Section 5.4 and the hypothetical balance of such Partner's Capital Account computed as if it had been maintained strictly in accordance with federal income tax accounting principles.

"Business Day" means Monday through Friday of each week, except that a legal holiday recognized as such by the government of the United States of America or the states of New York or Texas shall not be regarded as a Business Day.

"Capital Account" means the capital account maintained for a Partner pursuant to Section 5.4.

"Capital Contribution" means any cash, cash equivalents or the Net Agreed Value of Contributed Property that a Partner contributes to the Partnership pursuant to this Agreement.

"Carrying Value" means (a) with respect to a Contributed Property, the Agreed Value of such property reduced (but not below zero) by all depreciation, amortization and cost recovery deductions charged to the Partners' and Assignees' Capital Accounts in respect of such Contributed Property, and (b) with respect to any other Partnership property, the adjusted basis of such property for federal income tax purposes, all as of the time of the determination. The Carrying Value of any property shall be adjusted from time to time in accordance with Sections 5.4(d)(i) and 5.4(d)(ii) and to reflect changes, additions or other adjustments to the Carrying Value for dispositions and acquisitions of Partnership properties, as deemed appropriate by the General Partner.

"Certificate of Limited Partnership" means the Certificate of Limited Partnership of the Partnership filed with the Secretary of State of the State of Delaware as referenced in Section 2.1, as such Certificate of Limited Partnership may be amended, supplemented or restated from time to time.

"Closing Date" means the first date on which Common Units are sold by the MLP to the Underwriters pursuant to the provisions of the Underwriting Agreement.

"Code" means the Internal Revenue Code of 1986, as amended and in effect from time to time. Any reference herein to a specific section or sections of the Code shall be deemed to include a reference to any corresponding provision of future law.

"Common Unit" has the meaning assigned to such term in the MLP Agreement.

"Contributed Property" means each property or other asset, in such form as may be permitted by the Delaware Act, but excluding cash, contributed to the Partnership (or deemed contributed to the Partnership on termination and reconstitution thereof pursuant to Section 708 of the Code). Once the Carrying Value of a Contributed Property is adjusted pursuant to Section 5.4(d), such property shall no longer constitute a Contributed Property, but shall be deemed an Adjusted Property.

"Curative Allocation" means any allocation of an item of income, gain, deduction, loss or credit pursuant to the provisions of Section 6.1 (d) (ix).

"Delaware Act" means the Delaware Revised Uniform Limited Partnership Act, 6 Del C. ss.17-101, et seq., as amended, supplemented or restated from time to time, and any successor to such statute.

"Departing Partner" means a former General Partner from and after the effective date of any withdrawal or removal of such former General Partner pursuant to Section 11.1 or 11.2.

"Economic Risk of Loss" has the meaning set forth in Treasury Regulation Section 1.752-2(a).

"EPC" means Enterprise Products Company, a Delaware corporation.

"EPC PartnersII" means EPC PartnersII, Inc., a Delaware corporation.

"EPCO Agreement" has the meaning assigned to such term in the MLP Agreement.

"Event of Withdrawal" has the meaning assigned to such term in Section 11.1.

"First Amendment" means the First Amendment to Agreement of Limited Partnership of the Partnership dated effective June 1, 1998 among the MLP General Partner, the MLP, EPC and EPC Partners II.

"General Partner" means OLPGP and its successors and permitted assigns as general partner of the Partnership. References herein to actions taken by the General Partner prior to December 10, 2003, shall refer to actions taken by the MLP General Partner, as the predecessor of OLPGP as general partner of the Partnership.

"GP Reorganization Agreement" means the Reorganization Agreement, dated as of December 10, 2003 among the Partnership, the MLP, the General Partner and the MLP General Partner.

"Group Member" means a member of the Partnership Group.

"Indemnitee" means (a) the General Partner, any Departing Partner and any Person who is or was an Affiliate of the General Partner or any Departing Partner, (b) any Person who is or was a member, director, officer, employee, agent or trustee of the MLP or any Group Member, (c) any Person who is or was an officer, member, partner, director, employee, agent or trustee of the General Partner or any Departing Partner or any Affiliate of the General Partner or any

Departing Partner, or any Affiliate of any such Person and (d) any Person who is or was serving at the request of the General Partner or any Departing Partner or any such Affiliate as a director, officer, employee, member, partner, agent, fiduciary or trustee of another Person; provided, that a Person shall not be an Indemnitee by reason of providing, on a fee-for-services basis, trustee, fiduciary or custodial services.

"Initial Offering" means the initial offering and sale of Common Units to the public, as described in the Registration Statement.

"Interim Capital Transactions" has the meaning assigned to such term in the MLP Agreement.

"Limited Partner" means any Person that is admitted to the Partnership as a limited partner pursuant to the terms and conditions of this Agreement; but the term Limited Partner shall not include any Person from and after the time such Person withdraws as a Limited Partner from the Partnership.

"Liquidation Date" means (a) in the case of an event giving rise to the dissolution of the Partnership of the type described in clauses (a) and (b) of the first sentence of Section 12.2, the date on which the applicable time period during which the Partners have the right to elect to reconstitute the Partnership and continue its business has expired without such an election being made, and (b)in the case of any other event giving rise to the dissolution of the Partnership, the date on which such event occurs.

"Liquidator" means one or more Persons selected by the General Partner to perform the functions described in Section 12.3.

"Merger Agreement" has the meaning assigned to such term in Section 14.1.

"Minimum Quarterly Distribution" has the meaning assigned to such term in the MLP Agreement.

"MLP" has the meaning assigned to such term in the introductory paragraph hereof.

"MLP Agreement" means the Third Amended and Restated Agreement of Limited Partnership of the MLP, dated May 15, 2002, as amended, supplemented or restated from time to time.

"MLP General Partner" means Enterprise Products GP, LLC and its successors and permitted assigns in its capacity as general partner of the MLP.

"National Securities Exchange" means an exchange registered with the Commission under Section 6(a) of the Securities Exchange Act of 1934, as amended, supplemented or restated from time to time, and any successor to such statute, or the Nasdaq Stock Market or any successor thereto.

"Net Agreed Value" means, (a) in the case of any Contributed Property, the Agreed Value of such property reduced by any liabilities either assumed by the Partnership upon such

contribution or to which such property is subject when contributed, and (b) in the case of any property distributed to a Partner by the Partnership, the Partnership's Carrying Value of such property (as adjusted pursuant to Section 5.4(d)(ii)) at the time such property is distributed, reduced by any indebtedness either assumed by such Partner upon such distribution or to which such property is subject at the time of distribution, in either case, as determined under Section 752 of the Code.

"Net Income" means, for any taxable year, the excess, if any, of the Partnership's items of income and gain (other than those items taken into account in the computation of Net Termination Gain or Net Termination Loss) for such taxable year over the Partnership's items of loss and deduction (other than those items taken into account in the computation of Net Termination Gain or Net Termination Loss) for such taxable year. The items included in the calculation of Net Income shall be determined in accordance with Section 5.4(b) and shall not include any items specially allocated under Section 6.1(d).

"Net Loss" means, for any taxable year, the excess, if any, of the Partnership's items of loss and deduction (other than those items taken into account in the computation of Net Termination Gain or Net Termination Loss) for such taxable year over the Partnership's items of income and gain (other than those items taken into account in the computation of Net Termination Gain or Net Termination Loss) for such taxable year. The items included in the calculation of Net Loss shall be determined in accordance with Section 5.4(b) and shall not include any items specially allocated under Section 6.1(d).

"Net Termination Gain" means, for any taxable year, the sum, if positive, of all items of income, gain, loss or deduction recognized by the Partnership after the Liquidation Date. The items included in the determination of Net Termination Gain shall be determined in accordance with Section 5.4(b) and shall not include any items of income, gain or loss specially allocated under Section 6.1(d).

"Net Termination Loss" means, for any taxable year, the sum, if negative, of all items of income, gain, loss or deduction recognized by the Partnership after the Liquidation Date. The items included in the determination of Net Termination Loss shall be determined in accordance with Section 5.4(b) and shall not include any items of income, gain or loss specially allocated under Section 6.1(d).

"Nonrecourse Built-in Gain" means with respect to any Contributed Properties or Adjusted Properties that are subject to a mortgage or pledge securing a Nonrecourse Liability, the amount of any taxable gain that would be allocated to the Partners pursuant to Sections 6.2(b)(i)(A), 6.2(b)(ii)(A) and 6.2(b)(iii) if such properties were disposed of in a taxable transaction in full satisfaction of such liabilities and for no other consideration.

"Nonrecourse Deductions" means any and all items of loss, deduction or expenditures (described in Section 705(a)(2)(B) of the Code) that, in accordance with the principles of Treasury Regulation Section 1.704-2(b), are attributable to a Nonrecourse Liability.

"Nonrecourse Liability" has the meaning set forth in Treasury Regulation Section 1.752-l(a)(2).

- "OLPGP" means Enterprise Products OLPGP, Inc., a Delaware corporation.
- "OLP Subsidiary" means a Subsidiary of the Partnership.
- "Operating Surplus" has the meaning assigned to such term in the MLP Agreement.
- "Opinion of Counsel" means a written opinion of counsel (who may be regular counsel to the Partnership, the General Partner or any of their Affiliates) acceptable to the General Partner in its reasonable discretion.
  - "Over-Allotment Option" has the meaning assigned to such term in the MLP Agreement.
  - "Partner Nonrecourse Debt" has the meaning set forth in Treasury Regulation Section 1.704-2(b)(4).
  - "Partner Nonrecourse Debt Minimum Gain" has the meaning set forth in Treasury Regulation Section 1.704-2(i)(2).
- "Partner Nonrecourse Deductions" means any and all items of loss, deduction or expenditure (including, without limitation, any expenditure described in Section 705(a)(2)(B) of the Code) that, in accordance with the principles of Treasury Regulation Section 1.704-2(i), are attributable to a Partner Nonrecourse Debt.
  - "Partners" means the General Partner and the Limited Partner.
  - "Partnership" means Enterprise Products Operating L.P., a Delaware limited partnership, and any successors thereto.
  - "Partnership Group" means the Partnership and the OLP Subsidiaries, treated as a single consolidated entity.
  - "Partnership Interest" means the interest of a Partner in the Partnership.
  - "Partnership Minimum Gain" means that amount determined in accordance with the principles of Treasury Regulation Section 1.704-2(d).
  - "Partnership Securities" shall have the meaning given to such term in the MLP Agreement.
  - "Percentage Interest" means as of the date of such determination (a) as to the General Partner, 0.001% and (b) as to the MLP, 99.999%.
- "Person" means an individual or a corporation, limited liability company, partnership, joint venture, trust, unincorporated organization, association, government agency or political subdivision thereof or other entity.
  - "Quarter" means, unless the context requires otherwise, a fiscal quarter of the Partnership.

"Recapture Income" means any gain recognized by the Partnership (computed without regard to any adjustment required by Sections 734 or 743 of the Code) upon the disposition of any property or asset of the Partnership, which gain is characterized as ordinary income because it represents the recapture of deductions previously taken with respect to such property or asset.

"Registration Statement" means the Registration Statement on Form S-1 (Registration No.333-52537), as it has been or as it may be amended or supplemented from time to time, filed by the MLP with the Securities and Exchange Commission under the Securities Act to register the offering and sale of the Common Units in the Initial Offering.

"Required Allocations" means (a) any limitation imposed on any allocation of Net Losses or Net Termination Losses under Section 6.1(b) or 6.1(c)(ii) and (b) any allocation of an item of income, gain, loss or deduction pursuant to Section 6.1(d)(i), 6.1(d)(ii), 6.1(d)(iii), 6.1(d)(vi), 6.1(d)(vi) or 6.1(d)(viii).

"Residual Gain" or "Residual Loss" means any item of gain or loss, as the case may be, of the Partnership recognized for federal income tax purposes resulting from a sale, exchange or other disposition of a Contributed Property or Adjusted Property, to the extent such item of gain or loss is not allocated pursuant to Section 6.2(b)(i)(A) or 6.2(b)(ii)(A), respectively, to eliminate Book-Tax Disparities.

"Restricted Activities" has the meaning assigned to such term in the MLP Agreement.

"Securities Act" means the Securities Act of 1933, as amended, supplemented or restated from time to time and any successor to such statute.

"Special Approval" means approval by a majority of the members of the Audit and Conflicts Committee.

"Subordinated Unit" has the meaning assigned to such term in the MLP Agreement.

"Subordination Period" has the meaning assigned to such term in the MLP Agreement.

"Subsidiary" means, with respect to any Person, (a) a corporation of which more than 50% of the voting power of shares entitled (without regard to the occurrence of any contingency) to vote in the election of directors or other governing body of such corporation is owned, directly or indirectly, at the date of determination, by such Person, by one or more Subsidiaries of such Person or a combination thereof, (b) a partnership (whether general or limited) in which such Person or a Subsidiary of such Person is, at the date of determination, a general or limited partner of such partnership, but only if more than 50% of the Partnership Interests of such partnership (considering all of the Partnership Interests of the partnership as a single class) is owned, directly or indirectly, at the date of determination, by such Person, by one or more Subsidiaries of such Person or a combination thereof or (c) any other Person (other than a corporation or a partnership) in which such person, one or more Subsidiaries of such Person, or a combination thereof, directly or indirectly, at the date of determination has (i) at least a majority ownership interest or (ii) the power to elect or direct the election of a majority of the directors or other governing body of such Person.

"Substituted Limited Partner" means a Person who is admitted as a Limited Partner to the Partnership pursuant to Section 10.2 in place of and with all the rights of a Limited Partner and who is shown as a Limited Partner on the books and records of the Partnership.

"Surviving Business Entity" has the meaning assigned to such term in Section 14.2(b).

"Transfer" has the meaning assigned to such term in Section 4.1(a).

"Underwriter" means each Person named as an underwriter in Exhibit A to the Underwriting Agreement that purchases Common Units pursuant thereto.

"*Underwriting Agreement*" means the Underwriting Agreement dated July 27, 1998, among the Underwriters, the MLP, the General Partner and certain other parties, providing for the purchase of Common Units by such Underwriters.

"Unit" has the meaning assigned to such term in the MLP Agreement.

"Unrealized Gain" attributable to any item of Partnership property means, as of any date of determination, the excess, if any, of (a) the fair market value of such property as of such date (as determined under Section 5.4(d)) over (b) the Carrying Value of such property as of such date (prior to any adjustment to be made pursuant to Section 5.4(d) as of such date).

"Unrealized Loss" attributable to any item of Partnership property means, as of any date of determination, the excess, if any, of (a) the Carrying Value of such property as of such date (prior to any adjustment to be made pursuant to Section 5.4(d) as of such date) over (b) the fair market value of such property as of such date (as determined under Section 5.4(d)).

"U.S. GAAP" means United States Generally Accepted Accounting Principles consistently applied.

"Withdrawal Opinion of Counsel" has the meaning assigned to such term in Section 11.1(b).

### AMENDED AND RESTATED CREDIT AGREEMENT

dated as of June 29, 2005

among

### CAMERON HIGHWAY OIL PIPELINE COMPANY

as Borrower

#### THE LENDERS FROM TIME TO TIME PARTY HERETO

and

### SUNTRUST BANK,

as Administrative Agent and Collateral Agent

# SUNTRUST CAPITAL MARKETS, INC.,

as Arranger and Sole Book Manager

# TABLE OF CONTENTS

ARTICLE I	Page
ANTICLE I	
DEFINITIONS; CONSTRUCTION	1
Section 1.1. Definitions Section 1.2. Classifications of Loans and Borrowings Section 1.3. Accounting Terms and Determination Section 1.4. Terms Generally	1 22 22 22
ARTICLE II	
AMOUNT AND TERMS OF THE COMMITMENTS	23
Section 2.1. General Description of Facilities Section 2.2. Commitments Section 2.3. Funding of Borrowings Section 2.4. Interest Elections Section 2.5. Optional Reduction and Termination of Commitments Section 2.6. Repayment of Loans Section 2.7. Evidence of Indebtedness Section 2.8. Optional Prepayments Section 2.9. Mandatory Prepayments Section 2.10. Interest on Loans Section 2.11. Fees Section 2.12. Computation of Interest and Fees Section 2.13. Inability to Determine Interest Rates Section 2.14. Illegality Section 2.15. Increased Costs Section 2.16. Funding Indemnity Section 2.17. Taxes Section 2.18. Payments Generally; Pro Rata Treatment; Sharing of Set-offs Section 2.19. Mitigation of Obligations; Replacement of Lenders	23 23 24 25 25 25 26 26 26 27 28 28 28 28 29 30 30 32 33
ARTICLE III	
CONDITIONS PRECEDENT TO LOANS	33
Section 3.1. Conditions To Effectiveness Section 3.2. Delivery of Documents	33 37
ARTICLE IV	
REPRESENTATIONS AND WARRANTIES	37
Section 4.1. Organization; Powers Section 4.2. Authorization; Enforceability Section 4.3. Other Material Agreements Section 4.4. Governmental Approvals; Material Permits; No Conflicts Section 4.5. Financial Condition; No Material Adverse Change; Material Obligations	37 37 38 38 39
i	

	Page
Section 4.6. Properties	39
Section 4.7. Litigation and Environmental Matters	40
Section 4.8. Compliance with Laws and Agreements	40
Section 4.9. Investment Company Act Status	40
Section 4.10. Public Utility Holding Company Act Status	41
Section 4.11. Taxes	41
Section 4.12. ERISA	41
Section 4.13. Disclosure	41
Section 4.14. Business of the Borrower	42
Section 4.15. Subsidiaries	42
Section 4.16. OFAC	42
Section 4.17. Patriot Act	43
ARTICLE V	
AFFIRMATIVE COVENANTS	43
Section 5.1. Financial Statements and Other Information	43
Section 5.2. Notices of Material Events	44
Section 5.3. Existence; Conduct of Business	45
Section 5.4. Payment of Obligations	45
Section 5.5. Maintenance of Properties; Insurance	45
Section 5.6. Books and Records, Inspection Rights	47
Section 5.7. Compliance with Laws	48
Section 5.8. Use of Proceeds	48
Section 5.9. Environmental Matters	48
Section 5.10. Maintain Title to Collateral; Mortgages	48
Section 5.11. Further Assurance	49
Section 5.12. Performance of Transaction Documents	49
Section 5.13. New Subsidiaries	50
Section 5.15. Financial Covenants	50
Section 5.16. Valero/Valero LP Transaction	50
Section 5.17. Accounts	50
Section 5.18. Post-Closing Deliveries	50
ARTICLE VI	
NEGATIVE COVENANTS	51
Section 6.1. Indebtedness	51
Section 6.2. Liens	52
Section 6.3. Fundamental Changes	52
Section 6.4. Investments, Loans, Advances, Guarantees and Acquisitions	52
Section 6.5. Hedging Agreements	53
Section 6.6. Restricted Payments	53
Section 6.7. Transactions with Affiliates	53
Section 6.8. Restrictive Agreements	53
Section 6.10. Management Fees	53
Section 6.11. Subsidiaries	53
Section 6.12. Project Changes	54
Section 6.13. Compliance with ERISA	54
Section 6.14. Amendments	54
Section 6.15. Sale and Leasebacks	54

	Page
ARTICLE VII	
EVENTS OF DEFAULT	54
Section 7.1. Events of Default	54
Section 7.2. Application of Proceeds from Collateral	57
Section 7.3. Sponsor LOCs	58
ARTICLE VIII	
THE ADMINISTRATIVE AGENT	58
Section 8.1. Appointment of Administrative Agent	58
Section 8.2. Nature of Duties of Administrative Agent	59
Section 8.3. Lack of Reliance on the Administrative Agent	59
Section 8.4. Certain Rights of the Administrative Agent	60
Section 8.5. Reliance by Administrative Agent	60
Section 8.6. The Administrative Agent in its Individual Capacity	60
Section 8.7. Successor Administrative Agent	60
Section 8.8. Authorization to Execute other Loan Documents	61
ARTICLE IX	
THE COLLATERAL AGENT	61
Section 9.1. Appointment of Collateral Agent	61
Section 9.2. Nature of Duties of Collateral Agent	61
Section 9.3. Certain Rights of the Collateral Agent	62
Section 9.4. Reliance by Collateral Agent	62
Section 9.5. The Collateral Agent in its Individual Capacity	62
Section 9.6. Successor Collateral Agent	63
Section 9.7. Amendment	63
ARTICLE X	
MISCELLANEOUS	64
Section 10.1. Notices	64
Section 10.2. Waiver; Amendments	65
Section 10.3. Expenses; Indemnification	66
Section 10.4. Successors and Assigns	67
Section 10.5. Governing Law; Jurisdiction; Consent to Service of Process	70
Section 10.6. WAIVER OF JURY TRIAL	70
Section 10.7. Right of Setoff	71
Section 10.8. Counterparts; Integration	71
Section 10.9. Survival	71
Section 10.10. Severability	71 72
Section 10.11. Confidentiality	72
Section 10.11. Confidentiality Section 10.12. Interest Rate Limitation	72
Section 10.13. Waiver of Effect of Corporate Seal	72
Section 10.14. Limitation on Recourse	73
Section 10.15. Amendment and Restatement	73

### **Schedules**

Schedule I Commitments

Schedule II Material Project Documents

Schedule 4.5(b) Borrower Assets Schedule 5.5(b) Insurance

Schedule 5.18 Post-Closing Deliveries
Schedule 6.7(a) Transactions with Affiliates
Schedule 6.8 Restrictive Agreements

### **Annex**

Annex I Applicable Margin

# **Exhibits**

Exhibit A Form of Notice of Conversion/Continuation Exhibit B Form of Assignment and Acceptance

Exhibit C Manta Ray Consent

Exhibit D-1 Form of Texas Deed of Trust Exhibit D-2 Form of Louisiana Mortgage

Exhibit E Form of Note

Exhibit F-1 Form of Sponsor LOC – Enterprise Exhibit F-2 Form of Sponsor LOC — Valero

Exhibit G [Reserved]

Exhibit H Form of Notice to Material Producers

### AMENDED AND RESTATED CREDIT AGREEMENT

THIS AMENDED AND RESTATED CREDIT AGREEMENT is made and entered into as of June 29, 2005, by and among CAMERON HIGHWAY OIL PIPELINE COMPANY, a general partnership formed under the laws of the State of Delaware (the "Borrower"), the several banks and other financial institutions and lenders from time to time party hereto (the "Lenders"), and SUNTRUST BANK, in its capacity as administrative agent for the Lenders (the "Administrative Agent") and in its capacity as collateral agent for the Lenders (the "Collateral Agent"), and amends and restates the Existing Debt Documents.

#### WITNESSETH:

**WHEREAS**, the Borrower has entered into the Existing Debt Documents;

**WHEREAS**, subject to the terms and conditions of this Agreement (as defined below), the Borrower, the Lenders and the other parties hereto have agreed to amend and restate the Existing Debt Documents;

**WHEREAS**, it is the intent of the parties hereto that (x) this Agreement not constitute a novation of the obligations and liabilities existing under the Existing Debt Documents or evidence payment of all or any of such obligations and liabilities, (y) this Agreement amend and restate in its entirety the Existing Debt Documents and (z) from and after the Effective Date (as defined below), the Existing Debt Documents be of no further force or effect except as amended and restated hereby;

**NOW, THEREFORE**, in consideration of the premises and the mutual covenants herein contained, the Borrower, the Lenders, the Administrative Agent and the Collateral Agent agree as follows:

#### ARTICLE I

#### **DEFINITIONS; CONSTRUCTION**

**Section 1.1.** <u>Definitions</u>. In addition to the other terms defined herein, the following terms used herein shall have the meanings herein specified (to be equally applicable to both the singular and plural forms of the terms defined):

"Additional Purchase and Sale Agreement" shall mean any Purchase and Sale Agreement that is not an Effective Date Purchase and Sale Agreement, as such agreement may be amended, modified or supplemented from time to time. For the avoidance of doubt, the Effective Date Purchase and Sale Agreements shall not at any time be considered "Additional Purchase and Sale Agreements."

- "Adjusted LIBO Rate" shall mean, with respect to each Interest Period for a Eurodollar Borrowing, the rate per annum obtained by dividing (i) LIBOR for such Interest Period by (ii) a percentage equal to 1.00 minus the Eurodollar Reserve Percentage.
  - "Administrative Agent" shall have the meaning set forth in the introductory paragraph of this Agreement.
- "Administrative Questionnaire" shall mean, with respect to each Lender, an administrative questionnaire in the form prepared by the Administrative Agent and submitted to the Administrative Agent duly completed by such Lender.
- "Affiliate" shall mean, with respect to a specified Person, another Person that directly, or indirectly through one or more intermediaries, Controls or is Controlled by or is under common Control with the Person specified.
  - "Agreement" shall mean the Existing Debt Documents, as amended and restated by this Amended and Restated Credit Agreement.
- "Applicable Lending Office" shall mean, for each Lender and for each Type of Loan, the "Lending Office" of such Lender (or an Affiliate of such Lender) designated for such Type of Loan in the Administrative Questionnaire submitted by such Lender or such other office of such Lender (or an Affiliate of such Lender) as such Lender may from time to time specify to the Administrative Agent and the Borrower as the office by which its Loans of such Type are to be made and maintained.
- "Applicable Margin" shall mean, as of any date, 0.25% per annum on Base Rate Loans and 1.25% per annum on Eurodollar Loans; <u>provided</u> that upon the Borrower either (a) obtaining a New Rating or (b) not having obtained a New Rating by October 31, 2005, the Applicable Margin shall be based on the grid set forth on <u>Annex I</u>.
- "Approved Fund" shall mean any Person (other than a natural Person) that is (or will be) engaged in making, purchasing, holding or otherwise investing in commercial loans and similar extensions of credit in the ordinary course of its business and that is administered or managed by (i) a Lender, (ii) an Affiliate of a Lender or (iii) an entity or an Affiliate of an entity that administers or manages a Lender.
- "Assignment (CHOPS)" shall mean the assignment and assumption agreement dated as of July 10, 2003 by and between GulfTerra and Manta Ray, as assignors, and the Borrower, as assignee, relating to the assignment to and assumption by the Borrower of certain Project Documents and other assets from such assignors, in each case as such assignment and assumption agreement may be amended, modified or supplemented from time to time to the extent permitted by the terms hereof.
- "Assignment and Acceptance" shall mean an assignment and acceptance entered into by a Lender and an assignee (with the consent of any party whose consent is required by Section 10.4(b)) and accepted by the Administrative Agent, in the form of Exhibit B attached hereto or any other form approved by the Administrative Agent.

"Available Cash Flow" shall mean, for any Fiscal Quarter, EBITDA for such Fiscal Quarter, minus, without duplication, (a) Capital Expenditures made in cash during such Fiscal Quarter to the extent not funded by equity contributions from the Partners, (b) Taxes of the Borrower that were paid in cash during such Fiscal Quarter and (c) Cash Interest Expense paid during such Fiscal Quarter.

"Base Rate" shall mean the higher of (i) the per annum rate which the Administrative Agent publicly announces from time to time to be its prime lending rate, as in effect from time to time, and (ii) the Federal Funds Rate, as in effect from time to time, plus one-half of one percent (0.50%). The Administrative Agent's prime lending rate is a reference rate and does not necessarily represent the lowest or best rate charged to customers. The Administrative Agent may make commercial loans or other loans at rates of interest at, above or below the Administrative Agent's prime lending rate. Each change in the Administrative Agent's prime lending rate shall be effective from and including the entire day such change is publicly announced as being effective.

"BHP" shall mean BHP Billiton Petroleum (Deepwater) Inc., a Delaware corporation.

"BHP Consent" shall mean the Consent and Agreement, dated as of July 10, 2003, by and among BHP, the Borrower and JPMorgan Chase Bank as collateral agent, administrative agent and initial noteholder agent (with the Collateral Agent succeeding JPMorgan Chase Bank as "Collateral Agent" as defined therein), as such agreement may be amended, modified or supplemented from time to time to the extent permitted by the terms hereof.

"BHP Purchase and Sale Agreement" shall mean the Cameron Highway Purchase and Sale Agreement, dated as of June 23, 2003, by and between BHP and the Borrower, as such agreement may be amended, modified or supplemented from time to time to the extent permitted by the terms hereof.

"Borrowing" shall mean a borrowing consisting of Loans of the same Type, made, converted or continued on the same date and in the case of Eurodollar Loans, as to which a single Interest Period is in effect.

"BP" shall mean BP Exploration & Production Inc., a Delaware corporation.

"BP Consent" shall mean the Consent and Agreement, dated as of July 10, 2003, by and among BP, the Borrower, GulfTerra and JPMorgan Chase Bank as collateral agent, administrative agent and initial noteholder agent (with the Collateral Agent succeeding JPMorgan Chase Bank as "Collateral Agent" as defined therein), relating to the BP Purchase and Sale Agreement, as such agreement may be amended, modified or supplemented from time to time to the extent permitted by the terms hereof.

"BP Purchase and Sale Agreement" shall mean the Cameron Highway Purchase and Sale Agreement, dated as of June 23, 2003, by and between BP, GulfTerra and the

Borrower, as such agreement may be amended, modified or supplemented from time to time to the extent permitted by the terms hereof.

"Business Day" shall mean (i) any day other than a Saturday, Sunday or other day on which commercial banks in Atlanta, Georgia and New York, New York are authorized or required by law to close and (ii) if such day relates to a Borrowing of, a payment or prepayment of principal or interest on, a conversion of or into, or an Interest Period for, a Eurodollar Loan or a notice with respect to any of the foregoing, any day on which dealings in Dollars are carried on in the London interbank market.

"Caesar" shall mean Caesar Oil Pipeline Company LLC, a Delaware limited liability company.

"Cameron Highway Oil Pipeline" shall mean an oil pipeline system delivering oil from the Ship Shoal 332 Platform B to multiple delivery points. Such delivery points are located in Texas City, Texas and Port Arthur, Texas on the Effective Date and may include additional delivery points thereafter.

"Cameron Highway System" shall mean the Cameron Highway Oil Pipeline and all equipment and machinery used or useful in connection therewith and owned or leased by the Borrower.

"Capital Expenditures" shall mean, for any Fiscal Quarter, the sum of the aggregate amount of all expenditures of the Borrower (other than expenditures made with the proceeds of casualty insurance or the proceeds of equity contributions from the Partners) for fixed or capital assets made during such Fiscal Quarter that, in accordance with GAAP, would be classified as capital expenditures.

"Capital Lease Obligations" of any Person shall mean the obligations of such Person to pay rent or other amounts under any lease of (or other arrangement conveying the right to use) real or personal property, or a combination thereof, which obligations are required to be classified and accounted for as capital leases on a balance sheet of such Person under GAAP, and the amount of such obligations shall be the capitalized amount thereof determined in accordance with GAAP.

"Capital Stock" shall mean any and all shares, interests, participations or other equivalents (however designated) of capital stock of a corporation, any and all equivalent ownership interests in a Person (other than a corporation) and any and all warrants, rights or options to purchase any of the foregoing.

"Cash Interest Expense" shall mean, with respect to the Borrower for any Fiscal Quarter, (a) (i) the total interest expense of the Borrower for such period and including, in any event, interest capitalized during such Fiscal Quarter and net costs under Interest Rate Agreements for such Fiscal Quarter minus (ii) net gains of the Borrower under Interest Rate Agreements for such Fiscal Quarter, less (b) any non-cash interest for such Fiscal Quarter included in clause (a) above.

"Change in Control" shall mean either of the following: (a) the failure of Enterprise, Valero and Valero LP, collectively, to own, directly or indirectly at least 51% of the legal, beneficial and voting Capital Stock of the Borrower, and (b) at any time, the failure of Enterprise or Valero or a wholly-owned Subsidiary of Enterprise or Valero to be the operator of, and otherwise to operate and maintain, the Cameron Highway System.

"Change in Law" shall mean (i) the adoption of any applicable law, rule or regulation after the Effective Date, (ii) any change in any applicable law, rule or regulation, or any change in the interpretation or application thereof, by any Governmental Authority after the Effective Date, or (iii) compliance by any Lender (or its Applicable Lending Office) with any request, guideline or directive (whether or not having the force of law) of any Governmental Authority made or issued after the Effective Date.

"Code" shall mean the Internal Revenue Code of 1986, as amended and in effect from time to time.

"Collateral" shall mean the property of the Borrower or any other Person that is subject to a Lien created by any Security Document.

"Collateral Agent" shall have the meaning set forth in the introductory paragraph of this Agreement.

"Commitment" shall mean, with respect to each Lender, the obligation of such Lender to make a Loan hereunder on the Effective Date, in a principal amount not exceeding the amount set forth with respect to such Lender on Schedule I. The aggregate principal amount of all Lenders' Commitments is \$415,000,000.

"Connection Agreements" shall mean, collectively, each agreement, other than the Interconnect Agreement, under which the Borrower agrees with one or more counterparties to connect the Cameron Highway System to another pipeline system, refinery, terminal or other third party system, as such agreements may be amended, modified or supplemented from time to time to the extent permitted by the terms hereof.

"Consents" shall mean, collectively, the BP Consent, the UNOCAL Consent, the BHP Consent, and the Manta Ray Consent.

"Control" shall mean the possession, directly or indirectly, of the power to direct or cause the direction of the management or policies of a Person, whether through the ability to exercise voting power, by contract or otherwise. "Controlling" and "Controlled" have meanings correlative thereto.

"Control Account Agreements" shall mean each tri-party agreement by and among the Borrower, the Collateral Agent and a depository bank or securities intermediary at which the Borrower maintains a deposit account or investment account, granting "control" over such deposit accounts and investment accounts to the Collateral Agent in a manner that perfects the Lien of the Collateral Agent under the UCC.

- "Default" shall mean any condition or event that, with the giving of notice or the lapse of time or both, would constitute an Event of Default.
- "Default Interest" shall have the meaning set forth in Section 2.10(b).
- "Distribution Conditions" shall mean the following conditions:
- (a) No more than one Restricted Payment per Fiscal Quarter shall be permitted (and only permitted if all Distribution Conditions shall be satisfied with respect to any such Restricted Payment);
- (b) (i) If such Restricted Payment is to be made during the period when the immediately prior Fiscal Quarter for which financial statements have been or are required to be delivered pursuant to <u>Section 5.1</u> is the Fiscal Quarter ended on September 30, 2005, then the Interest Coverage Ratio for such Fiscal Quarter shall have been at least 2.0:1.0, and (ii) if such Restricted Payment is to be made during any period thereafter, then the Borrower shall be in compliance with <u>Section 5.15</u> with respect to the immediately prior Fiscal Quarter for which financial statements have been or are required to be delivered pursuant to <u>Section 5.1</u>;
- (c) No Restricted Payments, other than as permitted pursuant to <u>Section 6.6(a)</u>, shall be permitted prior to the delivery of the financial statements described in clause (b)(i) of this definition;
- (d) The amount of a given Restricted Payment made pursuant to Section 6.6(b) shall not exceed the amount of Available Cash Flow for the immediately prior Fiscal Quarter for which financial statements have been or are required to be delivered pursuant to Section 5.1 (the "Distribution Measurement Quarter"); provided, that if the Distribution Conditions were not satisfied with respect to one or more Fiscal Quarters occurring prior to the Distribution Measurement Quarter (the "Pre-Distribution Measurement Quarters"), then, if all other Distribution Conditions are satisfied before and after giving effect to the Restricted Payment to be made with respect to the Distribution Measurement Quarter, the amount of the Restricted Payment for the Distribution Measurement Quarter may be increased to include Available Cash Flow for the Pre-Distribution Measurement Quarters to the extent not previously included in a Restricted Payment; and
  - (e) No Default or Event of Default shall have occurred and be continuing.
  - "Dollar(s)" and the sign "\$" shall mean lawful money of the United States of America.

"EBITDA" shall mean, without duplication, as to the Borrower, for a given Fiscal Quarter, the amount equal to net income of the Borrower *less* any non-cash income and gains included in net income, *plus*, to the extent deducted in determining net income, interest expense, depreciation, depletion and impairment, amortization of leasehold and intangibles, other non-cash costs, expenses, charges, write-offs and similar items and taxes; provided that gains or losses on the disposition of assets not in the ordinary course of business shall not be included in EBITDA.

- "Effective Date" shall mean the date on which the conditions precedent set forth in Section 3.1 have been satisfied or waived in accordance with Section 10.2.
- "Effective Date Purchase and Sale Agreements" shall mean, collectively, the BP Purchase and Sale Agreement, the BHP Purchase and Sale Agreement and the UNOCAL Purchase and Sale Agreement.
- "Eligible Assignee" shall mean (i) a Lender; (ii) an Affiliate of a Lender; (iii) an Approved Fund; and (iv) any other Person (other than a natural Person) approved by the Administrative Agent and unless an Event of Default has occurred and is continuing, the Borrower (each such approval not to be unreasonably withheld or delayed).
  - "Enterprise" shall mean Enterprise Products Operating L.P., a Delaware limited partnership.
  - "Enterprise CHOPS" shall mean Cameron Highway Pipeline I, L.P., a Delaware limited partnership.
- "*Environmental Laws*" shall mean all laws, rules, regulations, codes, ordinances, orders, decrees, judgments, injunctions, notices or binding agreements issued, promulgated or entered into by any Governmental Authority, relating in any way to the environment, preservation or reclamation of natural resources, the management, release or threatened release of any Hazardous Material or to health and safety matters.
- "Environmental Liability" shall mean any liability, contingent or otherwise (including any liability for damages, costs of environmental remediation, fines, penalties or indemnities), of the Borrower directly or indirectly resulting from or based upon (a) violation of any Environmental Law, (b) the generation, use, handling, transportation, storage, treatment or disposal of any Hazardous Materials, (c) exposure to any Hazardous Materials, (d) the release or threatened release of any Hazardous Materials into the environment or (e) any contract, agreement or other consensual arrangement pursuant to which liability is assumed or imposed with respect to any of the foregoing.
  - "ERISA" shall mean the Employee Retirement Income Security Act of 1974, as amended from time to time, and any successor statute.
- "ERISA Affiliate" shall mean any trade or business (whether or not incorporated) that, together with the Borrower, is treated as a single employer under Section 414(b) or (c) of the Code or, solely for purposes of Section 302 of ERISA and Section 412 of the Code, is treated as a single employer under Section 414 of the Code.
- "ERISA Reportable Event" shall mean any of the events set forth in Section 4043(b) of ERISA, other than an event for which the 30-day notice period is waived under subsections .13, .14, .16, ..18, .19 or .20 of PBGC Reg. § 2615.

"*Eurodollar*" when used in reference to any Loan or Borrowing, refers to whether such Loan, or the Loans comprising such Borrowing, bears interest at a rate determined by reference to the Adjusted LIBO Rate.

"Eurodollar Reserve Percentage" shall mean the aggregate of the maximum reserve percentages (including, without limitation, any emergency, supplemental, special or other marginal reserves) expressed as a decimal (rounded upwards to the next 1/100th of 1%) in effect on any day to which the Administrative Agent is subject with respect to the Adjusted LIBO Rate pursuant to regulations issued by the Board of Governors of the Federal Reserve System (or any Governmental Authority succeeding to any of its principal functions) with respect to eurocurrency funding (currently referred to as "eurocurrency liabilities" under Regulation D). Eurodollar Loans shall be deemed to constitute eurocurrency funding and to be subject to such reserve requirements without benefit of or credit for proration, exemptions or offsets that may be available from time to time to any Lender under Regulation D. The Eurodollar Reserve Percentage shall be adjusted automatically on and as of the effective date of any change in any reserve percentage.

# "Event of Default" shall have the meaning provided in Article VII.

"Event of Loss" shall mean, with respect to the Project, (a) the loss of all or any material portion of the Project due to destruction or damage beyond repair; (b) the loss of use of all or any material portion of the Project for a period reasonably expected to extend for at least six months for any of the reasons referenced in clause (a); (c) the receipt of insurance proceeds based upon an actual or constructive total loss of the Project; (d) the receipt of insurance proceeds, except for business interruption insurance proceeds, in excess of \$5,000,000; (e) the Taking of all or any portion of the Project such that the then remaining portion cannot practically be utilized for the purposes intended; (f) a Taking or requisition by a Governmental Authority for use of the Project (in its entirety or a material portion as aforesaid) for a stated period which shall exceed, or for an indefinite period which is reasonably expected to exceed, six months; or (g) shutdown of the Project as a result of any Governmental Requirement for a period exceeding six consecutive months.

"Excluded Taxes" shall mean with respect to the Administrative Agent, the Collateral Agent, any Lender or any other recipient of any payment to be made by or on account of any obligation of the Borrower hereunder, (a) income or franchise taxes imposed on (or measured by) its net income by the United States of America, or by the jurisdiction under the laws of which such recipient is organized or in which its principal office is located or, in the case of any Lender, in which its applicable lending office is located, (b) any branch profits taxes imposed by the United States of America or any similar tax imposed by any other jurisdiction in which any Lender is located and (c) in the case of a Foreign Lender, any withholding tax that (i) is imposed on amounts payable to such Foreign Lender at the time such Foreign Lender becomes a party to this Agreement, (ii) is imposed on amounts payable to such Foreign Lender at any time that such Foreign Lender designates a new lending office, other than taxes that have accrued prior to the designation of such lending office that are otherwise not Excluded Taxes, and (iii) is attributable to such Foreign Lender's failure to comply with Section 2.17(e).

"Existing Debt" shall mean the Indebtedness represented by the Existing Debt Documents.

"Existing Debt Documents" shall mean (a) that certain Credit Agreement, dated as of July 10, 2003, among the Borrower, the lenders party thereto, and the administrative agent and the other agents party thereto and (b) that certain Note Purchase Agreement, dated as of July 10, 2003, made by the Borrower in favor of the noteholders party thereto, for the issuance of 7.36% senior secured notes (and such notes issued thereunder), in each case (x) as accompanied by that certain Common Agreement, dated as of July 10, 2003, among the Borrower, such lenders, such noteholders, and the agents party thereto, and (y) as amended, supplemented or otherwise modified prior to the execution and delivery of this Agreement.

"Existing Lenders" shall mean all lenders and other credit providers to the Borrower party to the Existing Debt Documents.

"Federal Funds Rate" shall mean, for any day, the rate per annum (rounded upwards, if necessary, to the next 1/100th of 1%) equal to the weighted average of the rates on overnight Federal funds transactions with member banks of the Federal Reserve System arranged by Federal funds brokers, as published by the Federal Reserve Bank of New York on the next succeeding Business Day or if such rate is not so published for any Business Day, the Federal Funds Rate for such day shall be the average rounded upwards, if necessary, to the next 1/100th of 1% of the quotations for such day on such transactions received by the Administrative Agent from three Federal funds brokers of recognized standing selected by the Administrative Agent.

"Fee Letter" shall mean that certain fee letter, dated as of June 13, 2005, executed by SunTrust Capital Markets, Inc. and SunTrust Bank and accepted by Borrower.

"Financial Officer" shall mean, with respect to any Person, its president, chief financial officer, principal accounting officer, vice president (provided that such vice president has actual knowledge of the ongoing financial affairs of such Person), treasurer or controller.

"Fiscal Quarter" shall mean any fiscal quarter of the Borrower.

"Fiscal Year" shall mean any fiscal year of the Borrower.

"Fitch" shall mean Fitch Ratings Ltd.

"force majeure" shall mean the causes, conditions, events or circumstances which are beyond the reasonable control of the Person claiming force majeure, and could not have been avoided or prevented by such Person's reasonable and diligent foresight, planning and implementation. Such causes, conditions, events and circumstances shall include, without limitation, acts of God, war (declared or undeclared), terrorism, insurrections, hostilities, strikes (other than strikes by the employees of such Person or any of its Affiliates, which strikes shall be deemed not to be a force majeure event), lockouts, riots, floods, fires, storms and conduct which would violate any applicable material criminal laws of a Governmental Authority.

"Foreign Lender" shall mean any Lender that is not a United States person under Section 7701(a)(3) of the Code.

"GAAP" shall mean generally accepted accounting principles in the United States applied on a consistent basis and subject to the terms of Section 1.3.

"GB72 Platform Space Agreement" shall mean the Platform Space Agreement between the Borrower, Enterprise Field Services, LLC and GOM Shelf LLC (and/or other counterparties at the time of execution), as such agreement may be amended, modified or supplemented from time to time to the extent permitted by the terms hereof.

"Governmental Approval" shall mean any authorization, consent, approval, license, lease, ruling, permit, tariff, rate, certification, exemption, filing, variance, claim, order, judgment, decree, publication, notice to, declaration of or with or registration by or with any Governmental Authority.

"Governmental Authority" shall mean the government of the United States of America, any other nation or any political subdivision thereof, whether state or local, and any agency, authority, instrumentality, regulatory body, court, central bank or other entity exercising executive, legislative, judicial, taxing, regulatory or administrative powers or functions of or pertaining to government.

"Governmental Requirement" shall mean any Governmental Approval, law, statute, code, ordinance, order, rule, regulation, judgment, decree, injunction, franchise, certificate, license, or other direction or requirement (including, but not limited to, any of the foregoing which relate to Environmental Laws, energy regulations and occupational, safety and health standards or controls) of any Governmental Authority.

"Guarantee" of or by any Person (the "guarantor") shall mean any obligation, contingent or otherwise, of the guarantor guaranteeing or having the economic effect of guaranteeing any Indebtedness or other obligation of any other Person (the "primary obligor") in any manner, whether directly or indirectly, and including any obligation of the guarantor, direct or indirect, (a) to purchase or pay (or advance or supply funds for the purchase or payment of) such Indebtedness or other obligation or to purchase (or to advance or supply funds for the purchase of) any security for the payment thereof, (b) to purchase or lease property, securities or services for the purpose of assuring the owner of such Indebtedness or other obligation of the payment thereof, (c) to maintain working capital, equity capital or any other financial statement condition or liquidity of the primary obligor so as to enable the primary obligor to pay such Indebtedness or other obligation or (d) as an account party in respect of any letter of credit or letter of guaranty issued to support such Indebtedness or obligation; provided, that the term Guarantee shall not include endorsements for collection or deposit in the ordinary course of business.

"*GulfTerra*" shall mean GulfTerra Energy Partners, L.P., a Delaware limited partnership (n/k/a Enterprise GTM Holdings L.P., a Delaware limited partnership).

"Hazardous Materials" shall mean all explosive or radioactive substances or wastes and all hazardous or toxic substances, wastes or other pollutants, including petroleum or petroleum distillates, asbestos or asbestos containing materials, polychlorinated biphenyls, radon gas, infectious or medical wastes and all other substances or wastes of any nature regulated pursuant to any Environmental Law.

"Hedging Agreement" shall mean any agreement evidencing a Hedging Transaction.

"Hedging Obligations" of any Person shall mean any and all obligations of such Person, whether absolute or contingent and howsoever and whensoever created, arising, evidenced or acquired under (i) any and all Hedging Transactions, (ii) any and all cancellations, buy backs, reversals, terminations or assignments of any Hedging Transactions and (iii) any and all renewals, extensions and modifications of any Hedging Transactions and any and all substitutions for any Hedging Transactions.

"Hedging Transaction" of any Person shall mean any transaction (including an agreement with respect thereto) now existing or hereafter entered into by such Person that is a rate swap, basis swap, forward rate transaction, commodity swap, interest rate option, foreign exchange transaction, cap transaction, floor transaction, collateral transaction, forward transaction, currency swap transaction, cross-currency rate swap transaction, currency option or any other similar transaction (including any option with respect to any of these transactions) or any combination thereof, whether linked to one or more interest rates, foreign currencies, commodity prices, equity prices or other financial measures.

"Indebtedness" of any Person shall mean, without duplication:

- (a) all obligations of such Person for borrowed money or with respect to deposits or advances of any kind;
- (b) all obligations of such Person evidenced by bonds, debentures, notes or other similar instruments;
- (c) all obligations of such Person upon which interest charges are customarily paid;
- (d) all obligations of such Person under conditional sale or other title retention agreements relating to property acquired by such Person;
- (e) all obligations of such Person in respect of the deferred purchase price of property or services (excluding current accounts payable incurred in the ordinary course of business and for borrowed money);
- (f) all Indebtedness of others secured by (or for which the holder of such Indebtedness has an existing right, contingent or otherwise, to be secured by) any Lien on property owned or acquired by such Person, whether or not the Indebtedness secured thereby has been assumed;

- (g) all Guarantees by such Person of Indebtedness of others;
- (h) all Capital Lease Obligations in respect of which such Person is liable, contingently or otherwise, as obligor, guarantor or otherwise, or in respect of which obligations such Person otherwise assures a creditor against loss;
  - (i) all obligations, contingent or otherwise, of such Person as an account party in respect of letters of credit and letters of guaranty; and
  - (j) all obligations, contingent or otherwise, of such Person in respect of bankers' acceptances.

The Indebtedness of any Person shall include the Indebtedness of any other entity (including any partnership in which such Person is a general partner) to the extent such Person is liable therefor as a result of such Person's ownership interest in or other relationship with such entity, except to the extent the terms of such Indebtedness provide that such Person is not liable therefor.

"Indemnified Taxes" shall mean Taxes other than Excluded Taxes.

"Interconnect Agreement" shall mean the Offshore Facilities Interconnection, Construction and Operating Agreement, effective as of June 23, 2003 among Caesar, Manta Ray, the Borrower and GulfTerra, as such agreement may be amended, modified or supplemented from time to time to the extent permitted by the terms hereof.

"Interest Coverage Ratio" shall mean, for any Fiscal Quarter, the ratio of (a) EBITDA for such Fiscal Quarter to (b) Cash Interest Expense for such Fiscal Quarter.

"Interest Period" shall mean with respect to any Eurodollar Borrowing, a period of one, two or three months; provided, that:

- (i) the initial Interest Period for such Borrowing shall commence on the date of such Borrowing (including the date of any conversion from a Borrowing of another Type), and each Interest Period occurring thereafter in respect of such Borrowing shall commence on the day on which the next preceding Interest Period expires;
- (ii) if any Interest Period would otherwise end on a day other than a Business Day, such Interest Period shall be extended to the next succeeding Business Day, unless such Business Day falls in another calendar month, in which case such Interest Period would end on the next preceding Business Day;
- (iii) any Interest Period which begins on the last Business Day of a calendar month or on a day for which there is no numerically corresponding day in the calendar month at the end of such Interest Period shall end on the last Business Day of such calendar month; and
  - (iv) no Interest Period may extend beyond the Maturity Date.

"Interest Rate Agreements" shall mean any interest rate swap, cap, collar or similar agreements between the Borrower and a financial institution for the transfer or mitigation of interest rate risks related to the Project.

"Lenders" shall have the meaning assigned to such term in the opening paragraph of this Agreement.

"LIBOR" shall mean, for any applicable Interest Period with respect to any Eurodollar Loan, the British Bankers' Association Interest Settlement Rate per annum for deposits in Dollars for a period equal to such Interest Period appearing on the display designated as Page 3750 on the Dow Jones Markets Service (or such other page on that service or such other service designated by the British Bankers' Association for the display of such Association's Interest Settlement Rates for Dollar deposits) as of 11:00 a.m. (London, England time) on the day that is two Business Days prior to the first day of the Interest Period or if such Page 3750 is unavailable for any reason at such time, the rate which appears on the Reuters Screen ISDA Page as of such date and such time; provided, that if the Administrative Agent determines that the relevant foregoing sources are unavailable for the relevant Interest Period, LIBOR shall mean the rate of interest determined by the Administrative Agent to be the average (rounded upward, if necessary, to the nearest 1/100th of 1%) of the rates per annum at which deposits in Dollars are offered to the Administrative Agent two (2) Business Days preceding the first day of such Interest Period by leading banks in the London interbank market as of 10:00 a.m. (New York time) for delivery on the first day of such Interest Period, for the number of days comprised therein and in an amount comparable to the amount of the Eurodollar Loan of the Administrative Agent.

"*License Agreements*" shall mean, collectively, each agreement under which one or more counterparties agrees to grant a license or right of way to the Borrower in connection with a Connection Agreement, as such agreements may be amended, modified or supplemented from time to time to the extent permitted by the terms hereof.

"Lien" shall mean any interest in any asset or property securing an obligation owed to, or securing a claim by, a Person other than the owner of the asset or property, whether such interest is based on common law, statute or contract, and whether such obligation or claim is fixed or contingent, and including, but not limited to the lien or security interest arising from any mortgage, deed of trust, security agreement, pledge, hypothecation, conditional sale or trust receipt or a lease, consignment or bailment for security purposes. The term "Lien" shall include reservations, exceptions, encroachments, easements, encumbrances, rights of way, covenants, conditions, restrictions, leases and other title exceptions affecting assets or property. For purposes of this Agreement, the Borrower shall be deemed to be the owner of any asset or property which it has acquired or holds subject to a conditional sale agreement, financing lease or other arrangement pursuant to which title to such asset or property has been retained by or vested in some other Person for security purposes.

"Loan" shall have the meaning set forth in Section 2.2.

"Loan Documents" shall mean, collectively, this Agreement, the Notes (if any), the Security Documents, all Notices of Conversion/Continuation and any and all other

instruments, agreements, documents and writings executed in connection with any of the foregoing.

- "Manta Ray" shall mean Manta Ray Gathering Company, L.L.C., a Delaware limited liability company.
- "Manta Ray Consent" shall mean a Consent and Agreement substantially in the form of Exhibit C.
- "Material Adverse Effect" shall mean any event, occurrence or condition which has, or which reasonably could be expected to have, a material and adverse effect on (a) the Project or on the business, assets, operations, or financial condition of the Borrower, (b) the Borrower's ability to perform any of its obligations under the Material Documents, (c) the validity or enforceability of any material term of any Material Document or (d) the validity or priority of, or the rights of or benefits available to the Administrative Agent, the Collateral Agent or the Lenders under, this Agreement or any of the other Loan Documents.
  - "Material Documents" shall mean each Loan Document and each Material Project Document.
- "Material Indebtedness" shall mean Indebtedness (other than under the Loan Documents), or obligations in respect of one or more Hedging Transactions, of the Borrower in an aggregate principal amount exceeding \$5,000,000. For purposes of determining Material Indebtedness, the "principal amount" of the obligations of the Borrower in respect of any Hedging Transaction at any time shall be the Net Mark-to-Market Exposure of such Hedging Transaction.
- "*Material Permits*" shall mean all Governmental Approvals necessary as of the date of determination for, and material to, the construction, development, use, operation, ownership, or maintenance of the Cameron Highway System.
- "*Material Producer*" shall mean, BP, BHP, and UNOCAL (or any permitted assignees thereof under the applicable Effective Date Purchase and Sale Agreement and the applicable Consent).
- "*Material Project Documents*" shall mean the documents listed on <u>Schedule II</u>, the GB72 Platform Space Agreement, the Interconnect Agreement, the Connection Agreements and the License Agreements (in each case as such contracts may be amended, modified, supplemented or replaced from time to time).
- "Maturity Date" shall mean the earlier of (i) June 28, 2006 or (ii) the date on which the principal amount of all outstanding Loans have been declared or automatically have become due and payable (whether by acceleration or otherwise).
  - "Moody's" shall mean Moody's Investors Service, Inc.

"Mortgages" shall mean each (a) Amended and Restated Deed of Trust, Assignment, Security Agreement and Financing Statement (Texas) substantially in the form of <a href="Exhibit D-1">Exhibit D-1</a> or in form and substance reasonably satisfactory to the Collateral Agent and (b) Amended and Restated Mortgage, Assignment, Security Agreement and Financing Statement (Louisiana) substantially in the form of <a href="Exhibit D-2">Exhibit D-2</a> or in form and substance reasonably satisfactory to the Collateral Agent, in each case as such agreements may be amended, modified or supplemented from time to time to the extent permitted by the terms hereof.

"Multiemployer Plan" shall mean a multiemployer plan as defined in Section 3(37) or Section 4001(a)(3) of ERISA.

"Net Cash Proceeds" shall mean, with respect to any event, (a) the cash proceeds actually received in respect of such event including any cash received in respect of any non-cash proceeds, but only as and when received, net of (b) the sum of (i) all reasonable fees and out-of-pocket expenses paid or payable by the Borrower to third parties (other than to Affiliates) in connection with such event and (ii) the amount of all taxes paid (or reasonably estimated to be payable) by the Borrower.

"Net Mark-to-Market Exposure" of any Person shall mean, as of any date of determination with respect to any Hedging Obligation, the excess (if any) of all unrealized losses over all unrealized profits of such Person arising from such Hedging Obligation. "Unrealized losses" shall mean the fair market value of the cost to such Person of replacing the Hedging Transaction giving rise to such Hedging Obligation as of the date of determination (assuming the Hedging Transaction were to be terminated as of that date), and "unrealized profits" means the fair market value of the gain to such Person of replacing such Hedging Transaction as of the date of determination (assuming such Hedging Transaction were to be terminated as of that date).

"New Rating" shall mean a new rating for proposed permanent senior secured debt that is to be issued in whole or in part for the purpose of refinancing the Loans.

"*Note*" shall mean a promissory note of the Borrower payable to the order of a requesting Lender in the principal amount of such Lender's Commitment, in substantially the form of <u>Exhibit E</u>.

"*Notice of Conversion/Continuation*" shall mean the notice given by the Borrower to the Administrative Agent in respect of the conversion or continuation of an outstanding Borrowing as provided in <u>Section 2.4(b)</u>.

"Obligations" shall mean all amounts owing by the Borrower to the Administrative Agent, the Collateral Agent or any Lender pursuant to or in connection with this Agreement or any other Loan Document, including without limitation, all principal, interest (including any interest accruing after the filing of any petition in bankruptcy or the commencement of any insolvency, reorganization or like proceeding relating to the Borrower, whether or not a claim for post-filing or post-petition interest is allowed in such proceeding), all reimbursement obligations, fees, expenses, indemnification and reimbursement payments, costs and expenses (including all reasonable fees, charges and disbursements of one law firm and one additional Louisiana local counsel, for the Administrative Agent, the Collateral Agent and their

respective Affiliates (and, if retention is deemed appropriate by the Administrative Agent and consented to by the Borrower (such consent not to be unreasonably withheld or delayed), additional local counsel to the Administrative Agent, the Collateral Agent and their respective Affiliates) and, only in the case of enforcement, any Lender, incurred pursuant to this Agreement or any other Loan Document), whether direct or indirect, absolute or contingent, liquidated or unliquidated, now existing or hereafter arising hereunder or thereunder, and all Hedging Obligations owed to the Administrative Agent, the Collateral Agent any Lender or any of their Affiliates incurred in order to limit interest rate or fee fluctuation with respect to the Loans, and all obligations and liabilities incurred in connection with collecting and enforcing the foregoing, together with all renewals, extensions, modifications or refinancings thereof.

"Operation and Management Agreement" shall mean the Operation and Management Agreement, dated as of June 30, 2003, by and between the Borrower and Manta Ray, as such agreement may be amended, modified or supplemented from time to time to the extent permitted by the terms hereof.

"Other Taxes" shall mean any and all present or future stamp or documentary taxes or any other excise or property taxes, charges or similar levies arising from any payment made hereunder or from the execution, delivery or enforcement of, or otherwise with respect to, this Agreement or any other Loan Document.

"Participant" shall have the meaning set forth in Section 10.4(d).

"Participation Agreement" shall mean the Participation Agreement and Assignment, dated as of July 10, 2003, by and between Valero, GulfTerra, Enterprise CHOPS and Manta Ray, as such agreement may be amended, modified or supplemented from time to time to the extent permitted by the terms hereof.

"Partner Distribution" shall mean up to a \$125,000,000 dividend made by the Borrower to the Partners on the Effective Date with proceeds of the Loan.

"Partners" shall mean each of Enterprise CHOPS, Valero CHOPS I and Valero CHOPS II.

"*Partnership Agreement*" shall mean the Amended and Restated Partnership Agreement of the Borrower, dated as of July 10, 2003, by and between the Partners, as such agreement may be amended, modified or supplemented from time to time to the extent permitted by the terms hereof.

"*Payment Office*" shall mean the office of the Administrative Agent located at 303 Peachtree Street, N.E., Atlanta, Georgia 30308, or such other location as to which the Administrative Agent shall have given written notice to the Borrower and the other Lenders.

"PBGC" shall mean the Pension Benefit Guaranty Corporation referred to and defined in ERISA, and any successor entity performing similar functions.

"Permitted Encumbrances" shall mean:

- (a) Liens imposed by law for Taxes that are not yet due or which are being contested in good faith by appropriate proceedings; *provided* that adequate reserves with respect thereto are maintained on the books of the Borrower, in conformity with GAAP;
- (b) carriers', warehousemen's, mechanics', materialmen's, repairmen's and other like Liens, imposed by law, arising in the ordinary course of business and securing obligations that are not overdue by more than 30 days or which are being contested in good faith by appropriate proceedings; *provided* that adequate reserves with respect thereto are maintained on the books of the Borrower, in conformity with GAAP;
- (c) Liens incurred (other than any inchoate Lien imposed by ERISA) or pledges or deposits made in the ordinary course of business in connection with workers' compensation, unemployment insurance and other social security legislation, or to secure the performance of tenders, statutory obligations, surety and appeal bonds, bids, leases, construction, operating and maintenance agreements, government contracts, performance and return-of-money bonds and other similar obligations (exclusive of obligations for the payment of borrowed money), and deposits securing liability to insurance carriers under insurance or self-insurance arrangements, in each case having ordinary and customary terms;
- (d) easements, rights-of-way, restrictions, servitudes, permits, reservations, encroachments, exceptions, conditions, covenants and other similar charges or encumbrances on real property imposed by law or arising in the ordinary course of business that do not secure any monetary obligations and do not materially detract from the value of the affected property or interfere with the ordinary conduct of business of the Borrower;
  - (e) inchoate Liens arising under ERISA;
- (f) any obligations or duties affecting any of the assets of the Borrower to any municipality or public authority with respect to any franchise, grant, license or permit which do not materially impair the use of such asset for the purposes for which it is held;
- (g) defects, irregularities and deficiencies in title of any rights-of-way or other property of the Borrower which in the aggregate do not materially and adversely affect the business of the Borrower, and defects, irregularities and deficiencies in title to any property of the Borrower which defects, irregularities or deficiencies have been cured by possession under applicable statutes of limitation;
- (h) security interests arising by operation of law solely under Article 2 of the Uniform Commercial Code to the extent and so long as the "debtor" with respect to such security interests does not have or does not lawfully obtain possession of the goods subject thereto; and

(i) any right of set off arising under common law or by statute;

provided that the term "Permitted Encumbrances" shall not include any Lien securing Indebtedness.

### "Permitted Investments" shall mean:

- (a) marketable direct obligations issued or unconditionally guaranteed by the United States of America (or by any agency thereof to the extent such obligations are backed by the full faith and credit of the United States of America), in each case maturing within one year from the date of acquisition thereof;
- (b) marketable direct obligations issued by any state or public instrumentality thereof maturing within one year from the date of acquisition thereof, and at the time of acquisition, having the highest rating obtainable from either S&P or Moody's;
- (c) investments in commercial paper maturing no more than one year from the date of creation thereof and, at the time of acquisition, having the highest credit rating obtainable from S&P or from Moody's;
- (d) investments in certificates of deposit or banker's acceptances maturing within one year from the date of acquisition thereof issued by (i) any Lender, (ii) any commercial bank organized under the laws of the United States of America or any state thereof or the District of Columbia having combined capital and surplus and undivided profits of not less than \$500,000,000 or (iii) any bank which has a short-term commercial paper rating meeting the requirements of clause (c) above (any such Lender or bank, a "Qualifying Lender");
- (e) investments in eurodollar time deposits having a maturity of less than one year purchased directly from any Lender (whether such deposit is with such Lender or any other Lender) or issued by any Qualifying Lender; and
- (f) fully collateralized repurchase agreements and reverse repurchase agreements with a term of not more than 14 days with any Qualifying Lender relating to securities described in clause (a) above.

"*Person*" shall mean any natural person, corporation, limited liability company, trust, joint venture, association, company, partnership, Governmental Authority or other entity.

"*Plan*" shall mean any employee pension benefit plan (other than a Multiemployer Plan) subject to the provisions of Title IV of ERISA or Section 412 of the Code or Section 302 of ERISA, and in respect of which the Borrower or any ERISA Affiliate is (or, if such plan were terminated, would under Section 4069 of ERISA be deemed to be) an "employer" as defined in Section 3(5) of ERISA.

- "Pledge Agreements" shall mean each Pledge Agreement, dated as of the date hereof, executed by a Partner in favor of the Collateral Agent for the benefit of the Lenders.
- "*Pro Rata Share*" shall mean with respect to any Lender at any time, a percentage, the numerator of which shall be such Lender's Loan and the denominator of which shall be the sum of all Loans.
  - "Project" shall mean the construction, installation and operation of the Cameron Highway System.
- "*Project Documents*" shall mean all of the contracts relating to ownership, operation and maintenance of the Cameron Highway System (as such contracts may be amended, modified, supplemented or replaced from time to time) entered into before, on or after the Effective Date.
  - "Projections" shall mean the financial projections provided by the Borrower to the Administrative Agent on or about June 22, 2005.
- "Purchase and Sale Agreements" shall mean, collectively, the BP Purchase and Sale Agreement, the BHP Purchase and Sale Agreement, and the UNOCAL Purchase and Sale Agreement, and each other agreement entered into by the Borrower or its Affiliates relating to the purchase of oil from and/or the sale of oil to the Cameron Highway Oil Pipeline as such agreements may be amended, modified or supplemented from time to time to the extent permitted by the terms hereof.
  - "Real Estate" shall mean all real property (a) owned or leased by the Borrower, and (b) if related to the Project, owned or leased by Manta Ray.
- "*Regulation D*" shall mean Regulation D of the Board of Governors of the Federal Reserve System, as the same may be in effect from time to time, and any successor regulations then in effect.
- "Related Parties" shall mean, with respect to any specified Person, such Person's Affiliates and the respective directors, officers, employees, agents and advisors of such Person and such Person's Affiliates.
- "*Release*" shall mean any release, spill, emission, leaking, dumping, injection, pouring, deposit, disposal, discharge, dispersal, leaching or migration into the environment (including ambient air, surface water, groundwater, land surface or subsurface strata) or within any building, structure, facility or fixture.
- "Required Lenders" shall mean, at any time, Lenders holding more than 50% of the aggregate outstanding Loans at such time; <u>provided</u> that if there are two Lenders at such time, and such Lenders each hold greater than 33% of the aggregate outstanding Loans at such time, then "Required Lenders" shall mean Lenders holding 100% of the aggregate outstanding Loans at such time.

- "Responsible Officer" shall mean any of the president, the chief executive officer, the chief operating officer, the chief financial officer, the treasurer or a vice president of the Borrower or such other representative of the Borrower as may be designated in writing by any one of the foregoing with the consent of the Administrative Agent.
- "Restricted Payment" shall mean any dividend or other distribution (whether in cash, securities or other property) with respect to any Capital Stock of the Borrower, or any payment (whether in cash, securities or other property), including any sinking fund or similar deposit, in connection with the purchase, redemption, retirement, acquisition, cancellation or termination of any Capital Stock of the Borrower or any option, warrant or other right to acquire any such Capital Stock of the Borrower and any payment with respect to Indebtedness owing to either Sponsor or an Affiliate of either Sponsor.
  - "S&P" shall mean Standard & Poor's, a Division of the McGraw-Hill Companies.
- "Secured Affiliate" shall mean any Affiliate of any Lender that has entered into an Interest Rate Agreement with the Borrower with the obligations of the Borrower thereunder being secured by one or more Security Documents.
- "Security Agreement" shall mean that certain Security Agreement, dated as of the date hereof, executed by the Borrower in favor of the Collateral Agent for the benefit of the Lenders.
- "Security Documents" shall mean, collectively, the Security Agreement, the Pledge Agreements, the Mortgages, the Control Account Agreements, the Consents, and all other instruments and agreements now or hereafter securing the whole or any part of the Obligations or any Guarantee thereof, all UCC financing statements, fixture filings, stock powers, and all other documents, instruments, agreements and certificates executed and delivered by the Borrower or any Affiliate thereof to the Administrative Agent, the Collateral Agent or the Lenders in connection with the foregoing.
  - "Ship Shoal 332 Platform B" shall mean a pipeline junction platform owned by the Borrower located on the outer continental shelf at Ship Shoal 332.
- "Sponsor LOCs" shall mean Letters of Credit in substantially the form of Exhibit F-1 and Exhibit F-2, as applicable, in the amount of \$14,000,000 each.
  - "Sponsors" shall mean each of Enterprise and Valero.
- "*Taking*" shall mean the condemnation, confiscation or seizure of title by any Governmental Authority (or otherwise by right of eminent domain or conveyance in lieu thereof) of all or any portion of the Project.
- "*Taxes*" shall mean any and all present or future taxes, levies, imposts, duties, deductions, charges or withholdings imposed by any Governmental Authority.

- "*Transaction Documents*" shall mean this Agreement, the Project Documents, the Loan Documents and all other documents, agreements, instruments and certificates in connection with the foregoing.
  - "Transaction Party" shall mean a Person that is a party to a Transaction Document.
- "*Transactions*" shall mean the execution, delivery and performance by the Borrower of the Transaction Documents, the making of Loans, and the use of the proceeds thereof.
- "*Type*", when used in reference to a Loan or Borrowing, refers to whether the rate of interest on such Loan, or on the Loans comprising such Borrowing, is determined by reference to the Adjusted LIBO Rate or the Base Rate.
- "*UCC*" shall mean the Uniform Commercial Code as from time to time in effect in the State of New York; *provided, however*, that, in the event that, by reason of mandatory provisions of law, any of the attachment, perfection or priority of the Collateral Agent's and the Lenders' security interest in any Collateral is governed by the Uniform Commercial Code as in effect in a jurisdiction other than the State of New York, the term "*UCC*" shall mean the Uniform Commercial Code as in effect in such other jurisdiction for purposes of the provisions hereof relating to such attachment, perfection or priority and for purposes of definitions related to such provisions.
  - "UNOCAL" shall mean Union Oil Company of California, a California corporation.
- "UNOCAL Consent" shall mean the Consent and Agreement, dated as of July 10, 2003, by and among UNOCAL, the Borrower, and JPMorgan Chase Bank as collateral agent, administrative agent and initial noteholder agent (with the Collateral Agent succeeding JPMorgan Chase Bank as "Collateral Agent" as defined therein), as such agreement may be amended, modified or supplemented from time to time to the extent permitted by the terms hereof.
- "UNOCAL Purchase and Sale Agreement" shall mean the Cameron Highway Purchase and Sale Agreement, dated as of June 23, 2003, by and between UNOCAL and the Borrower, as such agreement may be amended, modified or supplemented from time to time to the extent permitted by the terms hereof.
  - "Valero" shall mean Valero Energy Corporation, a Delaware corporation.
  - "Valero CHOPS" shall mean, collectively, Valero CHOPS I and Valero CHOPS II.
- "Valero CHOPS I" shall mean Valero CHOPS I, L.P., a Delaware limited partnership or, after Valero Energy Corporation transfers the direct partnership interest in the Borrower held by Valero CHOPS I to Valero LP or a Subsidiary thereof, pursuant to a

Valero/Valero LP Transfer (as defined in the Partnership Agreement), "Valero CHOPS I" shall mean Valero LP or such Subsidiary thereof.

"Valero CHOPS II" shall mean Valero CHOPS II, L.P., a Delaware limited partnership or, after Valero Energy Corporation transfers the direct partnership interest in the Borrower held by Valero CHOPS II to Valero LP or a Subsidiary thereof, pursuant to a Valero/Valero LP Transfer (as defined in the Partnership Agreement), "Valero CHOPS II" shall mean Valero LP or such Subsidiary thereof.

"Valero LP" shall mean Valero L.P., a Delaware limited partnership.

"Welfare Plan" shall mean any employee welfare benefit plan, as defined in section 3(1) of ERISA, including, without limitation, any such plan maintained to provide benefits to former employees of such entities, that may not be terminated by such entities in their sole discretion at any time without any material liability.

"Withdrawal Liability" shall mean liability to a Multiemployer Plan as a result of a complete or partial withdrawal from such Multiemployer Plan, as such terms are defined in Part I of Subtitle E of Title IV of ERISA.

**Section 1.2.** <u>Classifications of Loans and Borrowings</u>. For purposes of this Agreement, Loans may be classified and referred to by Type (*e.g.* a "Eurodollar Loan" or "Base Rate Loan"). Borrowings also may be classified and referred to by Type (*e.g.* "Eurodollar Borrowing").

**Section 1.3.** Accounting Terms and Determination. Unless otherwise defined or specified herein, all accounting terms used herein shall be interpreted, all accounting determinations hereunder shall be made, and all financial statements required to be delivered hereunder shall be prepared, in accordance with GAAP as in effect from time to time, applied on a basis consistent with the most recent audited financial statements of the Borrower delivered pursuant to Section 5.1(a); provided, that if the Borrower notifies the Administrative Agent that the Borrower wishes to amend any covenant in Section 5.15 or Section 6.6 to eliminate the effect of any change in GAAP on the operation of such covenant (or if the Administrative Agent notifies the Borrower that the Required Lenders wish to amend Section 5.15 or Section 6.6 for such purpose), then the Borrower's compliance with such covenant shall be determined on the basis of GAAP in effect immediately before the relevant change in GAAP became effective, until either such notice is withdrawn or such covenant is amended in a manner satisfactory to the Borrower and the Required Lenders.

**Section 1.4.** Terms Generally. The definitions of terms herein shall apply equally to the singular and plural forms of the terms defined. Whenever the context may require, any pronoun shall include the corresponding masculine, feminine and neuter forms. The words "include," "includes," and "including" shall be deemed to be followed by the phrase "without limitation". The word "will" shall be construed to have the same meaning and effect as the word "shall". In the computation of periods of time from a specified date to a later specified date, the word "from" means "from and including" and the word "to" means "to but excluding". Unless the context requires otherwise (i) any definition of or reference to any agreement, instrument or

other document herein shall be construed as referring to such agreement, instrument or other document as it was originally executed or as it may from time to time be amended, restated, supplemented or otherwise modified (subject to any restrictions on such amendments, supplements or modifications set forth herein), (ii) any reference herein to any Person shall be construed to include such Person's successors and permitted assigns, (iii) the words "hereof", "herein" and "hereunder" and words of similar import shall be construed to refer to this Agreement as a whole and not to any particular provision hereof, (iv) all references to Articles, Sections, Exhibits, Annexes and Schedules shall be construed to refer to Articles, Sections, Exhibits, Annexes and Schedules to this Agreement and (v) all references to a specific time shall be construed to refer to the time in the city and state of the Administrative Agent's principal office, unless otherwise indicated.

# **ARTICLE II**

### **AMOUNT AND TERMS OF THE COMMITMENTS**

**Section 2.1.** <u>General Description of Facilities</u>. Subject to and upon the terms and conditions herein set forth, each Lender severally agrees to make a Loan to the Borrower in a principal amount not exceeding such Lender's Commitment on the Effective Date.

Section 2.2. <u>Commitments</u>. Subject to the terms and conditions set forth herein, each Lender severally agrees to make a single loan (each, a "*Loan*") to the Borrower on the Effective Date in a principal amount not to exceed the Commitment of such Lender; <u>provided</u>, that if for any reason the full amount of such Lender's Commitment is not fully drawn on the Effective Date, the undrawn portion thereof shall automatically be cancelled. The Loans may be, from time to time, Base Rate Loans or Eurodollar Loans or a combination thereof. The execution and delivery of this Agreement by the Borrower and the satisfaction of all conditions precedent pursuant to <u>Article III</u> shall be deemed to constitute the Borrower's request to borrow the Loans on the Effective Date.

### Section 2.3. Funding of Borrowings.

- (a) Each Lender will make available each Loan to be made by it hereunder on the proposed date thereof by wire transfer in immediately available funds by 11:00 a.m. (New York time) to the Administrative Agent at the Payment Office. The Administrative Agent will make such Loans available to the Borrower by promptly crediting the amounts that it receives, in like funds by the close of business on such proposed date, to an account maintained by the Borrower with the Administrative Agent or at the Borrower's option, by effecting a wire transfer of such amounts to an account designated by the Borrower to the Administrative Agent.
- (b) Unless the Administrative Agent shall have been notified by any Lender prior to 5:00 p.m. (New York time) one (1) Business Day prior to the date of a Borrowing in which such Lender is to participate that such Lender will not make available to the Administrative Agent such Lender's share of such Borrowing, the Administrative Agent may assume that such Lender has made such amount available to the Administrative Agent on such date, and the Administrative Agent, in reliance on such assumption, may make available to the Borrower on such date a corresponding amount. If such corresponding amount is not in fact made available to

the Administrative Agent by such Lender on the date of such Borrowing, the Administrative Agent shall be entitled to recover such corresponding amount on demand from such Lender together with interest at the Federal Funds Rate until the second Business Day after such demand and thereafter at the Base Rate. If such Lender does not pay such corresponding amount forthwith upon the Administrative Agent's demand therefor, the Administrative Agent shall promptly notify the Borrower, and the Borrower shall immediately pay such corresponding amount to the Administrative Agent together with interest at the rate specified for such Borrowing. Nothing in this subsection shall be deemed to relieve any Lender from its obligation to fund its Pro Rata Share of any Borrowing hereunder or to prejudice any rights which the Borrower may have against any Lender as a result of any default by such Lender hereunder.

(c) No Lender shall be responsible for any default by any other Lender in its obligations hereunder, and each Lender shall be obligated to make its Loans provided to be made by it hereunder, regardless of the failure of any other Lender to make its Loans hereunder.

# Section 2.4. Interest Elections.

- (a) After the making of the initial Loan, the Borrower may elect to convert such Borrowing into a different Type or to continue such Borrowing, and in the case of a Eurodollar Borrowing, may elect Interest Periods therefor, all as provided in this Section. The Borrower may elect different options with respect to different portions of the affected Borrowing, in which case each such portion shall be allocated ratably among the Lenders holding Loans comprising such Borrowing, and the Loans comprising each such portion shall be considered a separate Borrowing.
- (b) To make an election pursuant to this Section, the Borrower shall give the Administrative Agent prior written notice (or telephonic notice promptly confirmed in writing) substantially in the form of Exhibit A (a "Notice of Conversion/Continuation") of each Borrowing that is to be converted or continued, as the case may be, (x) prior to 11:00 a.m. (New York time) on the requested date of a conversion into a Base Rate Borrowing and (y) prior to 11:00 a.m. (New York time) three (3) Business Days prior to a continuation of or conversion into a Eurodollar Borrowing. Each such Notice of Conversion/Continuation shall be irrevocable and shall specify (i) the Borrowing to which such Notice of Conversion/Continuation applies and if different options are being elected with respect to different portions thereof, the portions thereof that are to be allocated to each resulting Borrowing (in which case the information to be specified pursuant to clauses (iii) and (iv) shall be specified for each resulting Borrowing); (ii) the effective date of the election made pursuant to such Notice of Conversion/Continuation, which shall be a Business Day, (iii) whether the resulting Borrowing is to be a Base Rate Borrowing or a Eurodollar Borrowing; and (iv) if the resulting Borrowing is to be a Eurodollar Borrowing, the Interest Period applicable thereto after giving effect to such election, which shall be a period contemplated by the definition of "Interest Period". If any such Notice of Conversion/Continuation requests a Eurodollar Borrowing but does not specify an Interest Period, the Borrower shall be deemed to have selected an Interest Period of one month.
- (c) If, on the expiration of any Interest Period in respect of any Eurodollar Borrowing, the Borrower shall have failed to deliver a Notice of Conversion/Continuation, then,

unless such Borrowing is repaid as provided herein, the Borrower shall be deemed to have elected to convert such Borrowing to a Base Rate Borrowing. No Borrowing may be converted into, or continued as, a Eurodollar Borrowing if a Default or an Event of Default exists, unless the Administrative Agent and each of the Lenders shall have otherwise consented in writing. No conversion of any Eurodollar Loans shall be permitted except on the last day of the Interest Period in respect thereof.

- (d) Upon receipt of any Notice of Conversion/Continuation, the Administrative Agent shall promptly notify each Lender of the details thereof and of such Lender's portion of each resulting Borrowing.
- (e) The aggregate principal amount of each Eurodollar Borrowing shall be not less than \$5,000,000 or a larger multiple of \$1,000,000, and the aggregate principal amount of each Base Rate Borrowing shall not be less than \$1,000,000 or a larger multiple of \$500,000. At no time shall the total number of Eurodollar Borrowings outstanding at any time exceed five.
- **Section 2.5.** Optional Reduction and Termination of Commitments. The Commitments shall terminate on the Effective Date upon the making of the Loans pursuant to Section 2.2.
- **Section 2.6.** Repayment of Loans. The Borrower unconditionally promises to pay to the Administrative Agent for the account of each Lender the then unpaid principal amount of the Loan of such Lender payable on such date(s) and in such amounts as may be required from time to time pursuant to this Agreement; provided that, to the extent not previously paid, the aggregate unpaid principal balance of the Loans shall be due and payable on the Maturity Date.
- Section 2.7. Evidence of Indebtedness. (a) Each Lender shall maintain in accordance with its usual practice appropriate records evidencing the Indebtedness of the Borrower to such Lender resulting from each Loan made by such Lender from time to time, including the amounts of principal and interest payable thereon and paid to such Lender from time to time under this Agreement. The Administrative Agent shall maintain appropriate records in which shall be recorded (i) the Commitment of each Lender, (ii) the amount of each Loan made hereunder by each Lender, the Type thereof and the Interest Period applicable thereto, (iii) the date of each continuation thereof pursuant to Section 2.4, (iv) the date of each conversion of all or a portion thereof to another Type pursuant to Section 2.4, (v) the date and amount of any principal or interest due and payable or to become due and payable from the Borrower to each Lender hereunder in respect of such Loans and (vi) both the date and amount of any sum received by the Administrative Agent hereunder from the Borrower in respect of the Loans and each Lender's Pro Rata Share thereof. The entries made in such records shall be *prima facie* evidence of the existence and amounts of the obligations of the Borrower therein recorded; provided, that the failure or delay of any Lender or the Administrative Agent in maintaining or making entries into any such record or any error therein shall not in any manner affect the obligation of the Borrower to repay the Loans (both principal and unpaid accrued interest) of such Lender in accordance with the terms of this Agreement.
- (b) At the request of any Lender at any time, the Borrower agrees that it will execute and deliver to such Lender a Note payable to the order of such Lender.

Section 2.8. Optional Prepayments. The Borrower shall have the right at any time and from time to time to prepay any Borrowing, in whole or in part, without premium or penalty, by giving irrevocable written notice (or telephonic notice promptly confirmed in writing) to the Administrative Agent no later than (i) in the case of prepayment of any Eurodollar Borrowing, 11:00 a.m. (New York time) not less than three (3) Business Days prior to any such prepayment and (ii) in the case of any prepayment of any Base Rate Borrowing, 10:00 a.m. (New York time) on the date of such prepayment. Each such notice shall be irrevocable and shall specify the proposed date of such prepayment and the principal amount of each Borrowing or portion thereof to be prepaid. Upon receipt of any such notice, the Administrative Agent shall promptly notify each Lender of the contents thereof and of such Lender's Pro Rata Share of any such prepayment. If such notice is given, the aggregate amount specified in such notice shall be due and payable on the date designated in such notice, together with accrued interest to such date on the amount so prepaid in accordance with Section 2.10(d); provided, that if a Eurodollar Borrowing is prepaid on a date other than the last day of an Interest Period applicable thereto, the Borrower shall also pay all amounts required pursuant to Section 2.16. Each partial prepayment of any Loan shall be, in the case of Eurodollar Loans, in a minimum amount of \$3,000,000 (or lesser amount equal to outstanding Borrowings) and integral multiples of \$1,000,000 in excess thereof, and in the case of Base Rate Loans, in a minimum amount of \$1,000,000 (or lesser amount equal to outstanding Borrowings) and integral multiples of \$1,000,000 in excess thereof. Each prepayment of a Borrowing shall be applied ratably to the Loans comprising such Borrowing.

# Section 2.9. Mandatory Prepayments.

- (a) Each prepayment pursuant to this Section 2.9 shall be applied first to the Base Rate Loans to the full extent thereof, and second to the Eurodollar Loans to the full extent thereof.
- (b) If at any time (i) an Event of Loss with respect to the Project shall occur for which proceeds are received, or (ii) the Borrower shall, directly or indirectly, receive net proceeds from any sale or disposition of any asset or property (except for sales permitted by Section 6.3(c)), then the Borrower shall, immediately (in the case of clause (i) above, subject to the immediately following proviso) upon receipt of such proceeds, prepay the Obligations (other than Hedging Obligations) with 100% of such proceeds; provided that, if the Borrower shall have given the Administrative Agent written notice on or before the 60th day following the date of determination of the occurrence of such Event of Loss or within 90 days of the event giving rise to such determination, whichever period is shorter, stating that (i) it believes the Project could be repaired and/or restored and that the Restoration Conditions (as defined below) could be satisfied, and that the Borrower is diligently working to obtain the necessary information to meet the Restoration Conditions, the Borrower may defer such prepayment of such Obligations for a period of up to 30 additional days (and prepayment will not be required if the Restoration Conditions are met by the end of such period). From the date of the determination of the occurrence of such Event of Loss, the Borrower shall provide to the Administrative Agent a monthly status report describing the status of discussions with contractors and insurers or Governmental Authorities, as the case may be (including information on cost and scheduling estimates); and provided, further, that in the case of a prepayment required in accordance with

clause (i) above, such prepayment shall include accrued interest on the proceeds to the date of prepayment. As used in this Section 2.9(b), "Restoration Conditions" shall mean (a) no Default or Event of Default shall have occurred and be continuing and (b) the Borrower shall have delivered to the Administrative Agent (I) contracts for such repair or replacement demonstrating the Borrower's ability to effect such repair or replacement at a cost not greater than the insurance proceeds therefor (or, if such cost is greater, accompanied by an explanation of the source of funds for such excess amounts satisfactory to the Required Lenders), (II) cash-flow projections and other assurances satisfactory to the Required Lenders providing for the Borrower's ability to meet its obligations under the Loan Documents during the period from such loss until and following completion of such repair or replacement and (III) assurances that all Project Documents and Governmental Approvals shall remain in full force and effect during such period and thereafter to the satisfaction of the Required Lenders and that the Borrower and the Project are in compliance with all Governmental Requirements.

- (c) If the Borrower issues any debt or equity securities (other than (i) debt permitted pursuant to Section 6.1(a), (b), (c), or (d), or (ii) capital contributions by the Partners if such capital contributions are made for the purpose of funding capital expenditures of the Borrower (as determined in good faith by the Borrower and certified in writing to the Administrative Agent prior to the date of such capital contribution)), including debt issued or owing to an Affiliate of the Borrower, then no later than the Business Day following the date of receipt of the Net Cash Proceeds thereof, Borrower shall prepay the Obligations (other than Hedging Obligations) in an amount equal to all such Net Cash Proceeds.
- (d) Any prepayments made by the Borrower pursuant to paragraphs (b) or (c) above shall be applied as follows: <u>first</u>, to fees and reimbursable expenses of the Administrative Agent and the Collateral Agent then due and payable pursuant to any of the Loan Documents; <u>second</u>, to all reimbursable expenses, if any, of the Lenders then due and payable pursuant to any of the Loan Documents; <u>third</u>, to interest then due and payable on Loans made to the Borrower; and <u>fourth</u>, to the aggregate outstanding principal amount of the Loans, until the same shall have been paid in full, pro rata to the Lenders based on their respective Pro Rata Shares of the Loans.

### Section 2.10. Interest on Loans.

- (a) The Borrower shall pay interest on each Base Rate Loan at the Base Rate in effect from time to time and on each Eurodollar Loan at the Adjusted LIBO Rate for the applicable Interest Period in effect for such Loan, *plus*, in each case, the Applicable Margin in effect from time to time.
- (b) While an Event of Default exists or after acceleration, at the option of the Required Lenders, the Borrower shall pay interest ("**Default Interest**") with respect to all Eurodollar Loans then due (or overdue) and payable at the rate otherwise applicable for the then-current Interest Period *plus* an additional 2% per annum until the last day of such Interest Period, and thereafter, and with respect to all Base Rate Loans and all other Obligations then due (or overdue) and payable hereunder (other than Loans), at the rate in effect for Base Rate Loans, *plus* an additional 2% per annum.

- (c) Interest on the principal amount of all Loans shall accrue from and including the date such Loans are made to but excluding the date of any repayment thereof. Interest on all outstanding Base Rate Loans shall be payable quarterly in arrears on the last day of each March, June, September and December and on the Maturity Date. Interest on all outstanding Eurodollar Loans shall be payable on the last day of each Interest Period applicable thereto. Interest on any Loan which is converted into a Loan of another Type or which is repaid or prepaid shall be payable on the date of such conversion or on the date of any such repayment or prepayment (on the amount repaid or prepaid) thereof. All Default Interest shall be payable on demand.
- (d) The Administrative Agent shall determine each interest rate applicable to the Loans hereunder and shall promptly notify the Borrower and the Lenders of such rate in writing (or by telephone, promptly confirmed in writing). Any such determination shall be conclusive and binding for all purposes, absent manifest error.
- **Section 2.11.** <u>Fees</u>. The Borrower shall pay to the Administrative Agent for its own account fees in the amounts and at the times previously agreed upon in writing by the Borrower and the Administrative Agent.
- **Section 2.12.** Computation of Interest and Fees. Interest hereunder based on the Base Rate and all fees shall be computed on the basis of a year of 365 days (or 366 days in a leap year) and paid for the actual number of days elapsed (including the first day but excluding the last day). All other interest shall be computed on the basis of a year of 360 days and paid for the actual number of days elapsed (including the first day but excluding the last day).
  - Section 2.13. <u>Inability to Determine Interest Rates</u>. If prior to the commencement of any Interest Period for any Eurodollar Borrowing,
  - (i) the Administrative Agent shall have determined (which determination shall be conclusive and binding upon the Borrower) that, by reason of circumstances affecting the relevant interbank market, adequate means do not exist for ascertaining LIBOR for such Interest Period, or
  - (ii) the Administrative Agent shall have received notice from the Required Lenders that the Adjusted LIBO Rate does not adequately and fairly reflect the cost to such Lenders (or Lender, as the case may be) of making, funding or maintaining their (or its, as the case may be) Eurodollar Loans for such Interest Period,

the Administrative Agent shall give written notice (or telephonic notice, promptly confirmed in writing) to the Borrower and to the Lenders as soon as practicable thereafter. Until the Administrative Agent shall notify the Borrower and the Lenders that the circumstances giving rise to such notice no longer exist, (i) the obligations of the Lenders to continue or convert outstanding Loans as or into Eurodollar Loans shall be suspended and (ii) all such affected Loans shall be converted into Base Rate Loans on the last day of the then current Interest Period applicable thereto unless the Borrower prepays such Loans in accordance with this Agreement.

**Section 2.14.** <u>Illegality.</u> If any Change in Law shall make it unlawful or impossible for any Lender to make, maintain or fund any Eurodollar Loan and such Lender shall so notify

the Administrative Agent, the Administrative Agent shall promptly give notice thereof to the Borrower and the other Lenders, whereupon until such Lender notifies the Administrative Agent and the Borrower that the circumstances giving rise to such suspension no longer exist, the obligation of such Lender to continue or convert outstanding Loans as or into Eurodollar Loans, shall be suspended. If the affected Eurodollar Loan is then outstanding, such Loan shall be converted to a Base Rate Loan either (i) on the last day of the then current Interest Period applicable to such Eurodollar Loan if such Lender may lawfully continue to maintain such Loan to such date or (ii) immediately if such Lender shall determine that it may not lawfully continue to maintain such Eurodollar Loan to such date.

# Section 2.15. Increased Costs.

- (a) If any Change in Law shall:
- (i) impose, modify or deem applicable any reserve, special deposit or similar requirement that is not otherwise included in the determination of the Adjusted LIBO Rate hereunder against assets of, deposits with or for the account of, or credit extended by, any Lender (except any such reserve requirement reflected in the Adjusted LIBO Rate); or
- (ii) impose on any Lender or the eurodollar interbank market any other condition affecting this Agreement or any Eurodollar Loans made by such Lender;

and the result of either of the foregoing is to increase the cost to such Lender of making, converting into, continuing or maintaining a Eurodollar Loan or to reduce the amount received or receivable by such Lender hereunder (whether of principal, interest or any other amount), then the Borrower shall pay such additional amount or amounts sufficient to compensate such Lender, for such additional costs incurred or reduction suffered.

- (b) If any Lender shall have determined that on or after the date of this Agreement any Change in Law regarding capital requirements has or would have the effect of reducing the rate of return on such Lender's capital (or on the capital of such Lender's parent corporation) as a consequence of its obligations hereunder to a level below that which such Lender or such Lender's parent corporation could have achieved but for such Change in Law (taking into consideration such Lender's policies or the policies of such Lender's parent corporation with respect to capital adequacy) then, from time to time, the Borrower shall pay to such Lender such additional amounts as will compensate such Lender or such Lender's parent corporation for any such reduction suffered.
- (c) A certificate of a Lender setting forth the amount or amounts necessary to compensate such Lender or such Lender's parent corporation, as the case may be, specified in paragraph (a) or (b) of this Section shall be delivered to the Borrower (with a copy to the Administrative Agent and certifying that such Lender is similarly charging its other similarly situated borrowers) and shall be conclusive, absent manifest error. The Borrower shall pay any such Lender such amount or amounts within 10 days after receipt thereof.

(d) Failure or delay on the part of any Lender to demand compensation pursuant to this Section shall not constitute a waiver of such Lender's right to demand such compensation; <u>provided</u>, that the Borrower shall not be required to compensate a Lender under this Section for any increased costs or reductions incurred more than 90 days prior to the date that such Lender notifies the Borrower of such increased costs or reductions and of such Lender's intention to claim compensation therefor; <u>provided further</u>, that if the Change in Law giving rise to such increased costs or reductions is retroactive, then such 90-day period shall be extended to include the period of such retroactive effect.

Section 2.16. Funding Indemnity. In the event of (a) the payment of any principal of a Eurodollar Loan other than on the last day of the Interest Period applicable thereto (including as a result of an Event of Default), other than as a result of a mandatory prepayment required by Section 2.9, (b) the conversion or continuation of a Eurodollar Loan other than on the last day of the Interest Period applicable thereto, or (c) the failure by the Borrower to borrow, prepay, convert or continue any Eurodollar Loan on the date specified in any applicable notice (regardless of whether such notice is withdrawn or revoked), then, in any such event, unless waived by the Administrative Agent, the Borrower shall compensate each Lender, within ten (10) days after receipt from such Lender of the certificate referred to below, for any loss, cost or expense attributable to such event. Such loss, cost or expense shall be deemed to include an amount determined by such Lender to be the excess, if any, of (A) the amount of interest that would have accrued on the principal amount of such Eurodollar Loan if such event had not occurred at the Adjusted LIBO Rate applicable to such Eurodollar Loan for the period from the date of such event to the last day of the then current Interest Period therefor (or in the case of a failure to borrow, convert or continue, for the period that would have been the Interest Period for such Eurodollar Loan) over (B) the amount of interest that would accrue on the principal amount of such Eurodollar Loan for the same period if the Adjusted LIBO Rate were set on the date such Eurodollar Loan was prepaid or converted or the date on which the Borrower failed to borrow, convert or continue such Eurodollar Loan. A certificate as to any additional amount payable under this Section 2.16 submitted to the Borrower by any Lender (with a copy to the Administrative Agent) shall be conclusive, absent manifest error.

### Section 2.17. Taxes.

(a) Any and all payments by or on account of any obligation of the Borrower hereunder shall be made free and clear of and without deduction for any Indemnified Taxes or Other Taxes; <u>provided</u>, that if the Borrower shall be required to deduct any Indemnified Taxes or Other Taxes from such payments, then (i) the sum payable shall be increased as necessary so that after making all required deductions (including deductions applicable to additional sums payable under this Section) the Administrative Agent, the Collateral Agent or any Lender (as the case may be) shall receive an amount equal to the sum it would have received had no such deductions been made, (ii) the Borrower shall make such deductions and (iii) the Borrower shall pay the full amount deducted to the relevant Governmental Authority in accordance with applicable law.

(b) In addition, the Borrower shall pay any Other Taxes to the relevant Governmental Authority in accordance with applicable law.

- (c) The Borrower shall indemnify the Administrative Agent, the Collateral Agent and each Lender, within ten (10) days after written demand therefor, for the full amount of any Indemnified Taxes or Other Taxes paid by the Administrative Agent, the Collateral Agent or such Lender, as the case may be, on or with respect to any payment by or on account of any obligation of the Borrower hereunder (including Indemnified Taxes or Other Taxes imposed or asserted on or attributable to amounts payable under this Section) and any penalties, interest and reasonable expenses arising therefrom or with respect thereto, whether or not such Indemnified Taxes or Other Taxes were correctly or legally imposed or asserted by the relevant Governmental Authority. A certificate as to the amount of such payment or liability delivered to the Borrower by a Lender, or by the Administrative Agent or the Collateral Agent on its own behalf or on behalf of a Lender, shall be conclusive absent manifest error.
- (d) As soon as practicable after any payment of Indemnified Taxes or Other Taxes by the Borrower to a Governmental Authority, the Borrower shall deliver to the Administrative Agent the original or a certified copy of a receipt issued by such Governmental Authority evidencing such payment, a copy of the return reporting such payment or other evidence of such payment reasonably satisfactory to the Administrative Agent.
- (e) Any Foreign Lender that is entitled to an exemption from or reduction of withholding tax under the Code or any treaty to which the United States is a party, with respect to payments under this Agreement shall deliver to the Borrower (with a copy to the Administrative Agent), at the time or times prescribed by applicable law, such properly completed and executed documentation prescribed by applicable law or reasonably requested by the Borrower as will permit such payments to be made without withholding or at a reduced rate. Without limiting the generality of the foregoing, each Foreign Lender agrees that it will deliver to the Administrative Agent and the Borrower (or in the case of a Participant, to the Lender from which the related participation shall have been purchased), as appropriate, two (2) duly completed copies of (i) Internal Revenue Service Form W-8 ECI, or any successor form thereto, certifying that the payments received from the Borrower hereunder are effectively connected with such Foreign Lender's conduct of a trade or business in the United States; or (ii) Internal Revenue Service Form W-8 BEN, or any successor form thereto, certifying that such Foreign Lender is entitled to benefits under an income tax treaty to which the United States is a party which reduces the rate of withholding tax on payments of interest; or (iii) Internal Revenue Service Form W-8 BEN, or any successor form prescribed by the Internal Revenue Service, together with a certificate (A) establishing that the payment to the Foreign Lender qualifies as "portfolio interest" exempt from U.S. withholding tax under Code section 871(h) or 881(c), and (B) stating that (1) the Foreign Lender is not a bank for purposes of Code section 881(c)(3)(A), or the obligation of the Borrower hereunder is not, with respect to such Foreign Lender, a loan agreement entered into in the ordinary course of its trade or business, within the meaning of that section; (2) the Foreign Lender is not a 10% shareholder of the Borrower within the meaning of Code section 871(h)(3) or 881(c)(3)(B); and (3) the Foreign Lender is not a controlled foreign corporation that is related to the Borrower within the meaning of Code section 881(c)(3)(C); or (iv) such other Internal Revenue Service forms as may be applicable to the Foreign Lender, including Forms W-8 IMY or W-8 EXP. Each such Foreign Lender shall deliver to the Borrower and the Administrative Agent such forms on or before the date that it becomes a party to this Agreement (or in the case of a Participant, on or before the date such Participant

purchases the related participation). In addition, each such Foreign Lender shall deliver such forms promptly upon the obsolescence or invalidity of any form previously delivered by such Foreign Lender. Each such Foreign Lender shall promptly notify the Borrower and the Administrative Agent at any time that it determines that it is no longer in a position to provide any previously delivered certificate to the Borrower (or any other form of certification adopted by the Internal Revenue Service for such purpose).

# Section 2.18. Payments Generally; Pro Rata Treatment; Sharing of Set-offs.

- (a) The Borrower shall make each payment required to be made by it hereunder (whether of principal, interest or fees, or of amounts payable under Section 2.15, 2.16 or 2.17, or otherwise) prior to 12:00 noon (New York time), on the date when due, in immediately available funds, free and clear of any defenses, rights of set-off, counterclaim, or withholding or deduction of taxes. Any amounts received after such time on any date may, in the discretion of the Administrative Agent, be deemed to have been received on the next succeeding Business Day for purposes of calculating interest thereon. All such payments shall be made to the Administrative Agent at the Payment Office except that payments pursuant to Section 2.15, 2.16 or 2.17 and 10.3 shall be made directly to the Persons entitled thereto. The Administrative Agent shall distribute any such payments received by it for the account of any other Person to the appropriate recipient promptly following receipt thereof. If any payment hereunder shall be due on a day that is not a Business Day, the date for payment shall be extended to the next succeeding Business Day, and, in the case of any payment accruing interest, interest thereon shall be made payable for the period of such extension. All payments hereunder shall be made in Dollars.
- (b) If at any time insufficient funds are received by and available to the Administrative Agent to pay fully all amounts of principal, interest and fees then due hereunder, such funds shall be applied (i) first, towards payment of interest and fees then due hereunder, ratably among the parties entitled thereto in accordance with the amounts of interest and fees then due to such parties, and (ii) second, towards payment of principal then due hereunder, ratably among the parties entitled thereto in accordance with the amounts of principal then due to such parties.
- (c) Unless the Administrative Agent shall have received notice from the Borrower prior to the date on which any payment is due to the Administrative Agent for the account of the Lenders hereunder that the Borrower will not make such payment, the Administrative Agent may assume that the Borrower has made such payment on such date in accordance herewith and may, in reliance upon such assumption, distribute to the Lenders the amount or amounts due. In such event, if the Borrower has not in fact made such payment, then each of the Lenders severally agrees to repay to the Administrative Agent forthwith on demand the amount so distributed to such Lender with interest thereon, for each day from and including the date such amount is distributed to it to but excluding the date of payment to the Administrative Agent, at the greater of the Federal Funds Effective Rate and a rate determined by the Administrative Agent in accordance with banking industry rules on interbank compensation.
- (d) If any Lender shall fail to make any payment required to be made by it pursuant to <u>Section 2.3(b)</u>, <u>2.21(c)</u>, or <u>10.3(d)</u>, then the Administrative Agent may, in its

discretion (notwithstanding any contrary provision hereof), apply any amounts thereafter received by the Administrative Agent for the account of such Lender to satisfy such Lender's obligations under such Sections until all such unsatisfied obligations are fully paid.

### Section 2.19. Mitigation of Obligations; Replacement of Lenders.

- (a) If any Lender requests compensation under Section 2.15, or if the Borrower is required to pay any additional amount to any Lender or any Governmental Authority for the account of any Lender pursuant to Section 2.13, then such Lender shall use reasonable efforts to designate a different lending office for funding or booking its Loan hereunder or to assign its rights and obligations hereunder to another of its offices, branches or affiliates, if, in the judgment of such Lender, such designation or assignment (i) would eliminate or reduce amounts payable pursuant to Section 2.13 or Section 2.15, as the case may be, in the future and (ii) would not subject such Lender to any unreimbursed cost or expense and would not otherwise be disadvantageous to such Lender. The Borrower hereby agrees to pay all reasonable costs and expenses incurred by any Lender in connection with any such designation or assignment.
- (b) If any Lender requests compensation under Section 2.15, or if the Borrower is required to pay any additional amount to any Lender or any Governmental Authority for the account of any Lender pursuant to Section 2.13, or if any Lender defaults in its obligation to fund Loans hereunder, then the Borrower may, at its sole expense and effort, upon notice to such Lender and the Administrative Agent, require such Lender to assign and delegate, without recourse (in accordance with and subject to the restrictions contained in Section 10.4), all its interests, rights and obligations under this Agreement to an assignee that shall assume such obligations (which assignee may be another Lender, if a Lender accepts such assignment); provided that (i) the Borrower shall have received the prior written consent of the Administrative Agent, which consent shall not unreasonably be withheld, (ii) such Lender shall have received payment of an amount equal to the outstanding principal of its Loans, accrued interest thereon, accrued fees and all other amounts payable to it hereunder, from the assignee (to the extent of such outstanding principal and accrued interest and fees) or the Borrower (in the case of all other amounts), (iii) in the case of any such assignment resulting from a claim for compensation under Section 2.15 or payments required to be made pursuant to Section 2.13, such assignment will result in a reduction in such compensation or payments and (iv) this Section 2.19(b) shall have no effect during any period when there are two or fewer Lenders. A Lender shall not be required to make any such assignment and delegation cease to apply.

### **ARTICLE III**

# **CONDITIONS PRECEDENT TO LOANS**

**Section 3.1.** <u>Conditions To Effectiveness</u>. The obligations of the Lenders to make Loans shall not become effective until the date on which each of the following conditions is satisfied (or waived in accordance with <u>Section 10.2</u>).

- (a) The Administrative Agent shall have received all fees and other amounts due and payable on or prior to the Effective Date, including reimbursement or payment of all out-of-pocket expenses (including reasonable fees, charges and disbursements of counsel to the Administrative Agent as set forth in Section 10.3(a)) required to be reimbursed or paid by the Borrower hereunder, under any other Loan Document and under any agreement with the Administrative Agent, the Collateral Agent or SunTrust Capital Markets, Inc., as Arranger.
  - (b) The Administrative Agent (or its counsel) and the Collateral Agent shall have received the following:
  - (i) a counterpart of this Agreement signed by or on behalf of each party hereto or written evidence satisfactory to the Administrative Agent (which may include telecopy transmission of a signed signature page of this Agreement) that such party has signed a counterpart of this Agreement;
    - (ii) duly executed Notes payable to each Lender (if requested);
  - (iii) the duly executed Security Agreement, together with (A) UCC-1 financing statements (or assignments or amendments thereof) and other applicable documents under the laws of the jurisdictions with respect to the perfection of the Liens granted under the Security Agreement, as requested by the Collateral Agent in order to perfect such Liens, duly executed by the Borrower (if applicable) and (B) copies of favorable UCC, tax, judgment and fixture lien search reports in all necessary or appropriate jurisdictions and under all legal and trade names of the Borrower reasonably requested by the Lenders, indicating that there are no prior Liens on any of the Collateral other than Permitted Encumbrances;
  - (iv) duly executed Control Account Agreements with each bank that maintains deposit accounts, and each securities intermediary that maintains investment accounts, on behalf of the Borrower on the Effective Date;
  - (v) the duly executed Pledge Agreements, together with (if applicable) (A) original certificates evidencing the issued and outstanding Capital Stock pledged to the Collateral Agent pursuant to the Pledge Agreement, and (B) stock powers or other appropriate instruments of transfer executed in blank;
  - (vi) Mortgages covering all of the Real Estate (subject to exceptions specifically identified to the Collateral Agent by the Borrower and agreed to by the Collateral Agent in its sole discretion), duly completed and executed (as applicable) in sufficient number of counterparts and in proper form for recording and, if reasonably requested, such further supporting evidence of title, legal opinions and evidence that such Mortgages will create a valid and enforceable first priority Lien (except as permitted by Section 6.2) on such Real Estate in favor of the Collateral Agent for the benefit of the Lenders;
  - (vii) All property in which the Collateral Agent shall, at such time, be entitled to have a Lien pursuant to this Agreement or any Loan Document (physically delivered to

the possession of the Collateral Agent or any bailee accepted by the Collateral Agent) to the extent that such possession is necessary for the purpose of perfecting the Collateral Agent's Lien in such Collateral;

- (viii) copies of a duly executed letter, in form and substance satisfactory to Collateral Agent, executed by each of the Existing Lenders or the agent thereof, together with (a) UCC-3 or other appropriate termination, amendment or assignment statements, in form and substance satisfactory to Collateral Agent, releasing (or continuing in favor of the Lenders) all liens of the Existing Lenders upon any of the personal property of the Borrower, (b) cancellations and releases, or amendments or assignments, in form and substance satisfactory to the Collateral Agent, releasing (or continuing in favor of the Lenders) all liens of the Existing Lenders upon any of the real property of the Borrower, and (c) any other releases, terminations, similar amendments or assignments or other documents reasonably required by the Collateral Agent to evidence the refinancing in full, as and to the extent set forth in Section 10.15, of Indebtedness owed to the Existing Lenders;
- (ix) a certificate of the Secretary or Assistant Secretary of the Borrower and the Partners, attaching and certifying copies of its bylaws and of the resolutions of its boards of directors, or partnership agreement or limited liability company agreement, or comparable organizational documents and authorizations, authorizing the execution, delivery and performance of the Loan Documents to which it is a party and certifying the name, title and true signature of each officer of such Person executing the Loan Documents to which it is a party;
- (x) certified copies of the articles or certificate of incorporation, certificate of organization or limited partnership, or other registered organizational documents of the Borrower and the Partners, together with certificates of good standing or existence, as may be available from the Secretary of State of the jurisdiction of organization of such Person and each other jurisdiction where such Person is required to be qualified to do business as a foreign corporation;
- (xi) a favorable written opinion of Bracewell & Giuliani LLP, counsel to the Borrower, Enterprise CHOPS and Manta Ray, addressed to the Administrative Agent and each of the Lenders, and covering such matters relating to the Borrower, Enterprise CHOPS and Manta Ray, the Loan Documents and the transactions contemplated therein, as the Administrative Agent or the Required Lenders shall reasonably request; a favorable written opinion of Carol F. Melcher, Associate General Counsel of the Borrower, covering such matters relating to the Borrower, the Partners and Manta Ray, the Loan Documents and the transactions contemplated therein, as the Administrative Agent or the Required Lenders shall reasonably request; a favorable written opinion of Baker Botts L.L.P., counsel to Valero CHOPS, addressed to the Administrative Agent and each of the Lenders, and covering such matters relating to Valero CHOPS, the Loan Documents and the transactions contemplated therein, as the Administrative Agent or the Required Lenders shall reasonably request; and a favorable written opinion of Jay Browning, Vice President Corporate Law and Secretary of Valero, covering such

matters relating to Valero CHOPS, the Loan Documents and the transactions contemplated therein, as the Administrative Agent or the Required Lenders shall reasonably request;

- (xii) a certificate, dated the Effective Date and signed by a Responsible Officer, certifying that (x) no Default or Event of Default exists, (y) all representations and warranties of the Borrower and the Partners set forth in the Loan Documents are true and correct in all material respects, and (z) since December 31, 2004, there shall have been no change which has had or could reasonably be expected to have a Material Adverse Effect;
  - (xiii) a duly executed funds disbursement agreement, together with a report setting forth the sources and uses of the proceeds hereof;
- (xiv) certified (by a Responsible Officer or a secretary of the Borrower) copies of (a) all agreements, indentures or notes governing the terms of any Material Indebtedness, (b) all Material Project Documents, (c) the Assignment (CHOPS), and (d) the Participation Agreement, which together shall constitute all material agreements, documents and instruments to which the Borrower, the Partners or any of their respective assets are bound;
- (xv) certificates of insurance, in form and detail acceptable to the Collateral Agent, describing the types and amounts of insurance (property and liability) covering any of the tangible insurable Collateral maintained by the Borrower, and in each case to the extent required by Section 5.5 naming the Collateral Agent as loss payee or additional insured, as the case may be, together with a lender's loss payable endorsement in form and substance satisfactory to the Collateral Agent;
- (xvi) (a) a certificate, dated the Effective Date and signed by the chief financial officer (or another officer reasonably acceptable to the Administrative Agent) of the Borrower and each Partner, confirming the solvency of the Borrower and each Partner before and after giving effect to all transactions contemplated to occur on the Effective Date by the Loan Documents (including the refinancing of Existing Debt with Loans), together with (b) the Projections;
- (xvii) copies of (A) the internally prepared quarterly financial statements of Borrower for the Fiscal Quarter ending on March 31, 2005, and (B) the audited financial statements for Borrower and its subsidiaries for the Fiscal Year ending December 31, 2004;
  - (xviii) satisfactory evidence that the Partners own directly or indirectly 100% of the legal and beneficial interests in the Capital Stock of the Borrower.
  - (xix) satisfactory evidence regarding the scope and materiality of any environmental risks affecting the Borrower;
- (xx) each Effective Date Purchase and Sale Agreement, reflecting the dedication by the Material Producer party thereto of specified volumes of Dedicated

Production (as defined in such Purchase and Sale Agreements) for delivery on the Cameron Highway Oil Pipeline, the terms of which dedication shall be reasonably satisfactory to the Administrative Agent (it being understood that the terms of the copies of the Effective Date Purchase and Sale Agreements initially provided to the Administrative Agent in connection with its decision to enter into this Agreement were satisfactory to the Administrative Agent);

- (xxi) the Sponsor LOCs, each issued in form and from an issuer reasonably satisfactory to the Administrative Agent and noting the Administrative Agent as beneficiary;
  - (xxii) a notice, substantially in the form of Exhibit H, from the Borrower to each Material Producer that has executed a Consent; and
  - (xxiii) a duly executed Manta Ray Consent.
- (c) the making of the Loans does not violate any Governmental Requirement on the date of or immediately following the Loan and is not enjoined, temporarily, preliminarily or permanently.
- Section 3.2. <u>Delivery of Documents</u>. All of the Loan Documents, certificates, legal opinions and other documents and papers referred to in this <u>Article III</u>, unless otherwise specified, shall be delivered to the Administrative Agent and the Collateral Agent for the account of each of the Lenders and, except for the Notes, in sufficient counterparts or copies for each of the Lenders and shall be in form and substance satisfactory in all respects to the Administrative Agent and the Collateral Agent.

# ARTICLE IV

# **REPRESENTATIONS AND WARRANTIES**

The Borrower represents and warrants to the Administrative Agent and each Lender as follows:

**Section 4.1.** <u>Organization; Powers</u>. The Borrower is duly organized, validly existing and in good standing under the laws of the jurisdiction of its organization or formation, has all requisite power and authority to carry on its business as now conducted and, except where the failure to do so, individually or in the aggregate, could not reasonably be expected to result in a Material Adverse Effect, is qualified to do business in, and is in good standing in, every jurisdiction where such qualification is required.

### Section 4.2. Authorization; Enforceability.

(a) The execution, delivery and performance by the Borrower of each Transaction Document to which it is a party are within its powers, have been duly authorized by all necessary organizational, and if required, Partner action and do not contravene any provision of its organizational documents, any Governmental Requirement or contractual obligation binding on

or affecting it or any of its properties or assets that could reasonably be expected to have a Material Adverse Effect, and such execution, delivery and performance do not and will not result in or require the creation of any Lien upon or with respect to any of its properties, other than Permitted Encumbrances and the other Liens created or permitted by the Loan Documents, or result in the acceleration of any obligation under, or in a condition or event that constitutes (or that, upon notice or lapse of time or both, would constitute) an event of default under any material contractual obligation of such party. This Agreement has been, and each other Loan Document to which the Borrower, the Partners or the Sponsors will be a party when delivered hereunder will have been, duly executed and delivered by the Borrower, the Partners or the Sponsors, as the case may be.

(b) This Agreement and each other Loan Document to which the Borrower is a party are legal, valid and binding obligations of the Borrower enforceable against the Borrower in accordance with their respective terms, subject to the qualification that the enforcement of the rights and remedies herein and therein is subject to (i) bankruptcy and other similar laws of general application affecting rights and remedies of creditors and (ii) the application of general principles of equity (regardless of whether considered in a proceeding in equity or at law). The Borrower has delivered to the Lenders and the Administrative Agent (or the Collateral Agent, as applicable) true and complete copies of each executed Material Document, and except as disclosed to the Lenders and the Administrative Agent (or the Collateral Agent, as applicable) in writing, none of the Material Documents have been amended, modified or terminated. The Borrower has disclosed in writing to the Lenders and the Administrative Agent all notices of default under any Transaction Document. Neither the Borrower, the Partners, the Sponsors nor, to the Borrower's knowledge, any other Transaction Party is in default under any other executed Transaction Document, except (i) defaults that could not reasonably be expected to have a Material Adverse Effect and (ii) defaults by other Transaction Parties (other than the Borrower, the Partners and the Sponsors) under Additional Purchase and Sale Agreements.

**Section 4.3.** Other Material Agreements. Other than as permitted hereby, the Borrower is not a party to any agreement providing for, evidencing, securing or otherwise relating to any Indebtedness of the Borrower, and all obligations of the Borrower to issuers of surety or appeal bonds have been issued for account of the Borrower except to the extent permitted by the Transaction Documents.

# Section 4.4. Governmental Approvals; Material Permits; No Conflicts.

(a) All material Governmental Approvals, including without limitation, all Material Permits, that are at any time required to be obtained or made by the Borrower in connection with (i) the development, use, ownership, financing and maintenance of the Cameron Highway System and (ii) any Loans and granting of Liens under, and the execution, delivery and performance by the Borrower of, the then existing Transaction Documents, have been (or will be at such time as it is necessary to be) obtained and are in full force and effect. No material change in the facts or circumstances reported or assumed in the applications for the granting of such Material Permits exists (except such change which could not reasonably be expected to result in

any revocation of such Material Permits or any material fine, penalty or other action being taken against the Borrower by any Governmental Authority), and there are not any proceedings pending or, to the knowledge of the Borrower, threatened which would reasonably be expected to jeopardize the validity of such Material Permits. The Borrower does not have any reason to believe that it will be unable to obtain or maintain the Material Permits in the ordinary course of business and at such time or times as may be necessary to avoid any substantial delay in, or material impairment to, the consummation and performance of the Transactions.

- (b) All Governmental Approvals and other filings, recordings, registrations and other actions have been made, obtained and taken in all relevant jurisdictions that are necessary to create and perfect the Liens provided for in the Security Documents, and the Collateral Agent, for the benefit of itself and Lenders, will hold the Liens provided for in the Security Documents, and the Security Documents shall constitute a valid, direct, continuing first priority Lien on the Collateral, subject only to Permitted Encumbrances and other Liens permitted by the Loan Documents.
- (c) The execution, delivery and performance by the Borrower of the Loan Documents, including this Agreement, and the Material Project Documents do not contravene or violate any provision of the organizational documents of the Borrower.

# Section 4.5. Financial Condition; No Material Adverse Change; Material Obligations.

- (a) On the date of the initial Loan, the financial statements delivered pursuant to <u>Section 3.1(b)(xvii)</u> present fairly, in all material respects, the financial position of the Borrower as of such date in accordance with GAAP, subject to year-end adjustments and the absence of footnotes.
- (b) Since December 31, 2004, there has been no material adverse change in the business, assets, operations, prospects or condition, financial or otherwise, of the Borrower or the Project. As of the Effective Date, the Borrower has no material assets other than the Project Documents and the other assets listed on Schedule 4.5(b) and has not entered into any material agreements or incurred any material obligations other than as contemplated by this Agreement or the Project Documents.

# Section 4.6. Properties.

- (a) The Borrower has marketable title to the property material to its business, free and clear of all Liens except for Permitted Encumbrances or Liens otherwise permitted or contemplated by this Agreement or the other Loan Documents.
- (b) The Borrower owns, or is licensed to use, all trademarks, tradenames, copyrights, patents and other intellectual property material to its business or to the construction and operation of the Cameron Highway System, and the use thereof by the Borrower does not infringe upon the rights of any other Person, except for any such infringements that, individually or in the aggregate, could not reasonably be expected to result in a Material Adverse Effect.

# Section 4.7. Litigation and Environmental Matters.

- (a) There is no action, suit or proceeding by or before any arbitrator or Governmental Authority pending against or, to the knowledge of the Borrower, threatened against or affecting the Borrower which (i) could reasonably be expected to result in a Material Adverse Effect or (ii) challenges the validity of this Agreement, any Note, or any other Loan Document or Project Document.
- (b) The Borrower's conduct of operations on its properties is in substantial compliance with all applicable orders of any court or Governmental Authority and all applicable Environmental Laws, except where the failure to comply with such orders or Environmental Laws could not reasonably be expected to result in a Material Adverse Effect. The Borrower (i) has not failed to obtain, maintain or comply with any permit, license or other approval required under any Environmental Law, (ii) has not become subject to any Environmental Liability, (iii) has not received notice of any claim with respect to any Environmental Liability and (iv) does not know of any basis for any Environmental Liability, in each case that could reasonably be expected to result in a Material Adverse Effect.
- (c) All Hazardous Materials or solid waste generated at any and all property of the Borrower which have in the past been transported, treated and disposed of by the Borrower have been so transported, treated and disposed of only by carriers maintaining valid permits under applicable Environmental Laws, except to the extent the failure to have such Hazardous Materials transported, treated or disposed by such carriers could not reasonably be expected to have a Material Adverse Effect, and only at treatment, storage and disposal facilities which, to the Borrower's actual knowledge, have maintained valid permits under applicable Environmental Laws, and have been and are operating in compliance with such permits, except to the extent the failure to have such Hazardous Materials treated, stored or disposed of at such facilities, or the failure of such carriers or facilities to so operate, could not reasonably be expected to have a Material Adverse Effect.
- (d) To the Borrower's actual knowledge, no Hazardous Materials have been disposed of or otherwise released and there has been no threatened release of any Hazardous Materials on or to any property of the Borrower except in compliance with Environmental Laws, except to the extent the failure to do so could not reasonably be expected to have a Material Adverse Effect.
- **Section 4.8.** Compliance with Laws and Agreements. The Borrower is in compliance with all Governmental Requirements applicable to it or its property and all indentures, agreements and other instruments binding upon it or its property, except where the failure to do so could not reasonably be expected to result in a Material Adverse Effect.
- **Section 4.9.** <u>Investment Company Act Status</u>. The Borrower is not an "investment company" or a company "controlled" by an "investment company" that is incorporated in or organized under the laws of the United States of America or any "State," as those terms are defined in the Investment Company Act of 1940, as amended. The execution and delivery by the Borrower of this Agreement and the other Loan Documents and its performance

of the obligations provided for therein, will not result in a violation of the Investment Company Act of 1940, as amended.

**Section 4.10.** <u>Public Utility Holding Company Act Status</u>. The Borrower is not a "holding company" or a "subsidiary company" of a "holding company" or an "affiliate" of a "holding company" or of a "subsidiary company" of a "holding company" or a "public utility" within the meaning of the Public Utility Holding Company Act of 1935, as amended.

**Section 4.11.** <u>Taxes</u>. The Borrower has (a) timely filed or caused to be filed all Tax returns and reports required to have been filed, and (b) paid or caused to be paid all Taxes levied upon it or any of its property or income which are due and payable, including interest and penalties, except (i) such Taxes, interest and penalties as are being contested in good faith by appropriate proceedings and for which the Borrower has set aside on its books adequate reserves for the payment thereof or (ii) to the extent that the failure to do so could not reasonably be expected to result in a Material Adverse Effect.

**Section 4.12.** ERISA. No ERISA Reportable Event has occurred or is reasonably expected to occur that, when taken together with all other such ERISA Reportable Events for which liability is reasonably expected to occur, could reasonably be expected to result in a Material Adverse Effect. Neither the Borrower, nor any ERISA Affiliate sponsors, maintains or contributes to, or has at any time in the six-year period preceding the Effective Date sponsored, maintained or contributed to, any Plan or any Multiemployer Plan.

#### Section 4.13. <u>Disclosure</u>.

- (a) The Borrower has disclosed to the Lenders and the Administrative Agent all material agreements, documents and instruments to which the Borrower, the Partners or any of their respective assets are bound.
- (b) None of the statements, exhibits, documents, reports, financial statements, certificates or other information furnished by or on behalf of the Borrower to any Lender, the Administrative Agent or the Collateral Agent in connection with the negotiation of this Agreement or any other Loan Document or delivered hereunder or in any other Transaction Document (as modified or supplemented by other information so furnished) contains any material misstatement of fact or omits to state any material fact necessary to make the statements therein, in the light of the circumstances under which they were made, not misleading (other than omissions that pertain to matters of a general economic nature); provided, however, it is understood that, with respect to such statements, exhibits, documents, reports, financial statements, certificates or other information, any estimates or projections contained therein are based upon information that was currently available and believed to be correct on the Effective Date and upon assumptions believed to be reasonable on the Effective Date and the Borrower does not warrant that such estimates and projections will ultimately prove to have been accurate. The representations and warranties made by the Borrower, the Partners and the Sponsors in each Material Document are true and correct in all material respects as of the date made or deemed made, and, to the best of the Borrower's knowledge, the representations and warranties made by the Transaction Parties (other than the Borrower, the Partners and the Sponsors) in each Material Document are true and correct in all material respects as of the date made. The

representations and warranties made by each of the Borrower, the Partners and the Sponsors in each Transaction Document (other than the Material Documents) are true and correct in all material respects as of the date made or deemed made except as could not reasonably be expected to have a Material Adverse Effect, and, to the best of the Borrower's knowledge, the representations and warranties made by the Transaction Parties (other than the Borrower, the Partners and the Sponsors) in each such Transaction Document (other than the Additional Purchase and Sale Agreements) are true and correct in all material respects as of the date made or deemed made except as could not reasonably be expected to have a Material Adverse Effect. There is no fact known to the Borrower that has not been disclosed in writing to the Lenders and the Administrative Agent that could reasonably be expected to have a Material Adverse Effect.

## Section 4.14. Business of the Borrower.

- (a) The sole business of the Borrower is the ownership, development, construction, financing and operation of the Cameron Highway System.
- (b) All insurance required by <u>Section 5.5</u> is in full force and effect.
- (c) No event of force majeure under any Transaction Document has occurred and is continuing that could reasonably be expected to result in the termination of any Transaction Document or to have a Material Adverse Effect. Neither the business nor the property of the Borrower has been affected in a manner that has had or could reasonably be expected to have a Material Adverse Effect as a result of any Taking.
- (d) All easements, leaseholds, other property interests, all utility and other services, means of transportation, facilities, other materials and other rights that are necessary for the acquisition, development, construction, installation and operation of the Cameron Highway System in accordance with all Governmental Requirements and the Transaction Documents (including, without limitation, gas, electrical, water and sewage services and facilities) have been procured pursuant to Transaction Documents or, to the best of the Borrower's knowledge, are otherwise commercially available to the Project and, to the extent appropriate, arrangements have been made on commercially reasonable terms for such easements, interests, services, means of transportation, facilities, materials and rights.
- (e) None of the proceeds from the Borrowings will be used, directly or indirectly, for the purpose of buying or carrying any "margin stock" within the meaning of Regulation U or X or any other regulation of the Board of Governors of the Federal Reserve System or to violate Section 7 of the Securities Exchange Act of 1934 or any rule or regulation thereunder, in each case as now in effect or as the same may hereinafter be in effect.

#### **Section 4.15. Subsidiaries.** The Borrower has no Subsidiaries.

**Section 4.16.** OFAC. None of the Borrower, the Partners or the Sponsors (a) is a person whose property or interest in property is blocked or subject to blocking pursuant to Section 1 of Executive Order 13224 of September 23, 2001 Blocking Property and Prohibiting Transactions With Persons Who Commit, Threaten to Commit, or Support Terrorism (66 Fed. Reg. 49079 (2001)), (b) engages in any dealings or transactions prohibited by Section 2 of such

executive order, or is otherwise associated with any such person in any manner violative of Section 2, or (c) is a person on the list of Specially Designated Nationals and Blocked Persons or subject to the limitations or prohibitions under any other U.S. Department of Treasury's Office of Foreign Assets Control regulation or executive order.

Section 4.17. Patriot Act. Each of the Borrower, the Partners and the Sponsors is in compliance, in all material respects, with (a) the Trading with the Enemy Act, as amended, and each of the foreign assets control regulations of the United States Treasury Department (31 CFR, Subtitle B, Chapter V, as amended) and any other enabling legislation or executive order relating thereto, and (b) the Uniting And Strengthening America By Providing Appropriate Tools Required To Intercept And Obstruct Terrorism (USA Patriot Act of 2001). No part of the proceeds of the Loans will be used, directly or indirectly, for any payments to any governmental official or employee, political party, official of a political party, candidate for political office, or anyone else acting in an official capacity, in order to obtain, retain or direct business or obtain any improper advantage, in violation of the United States Foreign Corrupt Practices Act of 1977, as amended.

#### ARTICLE V

#### **AFFIRMATIVE COVENANTS**

Until the Commitments have expired or been terminated and the principal of and interest on each Loan and all fees payable hereunder or under the other Loan Documents shall have been paid in full, the Borrower covenants and agrees with the Lenders that:

#### Section 5.1. Financial Statements and Other Information.

- (a) The Borrower will furnish to the Administrative Agent and each Lender:
- (i) as soon as available and in any event within 120 days after the end of each Fiscal Year, an audited balance sheet of the Borrower as at the end of such year and the related statements of operations, partners' equity and cash flows of the Borrower for such year, setting forth in each case in comparative form the figures for the previous Fiscal Year, all in reasonable detail and accompanied by a report thereon of Deloitte or other independent public accountants of recognized national standing (without a "going concern" or like qualification or exception and without any qualification or exception as to the scope of such audit) to the effect that such financial statements present fairly, in all material respects, the financial condition of the Borrower as at the end of such Fiscal Year, and the results of operations and cash flows for such Fiscal Year in accordance with GAAP consistently applied;
- (ii) as soon as possible and in any event within 60 days after the end of each of the first three Fiscal Quarters of each Fiscal Year, an unaudited balance sheet of the Borrower as at the end of such quarter and the related statements of operations and partners' equity for such Fiscal Quarter and cash flows for the then elapsed portion of the Fiscal Year, setting forth in each case in comparative form the figures for the corresponding period or periods of (or, in the case of the balance sheet, as of the end of)

the previous Fiscal Year, all in reasonable detail and certified by a Financial Officer of the Borrower that such financial statements fairly present, in all material respects the financial condition of the Borrower as at the end of such Fiscal Quarter, and the results of operations for such Fiscal Quarter and the then elapsed portion of the Fiscal Year, and cash flows for such Fiscal Year-to-date, of the Borrower in accordance with GAAP consistently applied, subject to normal, year-end audit adjustments and the absence of MD&A and footnotes;

- (iii) concurrently with any delivery of financial statements under clause (i) or (ii) above, a certificate of a Financial Officer of the Borrower (A) certifying as to whether a Default has occurred and, if a Default has occurred, specifying the details thereof and any action taken or proposed to be taken with respect thereto, and (B) stating whether any change in GAAP or in the application thereof has occurred since the Effective Date and, if any such change has occurred, specifying the effect of such change on the financial statements accompanying such certificate;
- (iv) as soon as possible and in any event within five days after the occurrence of any Default or Event of Default, a certificate of a Financial Officer of the Borrower setting forth details of such Default or Event of Default and the action which the Borrower proposes to take with respect thereto;
- (v) all material written communications given or received by the Borrower relating to any claims or proceedings with respect to any Governmental Approval or amending, modifying or affecting any Governmental Approval then required to be in effect;
- (vi) written notice of the occurrence of any event giving rise to a claim under any insurance policy in an amount greater than \$5,000,000 together with copies of any document relating thereto, including copies of any such claim, in the possession or control of the Borrower or any agent of the Borrower; and
- (vii) promptly following any request therefor, such other information regarding the operations, business affairs and financial condition of the Borrower, or compliance with the terms of this Agreement or any Loan Agreement, as the Administrative Agent or any Lender may reasonably request.

The electronic posting of any financial statements, reports, notices or other items required to be furnished pursuant to this <u>Section 5.1</u> on a website established for Lender and Administrative access shall constitute delivery for all purposes of this <u>Section 5.1</u>.

**Section 5.2.** <u>Notices of Material Events</u>. The Borrower will furnish to each Lender written notice of the following promptly after the Borrower becomes aware thereof:

(a) the filing or commencement of any action, suit or proceeding by or before any arbitrator or Governmental Authority against or affecting the Borrower or any Affiliate thereof that, if adversely determined, could reasonably be expected to result in a Material Adverse Effect;

- (b) written notice of the occurrence of the commencement of any proceeding before any Governmental Authority seeking to subject the Cameron Highway System to the jurisdiction of any Governmental Authority;
- (c) the occurrence of any ERISA Reportable Event that, alone or together with any other ERISA Reportable Events that have occurred, could reasonably be expected to result in a Material Adverse Effect;
  - (d) any termination or event of default notice under any Transaction Document;
- (e) the Borrower obtaining actual knowledge that BP is engaged in ongoing, bona fide negotiations relating to the transfer of, or is transferring or has transferred, its obligations under the BP Purchase and Sale Agreement, in whole or in part; and
  - (f) any other development that results in, or could reasonably be expected to result in, a Material Adverse Effect.

Each notice delivered under this <u>Section 5.2</u> shall be accompanied by a statement of a Financial Officer of the Borrower setting forth the details of the event or development requiring such notice and any action taken or proposed to be taken with respect thereto.

- **Section 5.3.** Existence; Conduct of Business. The Borrower will do or cause to be done all things necessary and lawful in order to preserve, renew and keep in full force and effect its legal existence and the rights, licenses, permits, privileges and franchises material to the conduct of its business; provided that the foregoing shall not prohibit any merger, consolidation, liquidation or dissolution permitted under Section 6.3.
- **Section 5.4.** <u>Payment of Obligations</u>. The Borrower will pay its obligations, including (a) all material Tax liabilities imposed upon it or upon its property, and (b) all material claims (including, but not limited to, claims for labor, materials, supplies or services) which might, if not paid, become a Lien upon its property, unless, in each case, the validity or amount thereof is being contested in good faith by appropriate proceedings and the Borrower has set aside on its books adequate reserves with respect thereto in accordance with GAAP.

### Section 5.5. Maintenance of Properties; Insurance.

- (a) The Borrower will acquire, maintain and preserve, develop, operate and construct in substantial conformity with all Transaction Documents, prudent engineering and operating practices consistent with like facilities and in material conformity with all Governmental Requirements, all elements of the Project which are used or necessary in the conduct of its business in good working order and condition (including without limitation, inventories, spare parts and system redundancies), ordinary wear and tear excepted.
- (b) The Borrower will maintain, or cause to be maintained, insurance covering it, and its properties (including the Project) in effect at all times in such amounts and covering such risks as is usually carried by companies engaged in similar business and owning similar properties in the same general area in which the Borrower operates and as is reasonably

requested by the Collateral Agent, such insurance coverage to include limits and coverage provisions sufficient to satisfy the requirements set forth in the Transaction Documents, but in no event less than the limits and coverages described in <u>Schedule 5.5(b)</u> hereto. In addition:

- (i) Each policy for liability insurance shall provide for all losses to be paid directly to the Person who shall have incurred the damage or injury covered by such policy. If any policy for liability insurance described in Schedule 5.5(b) is permitted to be written on a "claims made" basis, such insurance policy shall have a retroactive date (as such term is specified in such policy) no earlier than the Effective Date. Each time any policy written on a "claims made" basis is not renewed or the retroactive date of such policy is to be changed, the Borrower shall obtain, or cause to be obtained, for each such policy or policies the broadest basic and supplemental extended reporting period coverage, or "tail", reasonably available in the commercial insurance market for each such policy or policies, as determined in the sole reasonable judgment of the Required Lenders, but in no event less than five years after the expiration of such policy or policies.
- (ii) Each policy for property damage insurance shall provide for all amounts payable by the insured and/or insurer with respect to any property damage (except for property damage or losses of less than \$5,000,000 per occurrence, so long as the Collateral Agent has not notified the payor under any such insurance policy that a Default or an Event of Default has occurred and is continuing) to be paid directly to the Collateral Agent on behalf of the Lenders.
- (iii) Each such policy shall (except as specifically set forth below), in addition, (A) subject to the last sentence of this clause (iii), be maintained with financially sound and responsible insurance carriers either (x) rated A- X or better by A.M. Best, (y) with an investment grade rating by S&P or a rating agency of recognized standing acceptable to the Administrative Agent or (z) otherwise acceptable to the Administrative Agent; (B) with respect to each policy for liability insurance only, name the Borrower as a named insured and the Administrative Agent (in its capacity as Administrative Agent and on behalf of Lenders) and Collateral Agent as an additional insured thereunder (without any representation or warranty by or obligation upon the Collateral Agent, the Administrative Agent or Lenders), (C) with respect to each policy for liability insurance only, contain a severability of interests or cross liability clause, (D) provide that there shall be no recourse against any Lender, the Administrative Agent or the Collateral Agent for payment of premiums or other amounts with respect thereto, (E) provide that, as against the Borrower, the Lenders, the Administrative Agent and the Collateral Agent, there shall be no rights of subrogation, set-off, counterclaim or any other deduction, (F) with respect to each policy for liability insurance only, provide that such insurance shall be primary, without right of contribution from any other insurance which may be carried by the Collateral Agent, any Lender or the Administrative Agent, (G) with respect to each policy for property insurance only, provide that the Borrower shall in no event be deemed to be a co-insurer in respect of any covered claim or loss thereunder, (H) provide that at least 30 days' prior written notice of reduction, cancellation or lapse and at least 10 days' prior written notice of non-payment of premium shall be given to the Collateral Agent by

the insurer and (I) provide that the Collateral Agent may (but shall not be obligated to) cure any lapse or breach by the Borrower during such 30 day period (or 10 day period, in the case of non-payment). Notwithstanding clause (A) above, the Borrower may utilize, or cause to be utilized, an Affiliate's captive insurance program ("*Captive Insurance*") to provide property insurance subject to such Affiliate maintaining an investment grade rating with any two of S&P, Moody's and Fitch, and if the Affiliate does not meet such ratings criteria, then the Captive Insurance shall not be accepted and the property insurance shall be required to be written directly with an insurance carrier or carriers that meet the ratings requirements set forth in clause (A) above, unless otherwise approved by the Administrative Agent in consultation with its insurance consultant.

- (iv) The Borrower shall deliver to the Collateral Agent cover notes, certificates of insurance, or equivalent documentation satisfactory to the Collateral Agent in sufficient copies for each Lender. In addition, if reasonably requested by the Collateral Agent, the Borrower shall furnish the Collateral Agent with approved certification of all required insurance. Such certification shall be executed by each insurer or by an authorized representative of each insurer where it is not practical for such insurer to execute the certificate itself. Such certification shall identify underwriters, the type of insurance, the insurance limits and the policy term and shall specifically list the special provisions enumerated for such insurance required by this Section 5.5(b) and Schedule 5.5(b) hereto.
- (v) Concurrently with the furnishing of the certification referred to in clause (iv) above and within 30 days following the renewal of any insurance policy, the Borrower shall furnish to the Administrative Agent and the Collateral Agent a report of an independent broker, signed by an officer of the broker, stating that all premiums then due have been paid and that, in the opinion of such broker, the insurance then carried or to be renewed is in accordance with the terms of this <u>Section 5.5(b)</u> and <u>Schedule 5.5(b)</u> hereto.
- (vi) No provision of this Section 5.5(b) and Schedule 5.5(b) hereto or any provision of this Agreement or any Transaction Document shall impose on any Lender, the Administrative Agent or the Collateral Agent any duty or obligation to verify the existence or adequacy of the insurance coverage maintained by the Borrower, nor shall any Lender, the Administrative Agent or the Collateral Agent be responsible for any representations or warranties made by or on behalf of the Borrower to any insurance company or underwriter.
- (vii) The Borrower upon the reasonable request of Administrative Agent or the Collateral Agent will amend the amount and scope of coverage of any of the insurance policies described in <u>Schedule 5.5(b)</u> hereto to cover such risks which, in the reasonable judgment of the Administrative Agent or the Collateral Agent, would render such coverage materially inadequate.

Section 5.6. Books and Records; Inspection Rights.

- (a) The Borrower will keep proper books of record and account, in which full, true and correct entries are made of all dealings and transactions in relation to its business and activities, revenues and assets, and all costs and expenses in connection with the Project, in accordance with GAAP.
- (b) The Borrower will at any time and from time to time upon reasonable notice, but subject nevertheless to the provisions of <u>Section 10.11</u>, permit the Lenders, the Administrative Agent, the Collateral Agent and any of their respective agents and representatives to examine and make copies of and abstracts from the records and books of account of, and the properties of, the Borrower and to discuss the affairs, finances and accounts of the Borrower and of the Project with the Borrower and its officers, accountants and engineers.
- Section 5.7. Compliance with Laws. The Borrower will (a) comply with all laws, rules, regulations and orders of any Governmental Authority applicable to it or its property or the development, construction, maintenance, ownership, operation or use of the Cameron Highway System, and will cause all persons using or occupying the Cameron Highway System to so comply, except where the failure to do so, individually or in the aggregate, could not reasonably be expected to result in a Material Adverse Effect and (b) obtain and maintain all Material Permits. The Borrower will immediately pay or cause to be paid when due all costs and expenses incurred in such compliance, except to the extent that such compliance is contested in good faith by the Borrower under circumstances where none of the Collateral or the continued operation thereof, or the Liens of the Collateral Agent thereon, will be endangered nor all or any portion of the Project will be subject to loss or forfeiture.
- **Section 5.8.** <u>Use of Proceeds</u>. The proceeds of the Loans will be used only (a) to refinance all Existing Debt (including any make-whole premiums, accrued interest and other obligations in connection therewith) as and to the extent set forth in <u>Section 10.15</u> and (b) to finance the Partner Distribution. No part of the proceeds of any Loan will be used, whether directly or indirectly, for any purpose that entails a violation of any of the Regulations of the Board of Governors of the Federal Reserve System, including Regulations U and X.
- **Section 5.9.** Environmental Matters. The Borrower will establish and implement such procedures as may be necessary to continuously determine and assure that any failure of the following does not have a Material Adverse Effect: (a) all property of the Borrower, including the Project, and the operations conducted thereon are in compliance with and do not violate the requirements of any Environmental Laws, (b) no oil or solid wastes are disposed of or otherwise released on or to any property owned by the Borrower except in compliance with Environmental Laws, (c) no Hazardous Material will be released on or to any such property in a quantity equal to or exceeding that quantity which requires reporting pursuant to Section 103 of CERCLA, and (d) no Hazardous Material is released on or to any such property so as to pose an imminent and substantial endangerment to public health or welfare or the environment.

## Section 5.10. Maintain Title to Collateral; Mortgages.

(a) Subject to Section 6.3, the Borrower will maintain good and valid rights and title to the Collateral purported to be covered by the Security Documents to which the Borrower is a party.

(b) The Borrower shall deliver Mortgages (or similar documents requested by the Collateral Agent) to the Collateral Agent covering the Real Estate obtained by the Borrower or Manta Ray after the Effective Date reasonably promptly after obtaining such Real Estate, such Mortgages (or similar documents) to be duly completed and executed (as applicable) in sufficient number of counterparts and in proper form for recording. If reasonably requested by the Collateral Agent, the Borrower shall provide further supporting evidence of title, legal opinions or other evidence that such Mortgages (or similar documents) will create a valid and enforceable first priority Lien (except as permitted by Section 4.6(a)) on such Real Estate in favor of the Collateral Agent for the benefit of the Lenders.

Section 5.11. Further Assurance. The Borrower will (a) promptly execute and deliver all further instruments and documents, and take all further action, that may be necessary or that the Administrative Agent or the Collateral Agent may reasonably request in order to fully give effect to the interests and properties purported to be covered by the Security Documents, (b) promptly correct, or cause to be corrected, any defect or error that may be discovered in any Loan Document or in the execution, acknowledgment or recordation thereof and execute, acknowledge and deliver, and record and re-record, file and refile and register and re-register, any and all such further acts, deeds, conveyances, mortgages, deeds of trust, trust deeds, assignments, estoppel certificates, financing statements and continuations thereof, notices of assignment, transfers, certificates, assurances and other instruments as the Administrative Agent or the Collateral Agent may reasonably require from time to time in order (i) to carry out more effectively the purposes of this Agreement or any other Loan Document and (ii) to better assure, convey, grant, assign, transfer, preserve, protect and confirm unto the Lenders, the Administrative Agent and the Collateral Agent the rights granted or now or hereafter intended to be granted to the Lenders, the Administrative Agent and the Collateral Agent under any Loan Document or under any other instrument executed in connection with any Loan Document or that the Borrower may be or become bound to convey, mortgage or assign to the Lenders, the Administrative Agent and the Collateral Agent in order to carry out the intention or facilitate the performance of the provisions of any Loan Document, and (c) take all other actions and make all other assurances required by the Security Documents. The Borrower will furnish to the Collateral Agent evidence satisfactory to it of every such recording, filing or registration.

## Section 5.12. Performance of Transaction Documents.

- (a) The Borrower shall perform and observe all terms and provisions of each Material Document to which it is a party, maintain such Material Documents in full force and effect in accordance with their terms (unless replaced in compliance with Section 7.1(t) for the applicable types of Material Documents), and enforce such Material Documents in accordance with their terms.
- (b) The Borrower shall perform and observe all terms and provisions of each Transaction Document (other than the Material Documents, all of which are governed by Section 5.12(a)) to which it is a party, maintain such Transaction Documents in full force and effect in accordance with their terms, and enforce such Transaction Documents in accordance with their terms, except in each case as could not reasonably be expected to have a Material Adverse Effect.

(c) The Borrower shall take all such action concerning enforcement of any Transaction Document as the Administrative Agent or the Collateral Agent may from time to time reasonably request. If any Transaction Party (other than the Borrower, a Lender, the Administrative Agent or the Collateral Agent) asserts in writing its belief that a Transaction Document is not valid or binding upon such obligor, the Borrower shall allow the Administrative Agent and the Collateral Agent the right to participate in any proceeding relating thereto.

**Section 5.13.** <u>Additional Purchase and Sale Agreements</u>. If any Affiliate of the Borrower enters into an Additional Purchase and Sale Agreement with respect to the Project, the Borrower shall promptly have such document assigned to it (if the Borrower is not already a party thereto).

## Section 5.14. [Reserved]

**Section 5.15.** <u>Financial Covenants</u>. On the last day of each Fiscal Quarter that is listed below for which financial statements have or are required to have been delivered pursuant to <u>Section 5.1</u>, EBITDA shall equal or exceed the threshold set forth opposite such Fiscal Quarter in the table below:

 Fiscal Quarter Ending:
 EBITDA

 December 31, 2005
 \$ 8,000,000

 March 31, 2006
 \$ 10,000,000

**Section 5.16.** <u>Valero/Valero LP Transaction</u>. Notwithstanding any provision in any Financing Document, the Partners and the Sponsors shall have the right to consummate the transfer of partnership interests in the Borrower held by Valero CHOPS to Valero LP or a Subsidiary of Valero LP as described in <u>Section 3.4(a)(viii)</u> of the Partnership Agreement, as such Section is in effect on the Effective Date; <u>provided</u> that, for the avoidance of doubt, subsequent thereto, the Collateral Agent shall continue to have a first priority perfected Lien on 100% of the Capital Stock of the Borrower.

**Section 5.17.** Accounts. With respect to each deposit account and securities account of the Borrower that is opened after the Effective Date, the Borrower shall obtain duly executed Control Account Agreements with each bank that maintains such deposit accounts, and each securities intermediary that maintains such investment accounts, on behalf of the Borrower, as soon as practicable (and in any event within ten (10) Business Days after the opening of such account, which ten (10) Business Day period may be extended by the Administrative Agent in its sole discretion).

**Section 5.18.** <u>Post-Closing Deliveries</u>. The Borrower shall (a) deliver, or cause to be delivered, to the Administrative Agent each item set forth in <u>Schedule 5.18</u>, in form and substance satisfactory to the Administrative Agent and together with each certificate or other

document ancillary thereto and reasonably requested by the Administrative Agent and (b) perform, or cause to be performed, each action set forth in Schedule 5.18 together with each ancillary action reasonably requested by the Administrative Agent to be performed by the Borrower or any Affiliate thereof in connection therewith, in each case within the periods set forth opposite each such item or action on such Schedule or such later period as may be agreed to by the Administrative Agent in its sole discretion.

## ARTICLE VI

## **NEGATIVE COVENANTS**

Until the Commitments have expired or terminated and the principal of and interest on each Loan and all fees payable hereunder or under the other Loan Documents have been paid in full, the Borrower covenants and agrees with the Lenders that:

Section 6.1. <u>Indebtedness</u>. The Borrower will not create, incur, assume or permit to exist any Indebtedness, except:

- (a) Indebtedness created under the Loan Documents;
- (b) Indebtedness existing in connection with Interest Rate Agreements, provided that such Interest Rate Agreements are entered into by the Borrower in the ordinary course of business and for the purpose of hedging against fluctuations in interest rates;
- (c) pledges or deposits made in the ordinary course of business in connection with workers' compensation, unemployment insurance and other social security legislation, or to secure the performance to tenders, statutory obligations, surety and appeal bonds, bids, leases, construction, operating and maintenance agreements, performance and return-of-money bonds and other similar obligations (exclusive of obligations for the payment of borrowed money) and deposits securing liability to insurance carriers under insurance or self-insurance arrangements; provided, that the aggregate amount of Indebtedness described by this Section 6.1(c) shall not exceed \$25,000,000 at any time;
- (d) unsecured Indebtedness having an aggregate principal amount not exceeding \$25,000,000 at any one time outstanding; provided that for such Indebtedness to be permitted under this clause (d), the weighted average life to maturity of such Indebtedness shall be required to be greater than, and the final maturity date of such Indebtedness shall be required to be later than, that of the Loans, and each of the other material terms (including financial covenants, negative covenants and defaults) of such Indebtedness shall be no more restrictive to the Borrower than the terms of this Agreement; and
  - (e) other unsecured Indebtedness consented to by the Administrative Agent, such consent not to be unreasonably withheld.

**Section 6.2.** <u>Liens</u>. The Borrower will not create, incur, assume or permit to exist any Lien on any property or asset now owned or hereafter acquired by it, or assign or sell any income or revenues (including accounts receivable) or rights in respect of any thereof, except:

- (a) Permitted Encumbrances; and
- (b) Liens securing the Obligations.

#### Section 6.3. Fundamental Changes.

- (a) The Borrower will not merge into or consolidate with any other Person, or permit any other Person to merge into or consolidate with it, or sell, transfer, lease or otherwise dispose of (in one transaction or in a series of transactions) all or any substantial part of its assets (except as permitted by Section 6.3(c)) or liquidate or dissolve.
- (b) The Borrower will not engage in any business other than the development, construction, financing and operation of the Cameron Highway System and businesses reasonably related thereto.
- (c) The Borrower will not sell, lease, transfer or otherwise dispose of all or any substantial part of its assets, including without limitation, any portion of the Project, except (i) sales pursuant to Transaction Documents (other than Transactions Documents entered into for the primary purpose of selling, leasing, transferring or disposing of assets), the net proceeds of which shall not exceed \$10,000,000 per calendar year, (ii) sales of inventory in the ordinary course of business, (iii) sales of equipment if the same have been replaced to the extent necessary for the continued operation of the Project, the net proceeds of which shall not exceed \$12,000,000 per calendar year, (iv) sales of obsolete, worn out or surplus property in the ordinary course of business, (v) sales of assets that will not have a material effect on the continued operation of the Project, the net proceeds of which shall not exceed \$15,000,000 per calendar year and (vi) the granting of rights of way and easements necessary in connection with the operation of the Project.
  - (d) The Borrower will not discount or sell (with or without recourse) any of its notes receivable or accounts receivable.

Section 6.4. <u>Investments, Loans, Advances, Guarantees and Acquisitions</u>. The Borrower will not purchase, hold or acquire (including pursuant to any merger) any Capital Stock, evidences of Indebtedness or other securities (including any option, warrant or other right to acquire any of the foregoing) of, make or permit to exist any loans or advances to, Guarantee any obligations of, or make or permit to exist any investment or any other interest in, any other Person, or purchase or otherwise acquire (in one transaction or a series of transactions) any assets of any other Person constituting a business unit, in each case except (a) Permitted Investments and (b) pledges or deposits made in the ordinary course of business in connection with workers' compensation, unemployment insurance and other social security legislation, or to secure the performance to tenders, statutory obligations, surety and appeal bonds, bids, leases, construction, operating and maintenance agreements, performance and return-of-money bonds and other similar obligations (exclusive of obligations for the payment of borrowed money) and deposits

securing liability to insurance carriers under insurance or self-insurance arrangements; provided, that the aggregate amount of investments described by this <u>Section 6.4(b)</u> shall not exceed \$25,000,000 at any time.

**Section 6.5.** <u>Hedging Agreements</u>. The Borrower will not enter into any Hedging Agreement, other than Interest Rate Agreements permitted under <u>Section 6.1(b)</u>.

**Section 6.6.** <u>Restricted Payments</u>. The Borrower will not declare or make, or agree to pay or make, directly or indirectly, any Restricted Payment, except (a) the Partner Distribution and (b) any other Restricted Payment made in cash for which all of the Distribution Conditions have been satisfied both before and after giving effect to such Restricted Payment.

**Section 6.7.** Transactions with Affiliates. The Borrower will not sell, lease or otherwise transfer any property or assets to, or purchase, lease or otherwise acquire any property or assets from, or otherwise engage in any other transactions with, any of its Affiliates, except (a) for agreements existing on the Effective Date and described on Schedule 6.7(a), a copy of each such agreement having been delivered to the Administrative Agent, or replacements thereof on substantially the same terms, (b) for agreements on terms no less favorable to the Borrower than if the transaction had been negotiated in good faith on an arm's length basis with a non-Affiliate; provided that such agreements shall have been approved by the Required Lenders (such consent not to be unreasonably withheld) to the extent that in the aggregate, on an annual basis, such agreements exceed \$10,000,000 in value (except to the extent such agreements are purchase and sale agreements or otherwise relate to delivery transactions on the Cameron Highway Oil Pipeline, in which case no such approval shall be required), and (c) any Restricted Payment permitted by Section 6.6.

**Section 6.8.** Restrictive Agreements. The Borrower will not directly or indirectly, enter into, incur or permit to exist any agreement or other arrangement that prohibits, restricts or imposes any condition upon the ability of the Borrower to create, incur or permit to exist any Lien upon any of its property or assets; provided that the foregoing shall not apply to (i) restrictions and conditions imposed by law or by this Agreement, (ii) restrictions and conditions existing on the Effective Date hereof identified on Schedule 6.8 (but shall apply to any extension or renewal of, or any amendment or modification expanding the scope of, any such restriction or condition), and (iii) customary provisions in leases restricting the assignment thereof.

### Section 6.9. [Reserved.]

**Section 6.10.** <u>Management Fees</u>. The Borrower will not pay any management fee to any Person other than Manta Ray or any successor thereto under the terms of the Operation and Management Agreement.

**Section 6.11.** <u>Subsidiaries</u>. Notwithstanding anything to the contrary in any Loan Document, the Borrower will not (a) form or own any Subsidiaries, (b) become a general or limited partner in any partnership or a joint venturer in any joint venture, (c) acquire any ownership or Capital Stock in or make any capital contribution to any Person, (d) enter into any profit sharing or royalty agreement or other similar arrangement whereby the Borrower's income or profits are, or might be, shared with any other Person or (e) enter into any management

contract or similar arrangement whereby its business or operations are managed by any other Person other than, in the case of clause (e), as expressly contemplated by the Loan Documents.

**Section 6.12. Project Changes**. The Borrower will not make any material changes to the operation of the Project without the prior written consent of the Administrative Agent, unless such changes could not reasonably be expected to result in a Material Adverse Effect.

**Section 6.13.** Compliance With ERISA. The Borrower will not establish or permit to exist, or participate in, any Plan, any Multiemployer Plan or any Welfare Plan.

**Section 6.14.** Amendments. The Borrower will not (a) cancel or terminate, or consent to or otherwise permit the cancellation or termination of, any Material Project Document (other than (i) termination in accordance with the terms thereof after full performance of the obligations by each party thereto or (ii) if a replacement contract is entered into in compliance with Section 7.1(t) (for the applicable types of Material Project Documents)), or (b) amend or otherwise modify, or give any consent, waiver or approval to any variation of or deviation from, the material terms of any Material Project Document, or (c) consent to or permit or accept any prepayment of amounts to become due under or in connection with any Material Project Document, or (d) sell, assign (other than pursuant to the Security Documents) or otherwise dispose of (by operation of law or otherwise) any part of its interest in any Material Project Document, or (e) waive, fail to enforce, forgive, compromise, settle, adjust or release any material right, interest or entitlement, howsoever arising, under, or in respect of any Material Project Document, or (f) consent to any assignment or transfer by any Material Producer party to any Effective Date Purchase and Sale Agreement of such Effective Date Purchase and Sale Agreement, except, in the case of each of (a) through (f), (X) if such action could reasonably be expected to result in a Material Adverse Effect, with the consent of all Lenders and (Y) otherwise, with the consent of the Required Lenders. The determination whether the action will require a vote under clause (X) or (Y) of this Section 6.14 will be made by the Administrative Agent acting reasonably in accordance with such provisions, after consultation with the Borrower.

Section 6.15. Sale and Leasebacks. The Borrower will not enter into any sale and leaseback transactions.

#### ARTICLE VII

## **EVENTS OF DEFAULT**

Section 7.1. Events of Default. If any of the following events (each an "Event of Default") shall occur:

(a) the Borrower shall fail to pay any principal of any Loan when and as the same shall become due and payable, whether at the due date thereof or at a date fixed for prepayment thereof or otherwise;

- (b) the Borrower shall fail to pay any interest on any Loan or any fee, or any other amount (other than an amount referred to in clause (a) of this Section) payable under any Interest Rate Agreement with a Lender or a Secured Affiliate, this Agreement or any other Loan Document, when and as the same shall become due and payable, and such failure shall continue unremedied for a period of five Business Days;
- (c) any representation or warranty made or deemed made by or on behalf of any Partner or the Borrower in or in connection with the Loan Documents or any amendment or modification thereof or waiver thereunder, or in any report, certificate, financial statement or other document furnished pursuant to or in connection with the Loan Documents or any amendment or modification thereof or waiver thereunder, shall prove to have been incorrect in any material respect when made or deemed made;
- (d) the Borrower shall fail to observe or perform any covenant, condition or agreement contained in <u>Sections 5.2</u>, <u>5.3</u> (with respect to the Borrower's existence), <u>5.5</u>, <u>5.8</u>, <u>5.15</u>, <u>5.16</u>, <u>5.18</u> or in <u>Article VI</u>;
- (e) Any Sponsor, Partner or the Borrower shall fail to observe or perform any covenant, condition or agreement contained (i) in this Agreement (other than those specified in clause (a), (b), or (d) of this Section), and such failure shall continue unremedied for a period of 15 Business Days after written notice thereof from the Administrative Agent to the Borrower (which notice shall be given at the request of any Lender), or (ii) in any other Loan Document and such failure shall continue unremedied for a period of 15 Business Days after written notice thereof from the Administrative Agent to the Borrower (which notice shall be given at the request of any Lender);
- (f) the Borrower shall fail to make any payment (whether of principal or interest and regardless of amount) in respect of any Material Indebtedness, when and as the same shall become due and payable;
- (g) any event or condition occurs that results in any Material Indebtedness becoming due prior to its scheduled maturity or that enables or permits (with or without the giving of notice, the lapse of time or both) the holder or holders of any Material Indebtedness or any trustee or agent on its or their behalf to cause any Material Indebtedness to become due, or to require the prepayment, repurchase, redemption or defeasance thereof, prior to its scheduled maturity; provided that this clause (g) shall not apply to secured Indebtedness that becomes due as a result of the voluntary sale or transfer of the property or assets securing such Indebtedness;
- (h) an involuntary proceeding shall be commenced or an involuntary petition shall be filed seeking (i) liquidation, reorganization or other relief in respect of the Borrower or any Material Producer or its debts, or of a substantial part of its assets, under any federal, state or foreign bankruptcy, insolvency, receivership or similar law now or hereafter in effect or (ii) the appointment of a receiver, trustee, custodian, sequestrator, conservator or similar official for the Borrower or any Material Producer or for a substantial part of its assets, and, in any such case, such proceeding or petition shall continue undismissed for 60 days or an order or decree approving or ordering any of the foregoing shall be entered; provided, however, that solely with respect to the Material Producers, an Event of Default shall not exist under this clause (h) if the

net economic value to the Borrower of the Transaction Document(s) to which such Material Producer is a party is less than \$10,000,000 per annum or such Transaction Document(s) is replaced by a contract with or assigned to a third party satisfactory to the Required Lenders, with substantially the same terms and conditions, within 30 days of the commencement of such involuntary proceeding or such filing of an involuntary petition;

- (i) The Borrower or any Material Producer shall (i) voluntarily commence any proceeding or file any petition seeking liquidation, reorganization or other relief under any federal, state or foreign bankruptcy, insolvency, receivership or similar law now or hereafter in effect, (ii) consent to the institution of, or fail to contest in a timely and appropriate manner, any proceeding or petition described in clause (h) of this Section, (iii) apply for or consent to the appointment of a receiver, trustee, custodian, sequestrator, conservator or similar official for the Borrower or for a substantial part of its assets, (iv) file an answer admitting the material allegations of a petition filed against it in any such proceeding, (v) make a general assignment for the benefit of creditors or (vi) take any action for the purpose of effecting any of the foregoing; provided, however, that solely with respect to the Material Producers, an Event of Default shall not exist under this subsection (i) if the net economic value to the Borrower of the Transaction Document(s) to which such Material Producer is a party is less than \$10,000,000 per annum or such Transaction Document(s) is replaced by a contract with or assigned to a third party satisfactory to the Required Lenders, with substantially the same terms and conditions, within 30 days of the commencement of such voluntary proceeding or such filing of a voluntary petition;
  - (j) the Borrower or any Material Producer shall become unable, admit in writing its inability, or fail generally, to pay its debts as they become due;
- (k) one or more judgments that are not covered by insurance for the payment of money in an aggregate amount in excess of \$10,000,000 shall be rendered against the Borrower and the same shall remain undischarged for a period of 30 consecutive days during which execution shall not be effectively stayed, or any action shall be legally taken by a judgment creditor to attach or levy upon any assets of the Borrower to enforce any such judgment;
- (l) an ERISA Reportable Event shall have occurred that, in the opinion of the Required Lenders, when taken together with all other ERISA Reportable Events that have occurred, could reasonably be expected to result in a Material Adverse Effect;
  - (m) a Change in Control shall occur;
- (n) a breach of representation or warranty having a Material Adverse Effect, or a material default, under any Effective Date Purchase and Sale Agreement shall occur and shall continue after the applicable grace period, if any, specified therein;
  - (o) the Borrower shall at any time abandon the Project;
- (p) the termination of the Interconnect Agreement or any Effective Date Purchase and Sale Agreement; provided, however that such termination shall not result in an Event of Default if (i) approved in advance in accordance with Section 6.14, if applicable or (ii) such

Effective Date Purchase and Sale Agreement or Interconnect Agreement is terminated in accordance with its terms after full performance of the obligations by each party thereto;

- (q) the Borrower or the Project becomes subject to an Environmental Liability that could reasonably be expected to result in a Material Adverse Effect;
- (r) the Security Documents, shall for any reason, except to the extent permitted by the terms hereof or thereof, cease to create a valid and perfected first priority Lien, subject to Permitted Encumbrances (to the extent purported to be granted by such documents), in any portion of the Collateral;
  - (s) there shall occur an actual or constructive total loss of the Project; or there shall occur a Taking of all or substantially all of the Project; or
- (t) any material provision of any Material Project Document shall for any reason other than the express terms thereof cease to be valid and binding on any party thereto or the Borrower or any Transaction Party which is a party to such Material Project Document shall so assert in writing, provided, however, that, except with respect to any Effective Date Purchase and Sale Agreement and the Interconnect Agreement, such cessation or assertion shall not result in an Event of Default if a replacement contract for such Material Project Document is entered into within 30 days following such termination and such replacement contract has been approved Administrative Agent;

then, and in every such event (other than an event with respect to the Borrower described in clause (h) or (i) of this Section) and at any time thereafter during the continuance of such event, the Administrative Agent or the Collateral Agent may, and upon the written request of the Required Lenders shall, by notice to the Borrower, take any or all of the following actions, at the same or different times: (i) terminate the Commitments, whereupon the Commitment of each Lender shall terminate immediately; (ii) declare the principal of and any accrued interest on the Loans, and all other Obligations owing hereunder, to be due and payable, whereupon the same shall become due and payable immediately, without presentment, demand, protest or other notice of any kind, all of which are hereby waived by the Borrower; (iii) exercise all remedies contained in any other Loan Document; and (iv) exercise any other remedies available at law or in equity; and that, if an Event of Default specified in either clause (h) or (i) shall occur, the Commitments shall automatically terminate and the principal of the Loans then outstanding, together with accrued interest thereon, and all fees, and all other Obligations shall automatically become due and payable, without presentment, demand, protest or other notice of any kind, all of which are hereby waived by the Borrower.

**Section 7.2.** <u>Application of Proceeds from Collateral.</u> All proceeds from each sale of, or other realization upon, all or any part of the Collateral by the Administrative Agent, the Collateral Agent or any of the Lenders after an Event of Default arises shall be applied as follows:

(a) <u>first</u>, to the reimbursable expenses of the Administrative Agent and the Collateral Agent incurred in connection with such sale or other realization upon the Collateral, until the same shall have been paid in full;

- (b) <u>second</u>, to the fees and other reimbursable expenses of the Administrative Agent and the Collateral Agent then due and payable pursuant to any of the Loan Documents, until the same shall have been paid in full;
- (c) third, to all reimbursable expenses, if any, of the Lenders then due and payable pursuant to any of the Loan Documents, until the same shall have been paid in full;
  - (d) fourth, to interest then due and payable under the terms of this Agreement, until the same shall have been paid in full;
- (e) <u>fifth</u>, to the aggregate outstanding principal amount of the Loans and the Net Mark-to-Market Exposure of the Borrower, to the extent secured by Liens, until the same shall have been paid in full, allocated pro rata among the Lenders and any Secured Affiliates that hold Net Mark-to-Market Exposure based on their respective pro rata shares of the aggregate amount of such Loans and Net Mark-to-Market Exposure; and
  - (f) to the extent any proceeds remain, to the Borrower.

All amounts allocated pursuant to the foregoing clauses third through fifth to the Lenders as a result of amounts owed to the Lenders under the Loan Documents shall be allocated among, and distributed to, the Lenders pro rata based on their respective Pro Rata Shares.

**Section 7.3.** Sponsor LOCs. Without limiting any other remedies of the Collateral Agent and the Administrative Agent set forth herein or in the other Loan Documents, upon the occurrence and during the continuance of an Event of Default, or upon the occurrence of any other conditions that permit drawing under and as set forth in the Sponsor LOCs (including any exhibits, annexes, and schedules thereto), the Administrative Agent may, and upon the written request of the Required Lenders shall, draw on the Sponsor LOCs, in whole or in part, in one draw or a series of draws, and in each case *pro rata* between the two Sponsor LOCs, and shall apply the proceeds thereof against the Obligations in the manner set forth in Section 7.2.

## ARTICLE VIII

#### THE ADMINISTRATIVE AGENT

Section 8.1. <u>Appointment of Administrative Agent</u>. Each Lender irrevocably appoints SunTrust Bank as the Administrative Agent and authorizes it to take such actions on its behalf and to exercise such powers as are delegated to the Administrative Agent under this Agreement and the other Loan Documents, together with all such actions and powers that are reasonably incidental thereto. The Administrative Agent may perform any of its duties hereunder or under the other Loan Documents by or through any one or more sub-agents or attorneys-in-fact appointed by the Administrative Agent. The Administrative Agent and any such sub-agent or attorney-in-fact may perform any and all of its duties and exercise its rights and powers through their respective Related Parties. The exculpatory provisions set forth in this Article shall

apply to any such sub-agent or attorney-in-fact and the Related Parties of the Administrative Agent, any such sub-agent and any such attorney-in-fact and shall apply to their respective activities in connection with the syndication of the credit facilities provided for herein as well as activities as Administrative Agent.

Section 8.2. Nature of Duties of Administrative Agent. The Administrative Agent shall not have any duties or obligations except those expressly set forth in this Agreement and the other Loan Documents. Without limiting the generality of the foregoing, (a) the Administrative Agent shall not be subject to any fiduciary or other implied duties, regardless of whether a Default or an Event of Default has occurred and is continuing, (b) the Administrative Agent shall not have any duty to take any discretionary action or exercise any discretionary powers, except those discretionary rights and powers expressly contemplated by the Loan Documents that the Administrative Agent is required to exercise in writing by the Required Lenders (or such other number or percentage of the Lenders as shall be necessary under the circumstances as provided in Section 10.2), and (c) except as expressly set forth in the Loan Documents, the Administrative Agent shall not have any duty to disclose, and shall not be liable for the failure to disclose, any information relating to the Borrower that is communicated to or obtained by the Administrative Agent or any of its Affiliates in any capacity. The Administrative Agent shall not be liable for any action taken or not taken by it, its sub-agents or attorneys-in-fact with the consent or at the request of the Required Lenders (or such other number or percentage of the Lenders as shall be necessary under the circumstances as provided in Section 10.2) or in the absence of its own gross negligence or willful misconduct. The Administrative Agent shall not be responsible for the negligence or misconduct of any sub-agents or attorneys-in-fact selected by it with reasonable care. The Administrative Agent shall not be deemed to have knowledge of any Default or Event of Default unless and until written notice thereof (which notice shall include an express reference to such event being a "Default" or "Event of Default" hereunder) is given to the Administrative Agent by the Borrower or any Lender, and the Administrative Agent shall not be responsible for or have any duty to ascertain or inquire into (i) any statement, warranty or representation made in or in connection with any Loan Document, (ii) the contents of any certificate, report or other document delivered hereunder or thereunder or in connection herewith or therewith, (iii) the performance or observance of any of the covenants, agreements, or other terms and conditions set forth in any Loan Document, (iv) the validity, enforceability, effectiveness or genuineness of any Loan Document or any other agreement, instrument or document, or (v) the satisfaction of any condition set forth in Article III or elsewhere in any Loan Document, other than to confirm receipt of items expressly required to be delivered to the Administrative Agent. The Administrative Agent may consult with legal counsel (including counsel for the Borrower) concerning all matters pertaining to such duties.

Section 8.3. <u>Lack of Reliance on the Administrative Agent</u>. Each of the Lenders acknowledges that it has, independently and without reliance upon the Administrative Agent or any other Lender and based on such documents and information as it has deemed appropriate, made its own credit analysis and decision to enter into this Agreement. Each of the Lenders also acknowledges that it will, independently and without reliance upon the Administrative Agent or any other Lender and based on such documents and information as it has deemed appropriate, continue to make its own decisions in taking or not taking of any action under or based on this Agreement, any related agreement or any document furnished hereunder or thereunder.

Section 8.4. <u>Certain Rights of the Administrative Agent</u>. If the Administrative Agent shall request instructions from the Required Lenders with respect to any action or actions (including the failure to act) in connection with the Loan Documents, the Administrative Agent shall be entitled to refrain from such act or taking such act, unless and until it shall have received instructions from such Lenders, and the Administrative Agent shall not incur liability to any Person by reason of so refraining. Without limiting the foregoing, no Lender shall have any right of action whatsoever against the Administrative Agent as a result of the Administrative Agent acting or refraining from acting hereunder in accordance with the instructions of the Required Lenders where required by the terms of the Loan Documents.

Section 8.5. Reliance by Administrative Agent. The Administrative Agent shall be entitled to rely upon, and shall not incur any liability for relying upon, any notice, request, certificate, consent, statement, instrument, document or other writing believed by it to be genuine and to have been signed, sent or made by the proper Person. The Administrative Agent may also rely upon any statement made to it orally or by telephone and believed by it to be made by the proper Person and shall not incur any liability for relying thereon. The Administrative Agent may consult with legal counsel (including counsel for the Borrower), independent public accountants and other experts selected by it and shall not be liable for any action taken or not taken by it in accordance with the advice of such counsel, accountants or experts.

Section 8.6. The Administrative Agent in its Individual Capacity. The bank serving as the Administrative Agent shall have the same rights and powers under this Agreement and any other Loan Document in its capacity as a Lender as any other Lender and may exercise or refrain from exercising the same as though it were not the Administrative Agent; and the terms "Lenders", "Required Lenders", "holders of Notes", or any similar terms shall, unless the context clearly otherwise indicates, include the Administrative Agent in its individual capacity. The bank acting as the Administrative Agent and its Affiliates may accept deposits from, lend money to, and generally engage in any kind of business with the Borrower or any Affiliate of the Borrower as if it were not the Administrative Agent hereunder.

### Section 8.7. Successor Administrative Agent.

- (a) The Administrative Agent may resign at any time by giving notice thereof to the Lenders and the Borrower. Upon any such resignation, the Required Lenders shall have the right to appoint a successor Administrative Agent, subject to the approval by the Borrower provided that no Default or Event of Default shall exist at such time. If no successor Administrative Agent shall have been so appointed, and shall have accepted such appointment within 30 days after the retiring Administrative Agent gives notice of resignation, then the retiring Administrative Agent may, on behalf of the Lenders, appoint a successor Administrative Agent, which shall be a commercial bank organized under the laws of the United States of America or any state thereof or a bank which maintains an office in the United States, having a combined capital and surplus of at least \$500,000,000.
- (b) Upon the acceptance of its appointment as the Administrative Agent hereunder by a successor, such successor Administrative Agent shall thereupon succeed to and become vested with all the rights, powers, privileges and duties of the retiring Administrative

Agent, and the retiring Administrative Agent shall be discharged from its duties and obligations under this Agreement and the other Loan Documents. If within 45 days after written notice is given of the retiring Administrative Agent's resignation under this Section 8.7 no successor Administrative Agent shall have been appointed and shall have accepted such appointment, then on such 45th day (i) the retiring Administrative Agent's resignation shall become effective, (ii) the retiring Administrative Agent shall thereupon be discharged from its duties and obligations under the Loan Documents and (iii) the Required Lenders shall thereafter perform all duties of the retiring Administrative Agent under the Loan Documents until such time as the Required Lenders appoint a successor Administrative Agent as provided above. After any retiring Administrative Agent's resignation hereunder, the provisions of this Article VIII shall continue in effect for the benefit of such retiring Administrative Agent and its representatives and agents in respect of any actions taken or not taken by any of them while it was serving as the Administrative Agent.

**Section 8.8.** <u>Authorization to Execute other Loan Documents</u>. Each Lender hereby authorizes the Administrative Agent to execute on behalf of all Lenders all Loan Documents other than this Agreement.

#### ARTICLE IX

## THE COLLATERAL AGENT

Section 9.1. Appointment of Collateral Agent. Each Lender irrevocably appoints SunTrust Bank as the Collateral Agent and authorizes it to take such actions on its behalf and to exercise such powers as are delegated to the Collateral Agent under this Agreement and the other Loan Documents, together with all such actions and powers that are reasonably incidental thereto. The Collateral Agent may perform any of its duties hereunder or under the other Loan Documents by or through any one or more sub-agents or attorneys-in-fact appointed by the Collateral Agent. The Collateral Agent and any such sub-agent or attorney-in-fact may perform any and all of its duties and exercise its rights and powers through their respective Related Parties. The exculpatory provisions set forth in this Article shall apply to any such sub-agent or attorney-in-fact and the Related Parties of the Collateral Agent, any such sub-agent and any such attorney-in-fact and shall apply to their respective activities in connection with the syndication of the credit facilities provided for herein as well as activities as Collateral Agent.

**Section 9.2.** Nature of Duties of Collateral Agent. The Collateral Agent shall not have any duties or obligations except those expressly set forth in this Agreement and the other Loan Documents. Without limiting the generality of the foregoing, (a) the Collateral Agent shall not be subject to any fiduciary or other implied duties, regardless of whether a Default or an Event of Default has occurred and is continuing, (b) the Collateral Agent shall not have any duty to take any discretionary action or exercise any discretionary powers, except those discretionary rights and powers expressly contemplated by the Loan Documents that the Collateral Agent is required to exercise in writing by the Required Lenders (or such other number or percentage of the Lenders as shall be necessary under the circumstances as provided in Section 10.2), and (c) except as expressly set forth in the Loan Documents, the Collateral Agent shall not have any duty to disclose, and shall not be liable for the failure to disclose, any information relating to the

Borrower that is communicated to or obtained by the Collateral Agent or any of its Affiliates in any capacity. The Collateral Agent shall not be liable for any action taken or not taken by it, its sub-agents or attorneys-in-fact with the consent or at the request of the Required Lenders (or such other number or percentage of the Lenders as shall be necessary under the circumstances as provided in Section 10.2) or in the absence of its own gross negligence or willful misconduct. The Collateral Agent shall not be responsible for the negligence or misconduct of any sub-agents or attorneys-in-fact selected by it with reasonable care. The Collateral Agent shall not be deemed to have knowledge of any Default or Event of Default unless and until written notice thereof (which notice shall include an express reference to such event being a "Default" or "Event of Default" hereunder) is given to the Collateral Agent by the Borrower, any Lender, or the Administrative Agent, and the Collateral Agent shall not be responsible for or have any duty to ascertain or inquire into (i) any statement, warranty or representation made in or in connection with any Loan Document, (ii) the contents of any certificate, report or other document delivered hereunder or thereunder or in connection herewith or therewith, (iii) the performance or observance of any of the covenants, agreements, or other terms and conditions set forth in any Loan Document, (iv) the validity, enforceability, effectiveness or genuineness of any Loan Document or any other agreement, instrument or document, or (v) the satisfaction of any condition set forth in Article III or elsewhere in any Loan Document, other than to confirm receipt of items expressly required to be delivered to the Collateral Agent. The Collateral Agent may consult with legal counsel (including counsel for the Borrower) concerning all matters pertaining to such duties.

**Section 9.3.** Certain Rights of the Collateral Agent. If the Collateral Agent shall request instructions from the Required Lenders with respect to any action or actions (including the failure to act) in connection with the Loan Documents, the Collateral Agent shall be entitled to refrain from such act or taking such act, unless and until it shall have received instructions from such Lenders, and the Collateral Agent shall not incur liability to any Person by reason of so refraining. Without limiting the foregoing, no Lender shall have any right of action whatsoever against the Collateral Agent as a result of the Collateral Agent acting or refraining from acting hereunder in accordance with the instructions of the Required Lenders where required by the terms of the Loan Documents.

Section 9.4. Reliance by Collateral Agent. The Collateral Agent shall be entitled to rely upon, and shall not incur any liability for relying upon, any notice, request, certificate, consent, statement, instrument, document or other writing believed by it to be genuine and to have been signed, sent or made by the proper Person. The Collateral Agent may also rely upon any statement made to it orally or by telephone and believed by it to be made by the proper Person and shall not incur any liability for relying thereon. The Collateral Agent may consult with legal counsel (including counsel for the Borrower), independent public accountants and other experts selected by it and shall not be liable for any action taken or not taken by it in accordance with the advice of such counsel, accountants or experts.

**Section 9.5.** The Collateral Agent in its Individual Capacity. The bank serving as the Collateral Agent shall have the same rights and powers under this Agreement and any other Loan Document in its capacity as a Lender as any other Lender and may exercise or refrain from exercising the same as though it were not the Collateral Agent; and the terms "Lenders",

"Required Lenders", "holders of Notes", or any similar terms shall, unless the context clearly otherwise indicates, include the Collateral Agent in its individual capacity. The bank acting as the Collateral Agent and its Affiliates may accept deposits from, lend money to, and generally engage in any kind of business with the Borrower or any Affiliate of the Borrower as if it were not the Collateral Agent hereunder.

#### Section 9.6. Successor Collateral Agent.

- (a) The Collateral Agent may resign at any time by giving notice thereof to the Lenders and the Borrower. Upon any such resignation, the Required Lenders shall have the right to appoint a successor Collateral Agent, subject to the approval by the Borrower provided that no Default or Event of Default shall exist at such time. If no successor Collateral Agent shall have been so appointed, and shall have accepted such appointment within 30 days after the retiring Collateral Agent gives notice of resignation, then the retiring Collateral Agent may, on behalf of the Lenders, appoint a successor Collateral Agent, which shall be a commercial bank organized under the laws of the United States of America or any state thereof or a bank which maintains an office in the United States, having a combined capital and surplus of at least \$500,000,000.
- (b) Upon the acceptance of its appointment as the Collateral Agent hereunder by a successor, such successor Collateral Agent shall thereupon succeed to and become vested with all the rights, powers, privileges and duties of the retiring Collateral Agent, and the retiring Collateral Agent shall be discharged from its duties and obligations under this Agreement and the other Loan Documents. If within 45 days after written notice is given of the retiring Collateral Agent's resignation under this Section 9.6(b) no successor Collateral Agent shall have been appointed and shall have accepted such appointment, then on such 45th day (i) the retiring Collateral Agent's resignation shall become effective, (ii) the retiring Collateral Agent shall thereupon be discharged from its duties and obligations under the Loan Documents (except that in the case of any Collateral held by the Collateral Agent on behalf of the Lenders under any of the Loan Documents, the retiring Collateral Agent shall continue to hold such Collateral until such time as the successor Collateral Agent shall have been appointed) and (iii) the Required Lenders shall thereafter perform all duties of the retiring Collateral Agent under the Loan Documents until such time as the Required Lenders appoint a successor Collateral Agent as provided above. After any retiring Collateral Agent's resignation hereunder, the provisions of this Article IX shall continue in effect for the benefit of such retiring Collateral Agent and its representatives and agents in respect of any actions taken or not taken by any of them while it was serving as the Collateral Agent.

**Section 9.7.** <u>Amendment</u>. None of the terms or provisions of this <u>Article IX</u> may be amended, modified or waived without the written consent of the then Collateral Agent.

## ARTICLE X

#### **MISCELLANEOUS**

## Section 10.1. Notices.

(a) Except in the case of notices and other communications expressly permitted to be given by telephone, all notices and other communications to any party herein to be effective shall be in writing and shall be delivered by hand or overnight courier service, mailed by certified or registered mail or sent by telecopy, as follows:

To the Borrower: Cameron Highway Oil Pipeline

2727 North Loop West, Suite 700

Houston, TX 77008

Attention: Mr. Randy Fowler Telecopy Number: 713-880-8200

To the Administrative Agent SunTrust Bank

or the Collateral Agent: 303 Peachtree Street, N. E.

Atlanta, Georgia 30308 Attention: Mr. Steve Newby Telecopy Number: 404-827-6514

With a copy to: SunTrust Bank

Agency Services

303 Peachtree Street, N. E./ 25th Floor

Atlanta, Georgia 30308 Attention: Ms. Doris Folsum Telecopy Number: (404) 658-4906

To any other Lender: the address set forth in the Administrative Questionnaire or the

Assignment and Acceptance Agreement executed by such Lender

Any party hereto may change its address or telecopy number for notices and other communications hereunder by notice to the other parties hereto. All such notices and other communications shall, when transmitted by overnight delivery, or faxed, be effective when delivered for overnight (next-day) delivery, or transmitted in legible form by facsimile machine, respectively, or if mailed, upon the third Business Day after the date deposited into the mail or if delivered, upon delivery; provided, that notices delivered to the Administrative Agent or the Collateral Agent shall not be effective until actually received by such Person at its address specified in this <u>Section 10.1</u>.

(b) Any agreement of the Administrative Agent, the Collateral Agent and the Lenders herein to receive certain notices by telephone or facsimile is solely for the convenience and at the request of the Borrower. The Administrative Agent, the Collateral Agent and the Lenders shall be entitled to rely on the authority of any Person purporting to be a Person authorized by the Borrower to give such notice and the Administrative Agent, the Collateral Agent and the Lenders shall not have any liability to the Borrower or other Person on account of any action taken or not taken by the Administrative Agent, the Collateral Agent or the Lenders in reliance upon such telephonic or facsimile notice. The obligation of the Borrower to repay the Loans and all other Obligations hereunder shall not be affected in any way or to any extent by any failure of the Administrative Agent, the Collateral Agent and the Lenders to receive written confirmation of any telephonic or facsimile notice or the receipt by the Administrative Agent, the Collateral Agent and the Lenders of a confirmation which is at variance with the terms understood by the Administrative Agent, the Collateral Agent and the Lenders of a confirmation or facsimile notice.

#### Section 10.2. Waiver; Amendments.

- (a) No failure or delay by the Administrative Agent, the Collateral Agent or any Lender in exercising any right or power hereunder or any other Loan Document, and no course of dealing between the Borrower and the Administrative Agent, the Collateral Agent or any Lender, shall operate as a waiver thereof, nor shall any single or partial exercise of any such right or power or any abandonment or discontinuance of steps to enforce such right or power, preclude any other or further exercise thereof or the exercise of any other right or power hereunder or thereunder. The rights and remedies of the Administrative Agent, the Collateral Agent and the Lenders hereunder and under the other Loan Documents are cumulative and are not exclusive of any rights or remedies provided by law. No waiver of any provision of this Agreement or any other Loan Document or consent to any departure by the Borrower therefrom shall in any event be effective unless the same shall be permitted by paragraph (b) of this Section, and then such waiver or consent shall be effective only in the specific instance and for the purpose for which given. Without limiting the generality of the foregoing, the making of a Loan shall not be construed as a waiver of any Default or Event of Default, regardless of whether the Administrative Agent, the Collateral Agent or any Lender may have had notice or knowledge of such Default or Event of Default at the time.
- (b) No amendment or waiver of any provision of this Agreement or the other Loan Documents, nor consent to any departure by the Borrower therefrom, shall in any event be effective unless the same shall be in writing and signed by the Borrower and the Required Lenders or the Borrower and the Administrative Agent with the consent of the Required Lenders and then such waiver or consent shall be effective only in the specific instance and for the specific purpose for which given; provided, that no amendment or waiver shall: (i) increase the Commitment of any Lender without the written consent of such Lender, (ii) reduce the principal amount of any Loan or reduce the rate of interest thereon, or reduce any fees payable hereunder, without the written consent of each Lender affected thereby, (iii) postpone the date fixed for any payment of any principal of, or interest on, any Loan or interest thereon or any fees hereunder or reduce the amount of, waive or excuse any such payment (excluding any mandatory prepayments required under Section 2.12(b) or (c)), or postpone the scheduled date for the termination or

reduction of any Commitment, without the written consent of each Lender affected thereby, (iv) change Section 2.18(b) in a manner that would alter the pro rata sharing of payments required thereby, without the written consent of each Lender, (v) change any of the provisions of this Section or the definition of "Required Lenders" or any other provision hereof specifying the number or percentage of Lenders which are required to waive, amend or modify any rights hereunder or make any determination or grant any consent hereunder, without the consent of each Lender; (vi) release any guarantor or limit the liability of any such guarantor under any guaranty agreement, without the written consent of each Lender; or (vii) release all or substantially all collateral (if any) securing any of the Obligations or agree to subordinate any Lien in such collateral to any other creditor of the Borrower, without the written consent of each Lender; provided further, that no such agreement shall amend, modify or otherwise affect the rights, duties or obligations of the Administrative Agent or the Collateral Agent, without the prior written consent of such Person.

## Section 10.3. Expenses; Indemnification.

(a) The Borrower shall pay (i) all reasonable, out-of-pocket costs and expenses of the Administrative Agent, the Collateral Agent and their respective Affiliates, including the reasonable fees, charges and disbursements of one law firm and one additional Louisiana local counsel, for the Administrative Agent, the Collateral Agent and their respective Affiliates (and, if retention is deemed appropriate by the Administrative Agent and consented to by the Borrower (such consent not to be unreasonably withheld or delayed), additional local counsel to the Administrative Agent, the Collateral Agent and their respective Affiliates), in connection with the syndication of the credit facilities provided for herein, the preparation and administration of the Loan Documents and any amendments, modifications or waivers thereof (whether or not the transactions contemplated in this Agreement or any other Loan Document shall be consummated) and (ii) all out-of-pocket costs and expenses (including, without limitation, the reasonable fees, charges and disbursements of outside counsel and the allocated cost of inside counsel) reasonably incurred by the Administrative Agent, the Collateral Agent or any Lender in connection with the enforcement or protection of its rights in connection with this Agreement, including its rights under this Section, or in connection with the Loans made hereunder, including all such out-of-pocket expenses incurred during any workout, restructuring or negotiations in respect of such Loans.

(b) The Borrower shall indemnify the Administrative Agent (and any sub-agent thereof), the Collateral Agent (and any sub-agent thereof) and each Lender, and each Related Party of any of the foregoing Persons (each such Person being called, an "*Indemnitee*") against, and hold each Indemnitee harmless from, any and all costs, losses, liabilities, claims, damages and related expenses, including the fees, charges and disbursements of any counsel for any Indemnitee, and shall indemnify and hold harmless each Indemnitee from all fees and time charges and disbursements for attorneys who may be employees of any Indemnitee, incurred by any Indemnitee or asserted against any Indemnitee by any third party or by the Borrower, the Partners or the Sponsors arising out of, in connection with, or as a result of (i) the execution or delivery of this Agreement, any other Transaction Document or any agreement or instrument contemplated hereby or thereby, the performance by the parties hereto of their respective obligations hereunder or thereunder or the consummation of the transactions contemplated

hereby or thereby, (ii) any Loan or the use or proposed use of the proceeds therefrom, (iii) any actual or alleged presence or Release of Hazardous Materials on or from any property owned or operated by the Borrower, or any Environmental Liability related in any way to the Borrower, or (iv) any actual or prospective claim, litigation, investigation or proceeding relating to any of the foregoing, whether based on contract, tort or any other theory, whether brought by a third party or by the Borrower, the Partners or the Sponsors, and regardless of whether any Indemnitee is a party thereto, <u>provided</u> that such indemnity shall not, as to any Indemnitee, be available to the extent that such losses, claims, damages, liabilities or related expenses (x) are determined by a court of competent jurisdiction by final and nonappealable judgment to have resulted from the gross negligence or willful misconduct of such Indemnitee or (y) result from a claim brought by the Borrower, the Partners or the Sponsors against an Indemnitee for breach in bad faith of such Indemnitee's obligations hereunder or under any other Transaction Document, if the Borrower, the Partners or the Sponsors has obtained a final and nonappealable judgment in its favor on such claim as determined by a court of competent jurisdiction.

- (c) The Borrower shall pay, and hold the Administrative Agent, the Collateral Agent and each of the Lenders harmless from and against, any and all present and future stamp, documentary, and other similar taxes with respect to this Agreement and any other Loan Documents, any collateral described therein, or any payments due thereunder, and save the Administrative Agent, the Collateral Agent and each Lender harmless from and against any and all liabilities with respect to or resulting from any delay or omission to pay such taxes.
- (d) To the extent that the Borrower fails to pay any amount required to be paid to the Administrative Agent or the Collateral Agent under clauses (a), (b) or (c) hereof, each Lender severally agrees to pay to the Administrative Agent such Lender's Pro Rata Share (determined as of the time that the unreimbursed expense or indemnity payment is sought) of such unpaid amount (which the Administrative Agent shall distribute to the Collateral Agent, as applicable); provided, that the unreimbursed expense or indemnified payment, claim, damage, liability or related expense, as the case may be, was incurred by or asserted against the Administrative Agent or the Collateral Agent in their respective capacities as such.
- (e) To the extent permitted by applicable law, the Borrower shall not assert, and hereby waives, any claim against any Indemnitee, on any theory of liability, for special, indirect, consequential or punitive damages (as opposed to actual or direct damages) arising out of, in connection with or as a result of, this Agreement or any agreement or instrument contemplated hereby, the transactions contemplated therein, any Loan or the use of proceeds thereof.
  - (f) All amounts due under this Section shall be payable promptly after written demand therefor.

#### Section 10.4. Successors and Assigns.

(a) The provisions of this Agreement shall be binding upon and inure to the benefit of the parties hereto and their respective successors and assigns permitted hereby, except that the Borrower may not assign or otherwise transfer any of its rights or obligations hereunder without the prior written consent of each Lender (and any attempted assignment or transfer by the Borrower without such consent shall be null and void). Nothing in this Agreement,

expressed or implied, shall be construed to confer upon any Person (other than the parties hereto, their respective successors and assigns permitted hereby and, to the extent expressly contemplated hereby, the Related Parties of each of the Administrative Agent, the Collateral Agent and the Lenders) any legal or equitable right, remedy or claim under or by reason of this Agreement.

(b) Any Lender may assign to one or more Eligible Assignees all or a portion of its rights and obligations under this Agreement (including all or a portion of the Loans at the time owing to it); provided that (i) except in the case of an assignment of the entire remaining amount of the assigning Lender's Loans at the time owing to it or in the case of an assignment to a Lender, an Affiliate of a Lender or an Approved Fund with respect to a Lender, the aggregate amount of the Loans of the assigning Lender subject to each such assignment (determined as of the date the Assignment and Acceptance with respect to such assignment is delivered to the Administrative Agent) shall not be less than \$5,000,000 and shall result in the assigning Lender either (x) continuing to have Loans outstanding equal to or in excess of \$10,000,000 or (y) having no Loans outstanding, unless each of the Administrative Agent and, so long as no Event of Default has occurred and is continuing, the Borrower otherwise consents (each such consent not to be unreasonably withheld or delayed), (ii) each partial assignment shall be made as an assignment of a proportionate part of all the assigning Lender's rights and obligations under this Agreement with respect to the Loan assigned and (iii) the parties to each assignment shall execute and deliver to the Administrative Agent an Assignment and Acceptance, together with a processing and recordation fee of \$1,000, and the Eligible Assignee, if it shall not be a Lender, shall deliver to the Administrative Agent an Administrative Questionnaire. Upon (i) the execution and delivery of the Assignment and Acceptance by the assigning Lender and assignee Lender, (ii) acceptance and recording thereof by the Administrative Agent pursuant to paragraph (c) of this Section, (iii) consent thereof from the Borrower to the extent required pursuant to this clause (b) and (iv) if such assignee Lender is a Foreign Lender, compliance by such Person with Section 2.17(e), from and after the effective date specified in each Assignment and Acceptance, the Eligible Assignee thereunder shall be a party hereto and, to the extent of the interest assigned by such Assignment and Acceptance, have the rights and obligations of a Lender under this Agreement, and the assigning Lender thereunder shall, to the extent of the interest assigned by such Assignment and Acceptance, be released from its obligations under this Agreement (and, in the case of an Assignment and Acceptance covering all of the assigning Lender's rights and obligations under this Agreement, such Lender shall cease to be a party hereto but shall continue to be entitled to the benefits of Sections 2.15, 2.16, 2.17 and 10.3). Any assignment or transfer by a Lender of rights or obligations under this Agreement that does not comply with this paragraph shall be treated for purposes of this Agreement as a sale by such Lender of a participation in such rights and obligations in accordance with paragraph (d) of this Section.

(c) The Administrative Agent, acting solely for this purpose as an agent of the Borrower, shall maintain at one of its offices in Atlanta, Georgia a copy of each Assignment and Acceptance delivered to it and a register for the recordation of the names and addresses of the Lenders, and the principal amount of the Loans owing to, each Lender pursuant to the terms hereof from time to time (the "Register"). The entries in the Register shall be conclusive, and the Borrower, the Administrative Agent and the Lenders may treat each Person whose name is

recorded in the Register pursuant to the terms hereof as a Lender hereunder for all purposes of this Agreement, notwithstanding notice to the contrary.

- (d) Any Lender may, without the consent of, or notice to, the Borrower or the Administrative Agent sell participations to one or more banks or other entities (a "Participant") in all or a portion of such Lender's rights and/or obligations under this Agreement (including all or a portion of the Loans owing to it); provided that (i) such Lender's obligations under this Agreement shall remain unchanged, (ii) such Lender shall remain solely responsible to the other parties hereto for the performance of such obligations and (iii) the Borrower, the Administrative Agent and the other Lenders shall continue to deal solely and directly with such Lender in connection with such Lender's rights and obligations under this Agreement. Any agreement or instrument pursuant to which a Lender sells such a participation shall provide that such Lender shall retain the sole right to enforce this Agreement and to approve any amendment, modification or waiver of any provision of this Agreement; provided that such agreement or instrument may provide that such Lender will not, without the consent of the Participant, agree to any amendment, modification or waiver with respect to the following to the extent affecting such Participant: (i) increase the Commitment of any Lender without the written consent of such Lender, (ii) reduce the principal amount of any Loan or reduce the rate of interest thereon, or reduce any fees payable hereunder, without the written consent of each Lender affected thereby, (iii) postpone the date fixed for any payment of any principal of, or interest on, any Loan or interest thereon or any fees hereunder or reduce the amount of, waive or excuse any such payment, or postpone the scheduled date for the termination or reduction of any Commitment, without the written consent of each Lender affected thereby, (iv) change Section 2.18(b) in a manner that would alter the pro rata sharing of payments required thereby, without the written consent of each Lender, (v) change any of the provisions of this Section or the definition of "Required Lenders" or any other provision hereof specifying the number or percentage of Lenders which are required to waive, amend or modify any rights hereunder or make any determination or grant any consent hereunder, without the consent of each Lender; or (vi) release all or substantially all collateral (if any) securing any of the Obligations. Subject to paragraph (e) of this Section, the Borrower agrees that each Participant shall be entitled to the benefits of Sections 2.15, 2.16, and 2.17 to the same extent as if it were a Lender and had acquired its interest by assignment pursuant to paragraph (b) of this Section. To the extent permitted by law, each Participant also shall be entitled to the benefits of Section 10.7 as though it were a Lender, provided such Participant agrees to be subject to <u>Section 10.7</u> as though it were a Lender.
- (e) A Participant shall not be entitled to receive any greater payment under <u>Section 2.15</u> and <u>Section 2.17</u> than the applicable Lender would have been entitled to receive with respect to the participation sold to such Participant, unless the sale of the participation to such Participant is made with the Borrower's prior written consent. A Participant that would be a Foreign Lender if it were a Lender shall not be entitled to the benefits of <u>Section 2.17</u> unless the Borrower is notified of the participation sold to such Participant and such Participant agrees, for the benefit of the Borrower, to comply with <u>Section 2.17(e)</u> as though it were a Lender.
- (f) Any Lender may at any time pledge or assign a security interest in all or any portion of its rights under this Agreement to secure obligations of such Lender, including without limitation any pledge or assignment to secure obligations to a Federal Reserve Bank; provided

that no such pledge or assignment of a security interest shall release a Lender from any of its obligations hereunder or substitute any such pledgee or assignee for such Lender as a party hereto.

### Section 10.5. Governing Law; Jurisdiction; Consent to Service of Process.

- (a) This Agreement and the other Loan Documents shall be construed in accordance with and be governed by the law (without giving effect to the conflict of law principles thereof) of the State of New York.
- (b) The Borrower hereby irrevocably and unconditionally submits, for itself and its property, to the non-exclusive jurisdiction of the United States District Court of the Southern District of New York, and of any state court of the Supreme Court of the State of New York sitting in New York county and any appellate court from any thereof, in any action or proceeding arising out of or relating to this Agreement or any other Loan Document or the transactions contemplated hereby or thereby, or for recognition or enforcement of any judgment, and each of the parties hereto hereby irrevocably and unconditionally agrees that all claims in respect of any such action or proceeding may be heard and determined in such New York state court or, to the extent permitted by applicable law, such Federal court. Each of the parties hereto agrees that a final judgment in any such action or proceeding shall be conclusive and may be enforced in other jurisdictions by suit on the judgment or in any other manner provided by law. Nothing in this Agreement or any other Loan Document shall affect any right that the Administrative Agent, the Collateral Agent or any Lender may otherwise have to bring any action or proceeding relating to this Agreement or any other Loan Document against the Borrower or its properties in the courts of any jurisdiction.
- (c) The Borrower irrevocably and unconditionally waives any objection which it may now or hereafter have to the laying of venue of any such suit, action or proceeding described in paragraph (b) of this Section and brought in any court referred to in paragraph (b) of this Section. Each of the parties hereto irrevocably waives, to the fullest extent permitted by applicable law, the defense of an inconvenient forum to the maintenance of such action or proceeding in any such court.
- (d) Each party to this Agreement irrevocably consents to the service of process in the manner provided for notices in <u>Section 10.1</u>. Nothing in this Agreement or in any other Loan Document will affect the right of any party hereto to serve process in any other manner permitted by law.

Section 10.6. WAIVER OF JURY TRIAL. EACH PARTY HERETO IRREVOCABLY WAIVES, TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW, ANY RIGHT IT MAY HAVE TO A TRIAL BY JURY IN ANY LEGAL PROCEEDING DIRECTLY OR INDIRECTLY ARISING OUT OF THIS AGREEMENT OR ANY OTHER LOAN DOCUMENT OR THE TRANSACTIONS CONTEMPLATED HEREBY OR THEREBY (WHETHER BASED ON CONTRACT, TORT OR ANY OTHER THEORY). EACH PARTY HERETO (A) CERTIFIES THAT NO REPRESENTATIVE, AGENT OR ATTORNEY OF ANY OTHER PARTY HAS REPRESENTED, EXPRESSLY OR OTHERWISE, THAT SUCH OTHER PARTY WOULD NOT, IN THE EVENT OF

LITIGATION, SEEK TO ENFORCE THE FOREGOING WAIVER, AND (B) ACKNOWLEDGES THAT IT AND THE OTHER PARTIES HERETO HAVE BEEN INDUCED TO ENTER INTO THIS AGREEMENT AND THE OTHER LOAN DOCUMENTS BY, AMONG OTHER THINGS, THE MUTUAL WAIVERS AND CERTIFICATIONS IN THIS SECTION.

Section 10.7. Right of Setoff. In addition to any rights now or hereafter granted under applicable law and not by way of limitation of any such rights, each Lender shall have the right, at any time or from time to time upon the occurrence and during the continuance of an Event of Default, without prior notice to the Borrower, any such notice being expressly waived by the Borrower to the extent permitted by applicable law, to set off and apply against all deposits (general or special, time or demand, provisional or final) of the Borrower at any time held or other obligations at any time owing by such Lender to or for the credit or the account of the Borrower against any and all Obligations held by such Lender, as the case may be, irrespective of whether such Lender shall have made demand hereunder and although such Obligations may be unmatured. Each Lender agrees promptly to notify the Administrative Agent and the Borrower after any such set-off and any application made by such Lender, as the case may be; provided, that the failure to give such notice shall not affect the validity of such set-off and application.

**Section 10.8.** Counterparts; Integration. This Agreement may be executed by one or more of the parties to this Agreement on any number of separate counterparts (including by telecopy), and all of said counterparts taken together shall be deemed to constitute one and the same instrument. This Agreement, the Fee Letter and the other Loan Documents constitute the entire agreement among the parties hereto and thereto regarding the subject matters hereof and thereof and supersede all prior agreements and understandings, oral or written, regarding such subject matters.

Section 10.9. Survival. All covenants, agreements, representations and warranties made by the Borrower herein and in the certificates or other instruments delivered in connection with or pursuant to this Agreement shall be considered to have been relied upon by the other parties hereto and shall survive the execution and delivery of this Agreement and the making of any Loans regardless of any investigation made by any such other party or on its behalf and notwithstanding that the Administrative Agent, the Collateral Agent or any Lender may have had notice or knowledge of any Default or incorrect representation or warranty at the time any credit is extended hereunder, and shall continue in full force and effect as long as the principal of or any accrued interest on any Loan or any fee or any other amount payable under this Agreement is outstanding and unpaid. The provisions of Sections 2.15, 2.16, 2.17, and 10.3 and Articles VIII and IX shall survive and remain in full force and effect regardless of the consummation of the transactions contemplated hereby, the repayment of the Loans, the expiration or termination of the Commitments or the termination of this Agreement or any provision hereof. All representations and warranties made herein, in the certificates, reports, notices, and other documents delivered pursuant to this Agreement shall survive the execution and delivery of this Agreement and the other Loan Documents and the making of the Loans.

**Section 10.10.** Severability. Any provision of this Agreement or any other Loan Document held to be illegal, invalid or unenforceable in any jurisdiction, shall, as to such jurisdiction, be ineffective to the extent of such illegality, invalidity or unenforceability without affecting the legality, validity or enforceability of the remaining provisions hereof or thereof; and the illegality, invalidity or unenforceability of a particular provision in a particular jurisdiction shall not invalidate or render unenforceable such provision in any other jurisdiction.

Section 10.11. Confidentiality. Each of the Administrative Agent, the Collateral Agent and each Lender agrees to take normal and reasonable precautions to maintain the confidentiality of any information received from the Borrower or its Affiliates relating to the Sponsors, the Pledgors, the Borrower or Manta Ray or their respective businesses, other than any such information that is available to the Administrative Agent, the Collateral Agent or any Lender on a nonconfidential basis prior to such disclosure by the Borrower or its Affiliates (collectively, the "Information"), except that such Information may be disclosed (i) to any Related Party of the Administrative Agent, the Collateral Agent or any such Lender, including without limitation accountants, legal counsel and other advisors, (ii) to the extent required by applicable laws or regulations or by any subpoena or similar legal process, (iii) to the extent requested by any regulatory agency or authority, (iv) to the extent that such Information becomes publicly available other than as a result of a breach of this Section, or which becomes available to the Administrative Agent, the Collateral Agent, any Lender or any Related Party of any of the foregoing on a nonconfidential basis from a source other than the Borrower, (v) in connection with the exercise of any remedy hereunder or any suit, action or proceeding relating to this Agreement or the enforcement of rights hereunder, and (ix) subject to provisions substantially similar to this Section 10.11, to any actual or prospective assignee or Participant, or (vi) with the consent of the Borrower. Any Person required to maintain the confidentiality of any Information as provided for in this Section shall be considered to have complied with its obligation to do so if such Person has exercised the same degree of care to maintain the confidentiality of such Information as such Person would accord its own confidential information.

Section 10.12. <u>Interest Rate Limitation</u>. Notwithstanding anything herein to the contrary, if at any time the interest rate applicable to any Loan, together with all fees, charges and other amounts which may be treated as interest on such Loan under applicable law (collectively, the "*Charges*"), shall exceed the maximum lawful rate of interest (the "*Maximum Rate*") which may be contracted for, charged, taken, received or reserved by a Lender holding such Loan in accordance with applicable law, the rate of interest payable in respect of such Loan hereunder, together with all Charges payable in respect thereof, shall be limited to the Maximum Rate and, to the extent lawful, the interest and Charges that would have been payable in respect of such Loan but were not payable as a result of the operation of this Section shall be cumulated and the interest and Charges payable to such Lender in respect of other Loans or periods shall be increased (but not above the Maximum Rate therefor) until such cumulated amount, together with interest thereon at the Federal Funds Rate to the date of repayment, shall have been received by such Lender.

**Section 10.13.** Waiver of Effect of Corporate Seal. The Borrower represents and warrants that neither it nor the Partners or the Sponsors is required to affix its corporate seal to this Agreement or any other Loan Document pursuant to any requirement of law or regulation

and waives any shortening of the statute of limitations that may result from not affixing the corporate seal to this Agreement or such other Loan Documents.

**Section 10.14.** <u>Limitation on Recourse.</u> Except as and to the extent otherwise expressly provided in respect of the Pledge Agreements, the Mortgages and the Sponsor LOCs:

- (a) payment or performance of any of the obligations of the Borrower, any Partner and Manta Ray under any of the Loan Documents will be obligations of the Borrower, such Partner and Manta Ray only, as applicable, and the Administrative Agent, the Collateral Agent and the Lenders will have recourse solely to (i) the assets and cash flow of the Partners (as expressly provided in the Pledge Agreements), (ii) Manta Ray (as expressly provided in the Mortgages) and (iii) the Borrower; and
- (b) the Administrative Agent, the Collateral Agent and the Lenders will not have recourse to, or claims upon, the assets of the Sponsors, the Partners or Manta Ray or to the shareholders, officers, directors or employees of the Borrower, the Sponsors, the Partners or Manta Ray, for the payment or performance of any of the obligations of the Borrower, any Partner or Manta Ray under any of the Loan Documents.

Notwithstanding the foregoing, the limitations on recourse provided in this Section 10.14 shall not apply with respect to any claim based on fraud or bad faith.

#### Section 10.15. Amendment and Restatement.

- (a) On the Effective Date: (w) JPMorgan Chase Bank shall resign as Administrative Agent, Depository Bank, Collateral Agent, Intercreditor Agent and Initial Noteholder Agent (all as defined in the Existing Debt Documents), (x) the Construction Loan Commitment and the Senior Note Commitment (as defined in the Existing Debt Documents) of any Existing Lender shall be restructured and rearranged as Commitments of the Lenders hereunder, (y) the outstanding Construction Loans and Institutional A Loans under and as defined in the Existing Debt Documents shall continue and be restructured, renewed and rearranged as Loans and (z) the Lenders shall hereby appoint SunTrust Bank as Administrative Agent and Collateral Agent as provided in Article VIII and Article IX.
- (b) On the Effective Date, the Existing Debt Documents shall be amended and restated in their entirety by this Agreement, and the Existing Debt Documents shall thereafter be of no further force and effect, except to the extent amended and restated hereby. This Agreement is not in any way intended to constitute a novation of the obligations and liabilities existing under the Existing Debt Documents or evidence payment of all or any portion of such obligations and liabilities, and it is the intention of the Borrower, the Administrative Agent, the Collateral Agent and the Lenders that the Liens securing the Secured Obligations (as defined in the Existing Debt Documents) will continue and be assigned to the Collateral Agent for the benefit of the Lenders.
- (c) On and after the Effective Date, (i) all references to the Existing Debt Documents (or to any amendment or any amendment and restatement thereof) in the Loan Documents (other than this Agreement) shall be deemed to refer to the Existing Debt

Documents, as amended and restated hereby, (ii) all references to any section (or subsection) of the Existing Debt Documents or in any Loan Document (but not herein) shall be amended to become, *mutatis mutandis*, references to the corresponding provisions of this Agreement and (iii) except as the context otherwise provides, on or after the Effective Date, all references to this Agreement herein (including for purposes of indemnification and reimbursement of fees) shall be deemed to be references to the Existing Debt Documents, as amended and restated hereby.

(d) Each of the Administrative Agent, the Collateral Agent, the Lenders and the Borrower acknowledges that SunTrust Bank has succeeded JPMorgan Chase Bank as Administrative Agent and Collateral Agent under the Existing Debt Documents, as amended and restated hereby.

(remainder of page left intentionally blank)

**IN WITNESS WHEREOF**, the parties hereto have caused this Agreement to be duly executed by their respective authorized officers as of the day and year first above written.

## **CAMERON HIGHWAY OIL PIPELINE COMPANY**

By /S/ Randy Fowler

Name: Randy Fowler

Title: Vice President and Treasurer

SUNTRUST BANK, as Administrative Agent, as Collateral Agent and as a Lender  $\,$ 

By /S/ Peter C. Vaky

Name: Peter C. Vaky Title: Managing Director

[SIGNATURE PAGE TO AMENDED AND RESTATED CREDIT AGREEMENT — 1]

# Schedule I

Commitments

 Lender
 Commitment Amount

 SunTrust Bank
 \$ 415,000,000.00

# SCHEDULE I

### Annex I

#### Pricing Grid

New Rating*	LIBOR Spread	Base Rate Spread
Greater than or equal to BBB- (Tier 1)	1.25%	0.25%
BB+ (Tier 2)	1.50%	0.50%
BB (Tier 3)	1.75%	0.75%
Less than BB (Tier 4)	2.00%	1.00%

<sup>\*</sup> New Rating to be based on S&P, Moody's and/or Fitch ratings. In the case of Moody's and Fitch ratings, the Tiers in the Pricing Grid above shall be determined by reference to the equivalent of the S&P ratings set forth in the Pricing Grid above and as further detailed in this paragraph. If none of S&P, Moody's or Fitch shall have in effect a New Rating (either after a New Rating has previously been obtained or after October 31, 2005), then Pricing shall be deemed to be Tier 2. If one of S&P, Moody's or Fitch shall have in effect a New Rating, then Pricing shall be deemed to be the applicable Tier corresponding to such rating. If two or more of Moody's, S&P and Fitch have established a New Rating and all such ratings differ by no more than one Tier, Pricing shall be determined by reference to the higher rating (i.e., the Tier with the smaller spreads). If two or more of Moody's, S&P and Fitch have established a New Rating and any two such ratings differ by two or more Tiers, Pricing shall be determined by reference to the Tier next below that corresponding to the highest of all such ratings. If the New Rating established or deemed to have been established by S&P, Moody's or Fitch shall be changed, such change shall be effective as of the date on which it is first publicly announced by such rating agency. Each change in the Pricing shall apply during the period commencing on the effective date of such change and ending on the date immediately preceding the effective date of the next such change. If the rating system of any of S&P, Moody's or Fitch shall change, or if any such rating agency shall cease to be in the business of rating corporate debt obligations, the Borrower and the Lenders shall negotiate in good faith to amend this definition to reflect such changed rating system or the unavailability of ratings form such rating agency and, pending the effectiveness of any such amendment, the Applicable Margin shall be determined by reference to the rating most recently in effect prior to such change or cessati

#### ANNEX I